3.7 Custom Business Services (continued)

3.7.41 Business Domestic Saver 15 Connections 2 Plus Service, 1 Year<sup>1</sup>

С

- (A) Business Domestic Saver 15 Connections 2 Plus Service 1 Year is a custom combination inbound, outbound, and calling card Flat Rate Service available to Business Customers that:
  - .1 request to be provisioned under this Service;
  - .2 utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS;

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

N N

Issued: January 28, 2004

Effective: February 28, 2004

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Original Sheet 194.120

Ν

N

**RECD APR 18** 2003

## SECTION 3 - DESCRIPTION OF SERVICES AVIO

- 3.7 Custom Business Services (continued)
  - 3.7.41 Business Domestic Saver 15 Connections 2 Plus Service, 1 Year
    - (A) Business Domestic Saver 15 Connections 2 Plus Service 1 Year is a custom combination inbound, outbound, and calling card Flat Rate Service available to Business Customers that:
      - .1 request to be provisioned under this Service;
      - .2 utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS:

CANCELLED

FEB 2 8 2004 By ISAS 194.120 Public Service Commission MISSOURI

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Servico Commission

- 3.7 Custom Business Services (continued)
  - 3.7.41 Business Domestic Saver 15 Connections 2 Plus Service 1 Year<sup>1</sup>
    - (A) (continued)
      - .3 subscribe to and maintain the following products or services provided by a SBC affiliate: (1) one DSL service or shared web hosting or dedicated Internet access or SBC® Yahoo!® dial-up Internet access or T1 Integrated Access or dedicated web hosting; and (2) a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business SolutionsK, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageK, Pacific Bell Instant OfficeK:, Power Office, The Business Plan, Custom Biz SaverK, Local Usage SaverK or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from a SBC Affiliate:
        - .a a minimum of one business access line, and,
        - .b an inside wire maintenance product associated with each business access line, and,
        - .c at least one instance of Caller ID, and,
        - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate;

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

N N

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

### Missourt Public SECTION 3 - DESCRIPTION OF SERVICESorviso Commission

3.7 Custom Business Services (continued)

- **RECD APR 18 2003** N
- 3.7.41 Business Domestic Saver 15 Connections 2 Plus Service 1 Year
  - (A) (continued)
    - .3 subscribe to and maintain the following products or services provided by a SBC affiliate: (1) one DSL service or shared web hosting or dedicated Internet access or SBC® Yahoo!® dial-up Internet access or T1 Integrated Access or dedicated web hosting; and (2) a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business Solutions<sup>SM</sup>, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature Package<sup>SM</sup>, Pacific Bell Instant Office<sup>SM</sup>:, Power Office, The Business Plan, Custom Biz Saver<sup>SM</sup>, Local Usage Saver<sup>SM</sup> or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from a SBC Affiliate:
      - .a a minimum of one business access line, and,
      - .b an inside wire maintenance product associated with each business access line, and,
      - .c at least one instance of Caller ID, and,
      - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate;

# CANCELLED

FEB 2 8 2004 By GARS 194,121 Public Service Commission MISSOURI

Issued: April 18, 2003

Effective: May 19, 2003

N

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

3.7 Custom Business Services (continued)

3.7.41 Business Domestic Saver 15 Connections 2 Plus Service 1 Year<sup>1</sup> (continued)

С

- (A) (continued)
  - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.41 (A).3 of this Tariff;
  - .5 except as described below, associate the billing for the products or services of a SBC Affiliate that are described in Section 3.7.41 (A).3 of this Tariff with the Customer's BTN for Business Domestic Saver 15 Connections 2 Plus Service 1 Year. Exceptions to this requirement are:
    (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Domestic Saver 15 Connections 2 Plus Service 1 Year will be billed; and

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations N effective April 1, 2004.

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

	western B SBC Long			cations Services, Inc. PSC Mo No. 1 Ist Revised Sheet 194.122 Replacing Original Sheet 194.122	
				SECTION 3 - DESCRIPTION OF SERVICES Missouri P	ublic
3.7	Custon	n Busi	ness S	ervices (continued) RECD JUL 2	5 2003
	3.7.41	Busi	ness D	omestic Saver 15 Connections 2 Plus Service 1 Year (Smithuele Comr	nission
		(A)	(con	tinued)	
•			.4	demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.41 (A).3 of this Tariff;	
			.5	except as described below, associate the billing for the products or services of a SBC Affiliate that are described in Section 3.7.41 (A).3 of this Tariff with the Customer's BTN for Business Domestic Saver 15	С
				Connections 2 Plus Service 1 Year. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Domestic Saver 15 Connections 2 Plus Service 1 Year will be billed; and	C       
					С

# CANCELIEN

FEB 2 8 2004 By 2. A.R.S. 194.122 Public Service Commission MISSOURI

Issued: July 25, 2003

\$

ł

ł

Effective: August 24, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Nesicianse à const

FILED AUG 2 4 2003

Service Commission

### Missouri Public SECTION 3 - DESCRIPTION OF SERVICES Sorvido Commission RECT APR 18 2003 N 3.7 Custom Business Services (continued) 3.7.41 Business Domestic Saver 15 Connections 2 Plus Service 1 Year (continued) (A) (continued) .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.41 (A).3 of this Tariff; .5 with the exception of Cingular service, associate the billing for the products or services of a SBC Affiliate that are described in Section 3.7.41 (A).3 of this Tariff with the Customer's BTN for Business



Domestic Saver 15 Connections 2 Plus Service 1 Year: and

Issued: April 18, 2003

Effective: May 19, 2003

Ν

Tawnya Rechtin. Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Sorvice Commission

Custom Business Services (continued) 3.7

- 3.7.41 Business Domestic Saver 15 Connections 2 Plus Service 1 Year<sup>1</sup> (continued) С
  - (A) (continued)
    - commit to an MMC of \$15 per month for a 1-year term. .6

For rules and regulations regarding MMCs and term plans, see Section 6.28 of this Tariff.

This Service is established at the BTN level and is only available for a single BTN. Service is only available for Customers that commit to a 1-year plan.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

Ν Ν

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Original Sheet 194.123

#### Missouri Public SECTION 3 - DESCRIPTION OF SERVICES Sorvice Commission

3.7 Custom Business Services (continued)

### **REGD** APR 18 2003 N

N

3.7.41 Business Domestic Saver 15 Connections 2 Plus Service 1 Year (continued)

(A) (continued)

.6 commit to an MMC of \$15 per month for a 1-year term.

For rules and regulations regarding MMCs and term plans, see Section 6.28 of this Tariff.

This Service is established at the BTN level and is only available for a single BTN. Service is only available for Customers that commit to a 1-year plan.

# CANCELLFD

FEB 2 8 2004 By GARS194.123 Public Service Commission MISSOURI

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin. Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton. California 94588 Missouri Public Service Commission

- 3.7 Custom Business Services (continued)
  - 3.7.41 Business Domestic Saver 15 Connections 2 Plus Service 1 Year<sup>1</sup> (continued)
    - **(B)** The Customer may subscribe to Business Domestic Saver 15 Connections 2 Plus Service 1 Year for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.41 (A).3 of this Tariff.
    - Toll free calls may originate on any type of access and are terminated via (C) Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
    - (D) The Customer's usage rate for each call is based a 1-year term plan.
    - (E) Outbound calls, TFS calls, and fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 are C/R billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.41 of this Tariff.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

Issued: March 1, 2004

Effective: April 1, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

	western B SBC Long		nmunications Services, Inc. PSC Mo No. 1	1st Revised Sheet 194.124 placing Original Sheet 194.124			
			SECTION 3 - DESCRIPTION OF SERVICE	s Missouri Public			
3.7	Custom Business Services (continued) RECD JAN 28						
	3.7.41	Busi	1 Year <sup>l</sup> (continued) C Service Commission				
		(B)	The Customer may subscribe to Business Domest Plus Service 1 Year for outbound Service only, Th outbound and TFS for a single BTN. The start of after the installation date of the required products Section 3.7.41 (A).3 of this Tariff.	FS only or for both Service date may be on or			
Switched Access to the Customer's loca			Toll free calls may originate on any type of access Switched Access to the Customer's location. See optional features, rules and regulations, and gener TFS.	Section 3.6 of this Tariff for			
		(D)	The Customer's usage rate for each call is based a	l-year term plan.			
		(E)	Outbound calls, TFS calls, and fully automated, o operator dialed calls billed to the Calling Card - C increments of one (1) second subject to a minimu period) of thirty (30) seconds. For rates and charge	Option 2 are billed in m connect time (initial			

this Tariff.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

### CANCELLED

Ν

Ν

APR 61 Public Service Commission 1ssued: January 28, 2004 Effective: February 28, 2004 MISSOURI Norm Descoteaux, Associate Director Regulatory

5850 W. Las Positas Blvd., Pleasanton, California 94588

FLED FEB 28 2004

Missouri Public Sorviso Commission

Original Sheet 194.124

<u>I ODC Lon</u>	<u>2 Dista</u>		Missouri Public Service Commission				
Custon	DECO ADD 1 9 2002						
3.7.41 Business Domestic Saver 15 Connections 2 Plus Service 1 Year (continued)							
	(B)	The Customer may subscribe to Business De Plus Service 1 Year for outbound Service or and TFS for a single BTN. The start of Serv installation date of the required products or s (A).3 of this Tariff.	nly, TFS only or for both outbound vice date may be on or after the				
	(C)	Toll free calls may originate on any type of Switched Access to the Customer's location, optional features, rules and regulations, and	. See Section 3.6 of this Tariff for				
	(D)	The Customer's usage rate for each call is ba	ased a 1-year term plan.				
	(E)	Outbound calls, TFS calls, and fully automa dialed calls billed to the Calling Card - Opti (1) second subject to a minimum connect tir seconds. For rates and charges, see Section	on 2 are billed in increments of one me (initial period) of thirty (30)				
			CANCELLED				
	Custon	Custom Busi 3.7.41 Busi (B) (C)	<ul> <li>(B) The Customer may subscribe to Business D Plus Service 1 Year for outbound Service or and TFS for a single BTN. The start of Ser- installation date of the required products or s (A).3 of this Tariff.</li> <li>(C) Toll free calls may originate on any type of Switched Access to the Customer's location optional features, rules and regulations, and</li> <li>(D) The Customer's usage rate for each call is ba (E) Outbound calls, TFS calls, and fully automa dialed calls billed to the Calling Card - Opti (1) second subject to a minimum connect time</li> </ul>				

Issued: April 18, 2003

ł

Effective: May 19, 2003

mission

FEB 2 8 2004

Tawnya Rechtin. Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Publie

Missouri Public Service Commission

3.7 Custom Business Services (continued)

3.7.41 Business Domestic Saver 15 Connections 2 Plus Service 1 Year<sup>1</sup> (continued)

С

Ν

Ν

(F) If the Customer fails to maintain the required products or services described in Section 3.7.41 (A).3 of this Tariff, the Customer will no longer qualify for Business Domestic Saver 15 Connections 2 Plus Service 1 Year and will be moved to Business Domestic Saver 15 for an additional 1-year term, unless the Customer selects an alternative Service. If the Customer is moved to Business Domestic Saver 15, the rates and charges in Section 4.7.11 of the Tariff will apply in lieu of the rates and charges in Section 4.7.41 of this Tariff.

If the Customer is moved to Business Domestic Saver 15 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Domestic Saver 15 Connections 2 Plus Service, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 6.28 of this Tariff.

At the end of the 1-year term, the Customer will be moved to Business Domestic Saver 15 Connections 2 Service for an additional 1-year term, as described in Section 3.7.21 of this tariff, unless otherwise specified by the Customer.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

1st Revised Sheet 194.125 Replacing Original Sheet 194.125 Missoun Public

### SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

REC'D JUL 2 5 2003

3.7.41 Business Domestic Saver 15 Connections 2 Plus Service 1 Year (continued)

 (F) If the Customer fails to maintain the required products or services described in Section 3.7.41 (A).3 of this Tariff, the Customer will no longer qualify for Business Domestic Saver 15 Connections 2 Plus Service 1 Year and will be moved to Business Domestic Saver 15 for an additional 1-year term, unless the Customer selects an alternative Service. If the Customer is moved to Business Domestic Saver 15, the rates and charges in Section 4.7.11 of the Tariff will apply in lieu of the rates and charges in Section 4.7.41 of this Tariff.

If the Customer is moved to Business Domestic Saver 15 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Domestic Saver 15 Connections 2 Plus Service, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 6.28 of this Tariff.

At the end of the 1-year term, the Customer will be moved to Business Domestic Saver 15 Connections 2 Service for an additional 1-year term, as described in Section 3.7.21 of this tariff, unless otherwise specified by the Customer.

# CANCELLED

8 2004 mission

Issued: July 25, 2003

Effective: August 24, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Miscouri Public

С

FILED AUG 2 4 2003

Service Commission

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

I

Original Sheet 194.125

Albooned English

		Service Commission Service Commission RECD APR 18 2003					
3.7	Custom Business Services (continued)						
	3.7.41	Business Domestic Saver 15 Connections 2 Plus Service 1 Year (continued)					
		<ul> <li>(F) If the Customer fails to maintain the required products or services described in Section 3.7.41 (A).3 of this Tariff, the Customer will no longer qualify for Business Domestic Saver 15 Connections 2 Plus Service 1 Year and will be moved to Business Domestic Saver 15 unless the Customer selects an alternative Service. If the Customer is moved to Business Domestic Saver 15, the rates and charges in Section 4.7.11 of the Tariff will apply in lieu of the rates and charges in Section 4.7.41 of this Tariff.</li> </ul>					
		If the Customer is moved to Business Domestic Saver 15 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Domestic Saver 15 Connections 2 Plus Service, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 6.28 of this Tariff.					
		At the end of the 1-year term, the Customer will be moved to Business Domestic Saver 15 Connections 2 Service, as described in Section 3.7.21 of this tariff, unless otherwise specified by the Customer.					
		TIED					



Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

<sup>8</sup> Missouri Public Service Commiccion

FILED MAY 19 2003

- 3.7 Custom Business Services (continued)
  - 3.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year<sup>1</sup>

С

- (A) Business Long Distance 50 Connections 2 Plus Service 1 Year is a custom combination switched TFS, outbound, and calling card Flat Rate Service available to Business Customers that:
  - .1 request to be provisioned under this Service;
  - .2 utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS;

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations	Ν
effective April 1, 2004.	Ν

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

	SECTIO	ON 3 - DESCRIPTION OF SERVICES Missouri Public Servico Commiani	on
Custom Busir	ess Services		N
3.7.42 Busir	less Long Dis	tance 50 Connections 2 Plus Service 1 Year	{ {
(A)	combination	ng Distance 50 Connections 2 Plus Service 1 Year is a custom switched TFS, outbound, and calling card Flat Rate Service Business Customers that:	
	.1	request to be provisioned under this Service;	i i
	.2	utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS;	   

CANCELLED FEB 2 8 2004 120 Public

Issued: April 18, 2003

3.7

ļ

l

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public Service Commission

- 3.7 Custom Business Services (continued)
  - 3.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year<sup>1</sup>

С

- (A) (continued)
  - .3 subscribe to and maintain the following products or services provided by a SBC affiliate: (1) one DSL service or shared web hosting or dedicated Internet access or SBC® Yahoo!® dial-up Internet access or T1 Integrated Access or dedicated web hosting; and (2) a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business SolutionsK, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageK, Pacific Bell Instant OfficeK :, Power Office, The Business Plan, Custom Biz SaverK, Local Usage SaverK or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from a SBC Affiliate:
    - .a a minimum of one business access line, and,
    - .b an inside wire maintenance product associated with each business access line, and,
    - .c at least one instance of Caller ID, and,
    - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

N N

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Ν

### SECTION 3 - DESCRIPTION OF SERVICES Missouri Public Service Commission

3.7 Custom Business Services (continued)

- REC'D APR 18 2003
- 3.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year
  - (A) (continued)
    - .3 subscribe to and maintain the following products or services provided by a SBC affiliate: (1) one DSL service or shared web hosting or dedicated Internet access or SBC® Yahoo!® dial-up Internet access or T1 Integrated Access or dedicated web hosting; and (2) a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business Solutions<sup>SM</sup>, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature Package<sup>SM</sup>, Pacific Bell Instant Office<sup>SM</sup>:, Power Office, The Business Plan, Custom Biz Saver<sup>SM</sup>, Local Usage Saver<sup>SM</sup> or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from a SBC Affiliate:
      - .a a minimum of one business access line, and,
      - .b an inside wire maintenance product associated with each business access line, and,
      - .c at least one instance of Caller ID, and,

Tawnya Rechtin, Associate Director Regulatory

d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.

FEB 2 8 2004

mission

Issued: April 18, 2003

5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public Service Commission

Effective: May 19, 2003



N

### 3.7 Custom Business Services (continued)

3.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year<sup>1</sup> (continued)

С

- (A) (continued)
  - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.42 (A).3 of this Tariff;
  - .5 except as described below, associate the billing for the products or services of a SBC Affiliate that are described in Section 3.7.42 (A).3 of this Tariff with the Customer's BTN for Business Long Distance 50 Connections 2 Plus Service 1 Year. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Long Distance 50 Connections 2 Plus Service 1 Year will be billed; and
  - .6 commit to an MMC of \$50 per month for a 1-year term

For rules and regulations regarding the MMC and term plans, see Section 6.28 of this Tariff. This Service is established at the BTN level and is only available for a single BTN. Service is only available for Customers that commit to a 1-year term plan.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

N N

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

1st Revised Sheet 194.128 Replacing Original Sheet 194.128

### SECTION 3 - DESCRIPTION OF SERVICES

### REC'D JUL 2 5 2003

3.7 Custom Business Services (continued)

Service Commission

3.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year (continued)

(A) (continued)

Issued: July 25, 2003

- .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.42 (A).3 of this Tariff;
- .5 except as described below, associate the billing for the products or C services of a SBC Affiliate that are described in Section 3.7.42 (A).3 of this Tariff with the Customer's BTN for Business Long Distance 50 Connections 2 Plus Service 1 Year. Exceptions to this requirement are: C (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC
  Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying IDSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business I Long Distance 50 Connections 2 Plus Service 1 Year will be billed; and

.6 commit to an MMC of \$50 per month for a 1-year term

For rules and regulations regarding the MMC and term plans, see Section 6.28 of this Tariff. This Service is established at the BTN level and is only available for a single BTN. Service is only available for Customers that commit to a 1-year term plan.

2 8 2004 mission

Effective: August 24, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Public

Miscouri Public

C

FILED AUG 2 4 2003

Service Commission

				SECTION 3 - DESCRIPTION OF SERVICES	Missouri Public Service Commission			
3.7	Custon	n Busi	iness Services (continued) <b>RECD APR 18</b> 2003 <sup>N</sup>					
	3.7.42	Busi	Business Long Distance 50 Connections 2 Plus Service 1 Year (continued)					
		(A)	(con	tinued)				
			.4	demonstrate to the satisfaction of the Compan	y at the time of			
				subscribing to the Service that the Customer a	also subscribes to the			
				products or services described in Section 3.7.4	42 (A).3 of this Tariff:			
			.5	with the exception of Cingular service, associ	ate the billing for the			
				products or services of a SBC Affiliate that at	e described in Section			
				3.7.42 (A).3 of this Tariff with the Customer's	S BTN for Business Long			
				Distance 50 Connections 2 Plus Service 1 Ye	ar; and			
			.6	commit to an MMC of \$50 per month for a 1	-year term			
			For rules and regulations regarding the MMC and term plans, see Section 6.2 of this Tariff. This Service is established at the BTN level and is only					
		available for a single BTN. Service is only available for Customers that commit to a 1-year term plan.						
					ł			
					N			



Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public Service Commiscien

- 3.7 Custom Business Services (continued)
  - 3.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year<sup>1</sup> (continued)
    - **(B)** The Customer may subscribe to Business Long Distance 50 Connections 2 Plus Service 1 Year for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.42 (A).3 of this Tariff.
    - Toll free calls may originate on any type of access and are terminated via (C) Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
    - (D) The Customer's usage rate for each call is based a 1-year term plan.
    - (E) Outbound calls, TFS calls, and fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.42 of this Tariff.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

Issued: March 1, 2004

Effective: April 1, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

SECTION 3 - DESCRIPTION OF SERVICES Missouri Public

3.7 Custom Business Services (continued)

### RECT JAN 2 8 2004

- 3.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year<sup>1</sup>. (continued) C Service Commission
  - (B) The Customer may subscribe to Business Long Distance 50 Connections 2 Plus Service 1 Year for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.42 (A).3 of this Tariff.
  - (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
  - (D) The Customer's usage rate for each call is based a 1-year term plan.
  - (E) Outbound calls, TFS calls, and fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.42 of this Tariff.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

CANCELLED

APR 0 1 2004 Public

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Missouri Public Service Ceramicalon

N N

FILE) FEB 28 2004

Missouri Public Servico Commiccion SECTION 3 - DESCRIPTION OF SERVICES Custom Business Services (continued) **RECD APR 18 2003** 3.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year (continued) (B) The Customer may subscribe to Business Long Distance 50 Connections 2 Plus Service I Year for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.42 (A).3 of this Tariff. (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS. (D) The Customer's usage rate for each call is based a 1-year term plan. (E) Outbound calls, TFS calls, and fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.42 of this Tariff. CANCELLED FEB 2 8 2004 VISP25194.129 N

Issued: April 18, 2003

3.7

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission



### 3.7 Custom Business Services (continued)

3.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year<sup>1</sup> (continued)

С

(F) If the Customer fails to maintain the required products or services described in Section 3.7.42 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 50 Connections 2 Plus Service 1 Year and will be moved to Business Long Distance 50 for an additional 1-year term, unless the Customer selects an alternative Service. If the Customer is moved to Business Long Distance 50, the rates and charges in Section 4.7.9 of the Tariff will apply in lieu of the rates and charges in Section 4.7.42 of this Tariff.

If the Customer is moved to Business Long Distance 50 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 50 Connections 2 Plus Service 1 Year, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 6.28 of this Tariff.

At the end of the 1-year term, the Customer will be moved to Business Long Distance 50 Connections 2 Service for an additional 1-year term, as described in Section 3.7.22 of this tariff, unless otherwise specified by the Customer.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

N N

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Ist Revised Sheet 194.130 Replacing Original Sheet 194.130 Wissouri Public

### SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

REC'D JUL 2 5 2003

С

С

3.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year (continued)

(F) If the Customer fails to maintain the required products or services described in Section 3.7.42 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 50 Connections 2 Plus Service 1 Year and will be moved to Business Long Distance 50 for an additional 1-year term, unless the Customer selects an alternative Service. If the Customer is moved to Business Long Distance 50, the rates and charges in Section 4.7.9 of the Tariff will apply in lieu of the rates and charges in Section 4.7.42 of this Tariff.

If the Customer is moved to Business Long Distance 50 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 50 Connections 2 Plus Service 1 Year, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 6.28 of this Tariff.

At the end of the 1-year term, the Customer will be moved to Business Long Distance 50 Connections 2 Service for an additional 1-year term, as described in Section 3.7.22 of this tariff, unless otherwise specified by the Customer.

CANCELLED

\$ 194.130 FEB 2 8 2004

Issued: July 25, 2003

Effective: August 24, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

FILED AUG 2 4 2003

Missouri Fuillic

Service Commission

Original Sheet 194.130

#### Missouri Public SECTION 3 - DESCRIPTION OF SERVICES Service Commission

3.7 Custom Business Services (continued)

## REGD APR 18 2003 N

- 3.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year (continued)
  - (F) If the Customer fails to maintain the required products or services described in Section 3.7.42 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 50 Connections 2 Plus Service 1 Year and will be moved to Business Long Distance 50 unless the Customer selects an alternative Service. If the Customer is moved to Business Long Distance 50, the rates and charges in Section 4.7.9 of the Tariff will apply in lieu of the rates and charges in Section 4.7.42 of this Tariff.

If the Customer is moved to Business Long Distance 50 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 50 Connections 2 Plus Service 1 Year, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 6.28 of this Tariff.

At the end of the 1-year term, the Customer will be moved to Business Long Distance 50 Connections 2 Service, as described in Section 3.7.22 of this tariff, unless otherwise specified by the Customer.



Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin. Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

Ν



### 3.7 Custom Business Services (continued)

3.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year<sup>1</sup>

С

- (A) Business Long Distance 100 Connections 2 Plus Service 1 Year is a custom combination switched TFS, outbound, and calling card Flat Rate Service available to Business Customers that:
  - .1 request to be provisioned under this optional Service;
  - .2 utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS; and

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations	Ν
effective April 1, 2004.	Ν

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Ν

Ν

### SECTION 3 - DESCRIPTION OF SERVICES Missourd Public

- 3.7 Custom Business Services (continued)
- **REGD APR 18** 2003
- 3.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year
  - (A) Business Long Distance 100 Connections 2 Plus Service 1 Year is a custom combination switched TFS, outbound, and calling card Flat Rate Service available to Business Customers that:
    - .1 request to be provisioned under this optional Service;
    - .2 utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS; and

# CANCELLED

FEB 2 8 2004

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public Service Commission

#### 3.7 Custom Business Services (continued)

3.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year<sup>1</sup>

С

- (A) (continued)
  - .3 subscribe to and maintain the following products or services provided by a SBC affiliate: (1) one DSL service or shared web hosting or dedicated Internet access or SBC® Yahoo!® dial-up Internet access or T1 Integrated Access or dedicated web hosting; and (2) a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business SolutionsK, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageK, Pacific Bell Instant OfficeK :, Power Office, The Business Plan, Custom Biz SaverK, Local Usage SaverK or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from a SBC Affiliate:
    - .a a minimum of one business access line, and,
    - .b an inside wire maintenance product associated with each business access line, and,
    - .c at least one instance of Caller ID, and,
    - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

N N

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

**REGD APR 18** 2003

N

### SECTION 3 - DESCRIPTION OF SERVICES Missouri Public Service Commission

- 3.7 Custom Business Services (continued)
  - 3.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year
    - (A) (continued)
      - .3 subscribe to and maintain the following products or services provided by a SBC affiliate: (1) one DSL service or shared web hosting or dedicated Internet access or SBC® Yahoo!® dial-up Internet access or T1 Integrated Access or dedicated web hosting; and (2) a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business Solutions<sup>SM</sup>, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature Package<sup>SM</sup>, Pacific Bell Instant Office<sup>SM</sup>:, Power Office, The Business Plan, Custom Biz Saver<sup>SM</sup>, Local Usage Saver<sup>SM</sup> or Ameritech Centrex Service (ACS) service package as defined in Section I of this Tariff or, subscribe to all of the following from a SBC Affiliate:
        - .a a minimum of one business access line, and,
        - .b an inside wire maintenance product associated with each business access line, and,
        - .c at least one instance of Caller ID, and,
        - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.



Ν

FEB 2 8 2004 By 6725 194.132 Ublic Service Commission MISSOURI

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

#### 3.7 Custom Business Services (continued)

3.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year<sup>1</sup>

С

- (A) (continued)
  - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.43 (A).3 of this Tariff;
  - .5 except as described below, associate the billing for the products or services of a SBC Affiliate that are described in Section 3.7.43 (A).3 of this Tariff with the Customer's BTN for Business Long Distance 100 Connections 2 Plus Service. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Long Distance 100 Connections 2 Plus Service 1 Year will be billed; and
  - .6 commit to an MMC of \$100 per month for a 1-year term

For rules and regulations regarding the MMC and term plans, see Section 6.28 of this Tariff.

This Service is established at the BTN level and is only available for a single BTN. Service is only available for Customers that commit to a 1-year term plan.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

N N

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

1st Revised Sheet 194.133 Replacing Original Sheet 194.133

### SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

REC'D JUL 2 5 2003

Service Commission

С

Missouri Public

3.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year

(A) (continued)

- .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.43 (A).3 of this Tariff;
- .5 except as described below, associate the billing for the products or C services of a SBC Affiliate that are described in Section 3.7.43 (A).3 of this Tariff with the Customer's BTN for Business Long Distance 100 Connections 2 Plus Service. Exceptions to this requirement are: (a) C Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Long Distance 100 Connections 2 Plus Service 1 Year will be billed; and

.6 commit to an MMC of \$100 per month for a 1-year term

For rules and regulations regarding the MMC and term plans, see Section 6.28 of this Tariff.

This Service is established at the BTN level and is only available for a single BTN. Service is only available for Customers that commit to a 1-year term plan.



Issued: July 25, 2003

Effective: August 24, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Miscerne manife

FILED AUG 2 4 2003

Service Commission



Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

### 3.7 Custom Business Services (continued)

3.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year<sup>1</sup> (continued)

С

- (B) The Customer may subscribe to Business Long Distance 100 Connections 2 Plus Service 1 Year for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.43 (A).3 of this Tariff.
- (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
- (D) For TFS, Customers may subscribe to TFS with or without the ability to receive intrastate Service from the Company.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations	Ν
effective April 1, 2004.	Ν

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588
Original Sheet 194.134

**REC'D APR 18** 2003

Ν

Ν

Missourl Public SECTION 3 - DESCRIPTION OF SERVICES orvice Commission

- 3.7 Custom Business Services (continued)
  - 3.7.43 Business Long Distance 100 Connections 2 Plus Service I Year (continued)
    - (B) The Customer may subscribe to Business Long Distance 100 Connections 2 Plus Service 1 Year for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.43 (A).3 of this Tariff.
    - (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
    - (D) For TFS, Customers may subscribe to TFS with or without the ability to receive intrastate Service from the Company.

CANCELLED FEB 2 8 2004 134

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public



- 3.7 Custom Business Services (continued)
  - 3.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year<sup>1</sup> (continued)
    - (E) The Customer's usage rate for each call is based a 1-year term plan.
    - (F) Outbound and TFS calls and calls billed to the Calling Card Option 2, T category 11 are billed in increments of one (1) second subject to a C/R minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.43 of this Tariff.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

Issued: March 1, 2004

Effective: April 1, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

	western Bell Co SBC Long Dist	ommunications Services, Inc. I ance	PSC Mo No. 1	1st Revised Sheet 194.135 Replacing Original Sheet 194.135
		SECTION 3 - DESCR	LIPTION OF SERV	ices Missouri Public
3.7	Custom Bu	siness Services (continued)		REC'D JAN 2 8 2004
	3.7.43 Bus	siness Long Distance 100 Con	nections 2 Plus Ser	vice 1 Year <sup>1</sup> (continued) Service Commission
	(E)	The Customer's usage rate	for each call is base	ed a 1-year term plan.
	(F)	Outbound and TFS calls a	nd calls billed to the	calling Card - Option 2 are

(initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.43 of this Tariff.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations N effective April 1, 2004. N

CANCELLED

APR 0 1 2004 200 RS 194.135 Public Service Commission MISSOURI

Issued: January 28, 2004

1

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missoufi Public Sorvice Commission

FILED FEB 28 2004

Original Sheet 194.135

		SECTION 3 - DESCRIPTION OF SERVICES	Missouri Public Servico Commicol	ien
Custon	n Busi	ness Services (continued)	<b>REG'D APR 18</b> 20	105
3.7.43	Busi	ness Long Distance 100 Connections 2 Plus Service 1 Y	(continued)	
	(E)	The Customer's usage rate for each call is based a 1-y	ear term plan.	
	(F)	Outbound and TFS calls and calls billed to the Callin billed in increments of one (1) second subject to a mi (initial period) of thirty (30) seconds. For rates and c of this Tariff.	nimum connect time	     N



Issued: April 18, 2003

3.7

ł

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missourl Public Service Commission

FILED MAY 19 2003

#### 3.7 Custom Business Services (continued)

3.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year<sup>1</sup> (continued)

С

(G) If the Customer fails to maintain the required products or services described in Section 3.7.43 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 100 Connections 2 Plus Service 1 Year and will be moved to Business Long Distance 100 for an additional 1-year term, unless the Customer selects an alternative Service. If the Customer is moved to Business Long Distance 100, the rates and charges in Section 4.7.13 of the Tariff will apply in lieu of the rates and charges in Section 4.7.43 of this Tariff.

If the Customer is moved to Business Long Distance 100 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 100 Connections 2 Plus Service 1 Year, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 6.28 of this Tariff.

At the end of the 1-year term, the Customer will be moved to Business Long Distance 100 Connections 2 Service for an additional 1-year term, as described in Section 3.7.23, unless otherwise specified by Customer.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations N effective April 1, 2004.

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

1st Revised Sheet 194.136 Replacing Original Sheet 194.136

### **SECTION 3 - DESCRIPTION OF SERVICES**

3.7 Custom Business Services (continued)

### Missouri Public

- 3.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year (continued) JUL 2 5 2003
  - (G) If the Customer fails to maintain the required products or services described in Section 3.7.43 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 100 Connections 2 Plus Service 1 Year and will be moved to Business Long Distance 100 for an additional 1-year term, unless the C Customer selects an alternative Service. If the Customer is moved to Business Long Distance 100, the rates and charges in Section 4.7.13 of the Tariff will

If the Customer is moved to Business Long Distance 100 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 100 Connections 2 Plus Service 1 Year, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 6.28 of this Tariff.

apply in lieu of the rates and charges in Section 4.7.43 of this Tariff.

At the end of the 1-year term, the Customer will be moved to Business Long Distance 100 Connections 2 Service for an additional 1-year term, as described in Section 3.7.23, unless otherwise specified by Customer.

CANCELLED

FEB 2 8 2004 2 12 5 194,136 amission

Issued: July 25, 2003

Effective: August 24, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Publis

С

FILED AUG 2 4 2003

### Service Commission

Original Sheet 194.136

N

Ν

Missouri Public

REGD APR 18 2003

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
  - 3.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year (continued)
    - (G) If the Customer fails to maintain the required products or services described in Section 3.7.43 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 100 Connections 2 Plus Service 1 Year and will be moved to Business Long Distance 100 unless the Customer selects an alternative Service. If the Customer is moved to Business Long Distance 100, the rates and charges in Section 4.7.13 of the Tariff will apply in lieu of the rates and charges in Section 4.7.43 of this Tariff.

If the Customer is moved to Business Long Distance 100 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 100 Connections 2 Plus Service 1 Year, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 6.28 of this Tariff.

At the end of the 1-year term, the Customer will be moved to Business Long Distance 100 Connections 2 Service, as described in Section 3.7.23, unless otherwise specified by Customer.



Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commissien

FILED MAY 19 2003

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

•

.

1

i

i

### **SECTION 3 - DESCRIPTION OF SERVICES**

3.7	Custom Business Services (continued)					
		1				
	3.7.44 Reserved for future use	Ν				

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

•

.

### SECTION 3 - DESCRIPTION OF SERVICES

3.7	Custom Business Services (continued)					
		1				
	3.7.45 Reserved for future use	Ν				

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

•

.

I.

I

### **SECTION 3 - DESCRIPTION OF SERVICES**

3.7	Custom Business Services (continued)					
		1				
	3.7.46 Reserved for future use	, N				

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

### 3.7 Custom Business Services (continued)

#### 3.7.47 Value Plans

- (A) The Value Plans are custom combination inbound, outbound, and calling card Flat Rate optional pricing plans. Unless otherwise specified in the description of the rate option described in Section 3.7.47 (H) of this Tariff, the Value Plans are available to new and existing Business Customers that:
  - .1 request to be provisioned under this optional pricing plan;
  - .2 utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS;
  - .3 commit to the MMC as specified in the description of the rate option selected by the Customers;
  - .4 subscribe to and maintain at least one business access line of an SBC Affiliate; and
  - .5 commit to a 1-year or 2-year term plan. For rules and regulations regarding term plans, see Section 2.26 of this Tariff.
- С
- (B) The Value Plans are established at the BTN level and are only available for a single BTN.
- (C) The Customer may subscribe to any of the Value Plans for outbound Service only, TFS only or for both outbound and TFS for a single BTN. Business Customers subscribing to the Value Plans may also subscribe to the Calling Card - Option 2, category 11 at the rates described in Section 4.7.47 of this Tariff.

Issued: April 2, 2004

Effective: May 3, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

				SECTION 3 - DESCRIPTION OF SEI	RVICES Missouri Public
3.7	Custon	n Busi	ness S	Services (continued)	RECD MÁR 01 2004
	3.7.47	Valu	e Plar	ns	Service Commissio
		(A)	card desc	Value Plans are custom combination i Flat Rate optional pricing plans. Unle cription of the rate option described in Value Plans are available to new and e	ess otherwise specified in the Section 3.7.47 (H) of this Tariff,
			.1	request to be provisioned under this	optional pricing plan;
			.2	utilize Switched Access to reach the outbound calling and/or utilize Switc the long distance network for TFS;	
			.3	commit to the MMC as specified in t selected by the Customers;	the description of the rate option
			.4	subscribe to and maintain at least on Affiliate; and	e business access line of an SBC
			.4	commit to a 1-year term plan. For reterm plans, see Section 2.26 of this 1	
		(B)		Value Plans are established at the BT ngle BTN.	N level and are only available for
Servi Busin the C			Ser Bus the	Customer may subscribe to any of the vice only, TFS only or for both outbou iness Customers subscribing to the Va Calling Card - Option 2, category 11 a 47 of this Tariff.	and and TFS for a single BTN. Hue Plans may also subscribe to at the rates described in Section T
				CANCELLE	D
				MAY 0 3 200	4.140
Issue	ed: March	1,200	4	Public Service Com MISSOURI	Effective: April 1, 2004
				Norm Descoteaux, Associate Director Re	egulatory Missouri Public fornia 94588 Service Commicei

.

1

i

I

i.

I.

T

.

i

I

FILED APR 01 2004

REC'D MAY 1 6 2003



3.7 Custom Business Services (continued)

.1

3.7.47 Value Plans

- (A) The Value Plans are custom combination inbound, outbound, and calling card Flat Rate optional pricing plans. Unless otherwise specified in the description of the rate option described in Section 3.7.47 (H) of this Tariff, the Value Plans are available to new and existing Business Customers that:
  - request to be provisioned under this optional pricing plan;
  - .2 utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS;
  - .3 commit to the MMC as specified in the description of the rate option selected by the Customers;
  - .4 subscribe to and maintain at least one business access line of an SBC Affiliate; and
  - .4 commit to a 1-year term plan. For rules and regulations regarding term plans, see Section 2.26 of this Tariff.
- (B) The Value Plans are established at the BTN level and are only available for a single BTN.
- (C) The Customer may subscribe to any of the Value Plans for outbound Service only, TFS only or for both outbound and TFS for a single BTN. Business Customers subscribing to the Value Plans may also subscribe to the Calling Card - Option 2 at the rates described in Section 4.7.47 of this Tariff.

Issued: May 16, 2003

Effective: June 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

Ν

FILED JUN 16 2003



Ν

Ν

#### SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

3.7.47 Value Plans (continued)

- (D) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
- (E) The Customer's usage rate for each call is based on the rate option selected by the Customer.
- (F) A description of the billing increments applicable to the rate option selected by the Customer may be found in Section 3.7.47 (H) of this Tariff.
- (G) Customers who cancel or discontinue a business access line of a SBC Affiliate or whose service is refused, cancelled or discontinued by an SBC Affiliate shall forfeit eligibility for rates under this plan. If the Customer is moved to an alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment under one of the Value Plan rate options, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

Issued: May 16, 2003

Effective: June 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

### 3.7 Custom Business Services (continued)

3.7.47 Value Plans (continued)

- (H) Rate Options
  - .1 Business Domestic Value Saver 15

Business Domestic Value Saver 15 is available to Customers that commit to an MMC of \$15 per month for a 1-year or 2-year term plan C agreement with the Company. Customers who cancel or discontinue a business access line of a SBC Affiliate or whose service is refused, cancelled or discontinued by an SBC Affiliate shall forfeit eligibility for rates under this rate option and will be moved to Business Domestic Saver 15 unless an alternative plan is selected by the Customer. Calls are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.47 (A) of this Tariff.

Issued: April 12, 2004

Effective: May 13, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

1st Revised Sheet 194.142 Replacing Original Sheet 194.142

#### **SECTION 3 - DESCRIPTION OF SERVICES**

3.7 Custom Business Services (continued)

3.7.47 Value Plans (continued)

- (H) Rate Options
  - .1 Business Domestic Value Saver 15

Business Domestic Value Saver 15 is available to Customers that commit to an MMC of \$15 per month for a 1-year term plan agreement with the Company. Customers who cancel or discontinue a business access line of a SBC Affiliate or whose service is refused, cancelled or discontinued by an SBC Affiliate shall forfeit eligibility for rates under this rate option and will be moved to Business Domestic Saver 15 unless an alternative plan is selected by the Customer. Calls are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.47 (A) of this Tariff.

CANCELLED MAY 13 2004 MAY 13 2004 MAY 13 2004 MAY 142 MAY 13 2004 MAY 142 MAY 13 2004 MAY 142 MAY 142 MAY 13 2004 MAY 142 MAY 143 MAY 14

Issued: March 10, 2004

Effective: April 10, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Missouri Public Service Commission

FILED APR 10 2004

# Missouri Public

REC'D MAR 1 0 2004

Service Commission

С

RECTI MAY 1 6 2003

## SECTION 3 - DESCRIPTION OF SERVICES OF SERVICES

3.7 Custom Business Services (continued)

3.7.47 Value Plans (continued)

- (H) Rate Options
  - .1 Business Domestic Value Saver 15

Business Domestic Value Saver 15 is available to Customers that commit to an MMC of \$15 per month for a 1-year term plan agreement with the Company. Customers who cancel or discontinue a business access line of a SBC Affiliate or whose service is refused, cancelled or discontinued by an SBC Affiliate shall forfeit eligibility for rates under this rate option and will be moved to Business Domestic Saver 15 unless an alternative plan is selected by the Customer. Calls are billed in increments of six (6) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.47 (A) of this Tariff.

### CANCELLED

Ν

Public S ice Commission

Issued: May 16, 2003

Effective: June 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public Service Commiscien

FILED JUN 1 6 2003

### 3.7 Custom Business Services (continued)

3.7.47 Value Plans (continued)

- (H) Rate Options (continued)
  - .2 Business Long Distance Value 50

Business Long Distance Value 50 is available to Customers that commit to an MMC of \$50 per month for a 1-year or 2-year term plan C agreement with the Company. Customers who cancel or discontinue a business access line of a SBC Affiliate or whose service is refused, cancelled or discontinued by an SBC Affiliate shall forfeit eligibility for rates under this rate option and will be moved to Business Long Distance 50 unless an alternative plan is selected by the Customer. Calls are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds.For rates and charges, see Section 4.7.47 (B) of this Tariff.

Issued: April 2, 2004

Effective: May 3, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

SECTION 3 - DESCRIPTION OF SERVICES 3.7 Custom Business Services (continued) 3.7.47 Value Plans (continued) (H) Rate Options (continued) .2 Business Long Distance Value 50

> Business Long Distance Value 50 is available to Customers that commit to an MMC of \$50 per month for a 1-year term plan agreement with the Company. Customers who cancel or discontinue a business access line of a SBC Affiliate or whose service is refused, cancelled or discontinued by an SBC Affiliate shall forfeit eligibility for rates under this rate option and will be moved to Business Long Distance 50 unless an alternative plan is selected by the Customer. Calls are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds.For rates and charges, see Section 4.7.47 (B) of this Tariff.

### CANCELLED

MAY 0 3 2004 HISHAS 194,143 PUDIAC Service Commission MISSOURI

Issued: May 16, 2003

Effective: June 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public Sorvico Commicelon

FILED JUN 1 6 2003

Ν

#### 3.7 Custom Business Services (continued)

3.7.47 Value Plans (continued)

- (H) Rate Options (continued)
  - .3 Business Long Distance Value 100

Business Long Distance Value 100 is available to Customers that commit to an MMC of \$100 per month for a 1-year or 2-year term plan agreement with the Company. Customers who cancel or discontinue a business access line of a SBC Affiliate or whose service is refused, cancelled or discontinued by an SBC Affiliate shall forfeit eligibility for rates under this rate option and will be moved to Business Long Distance 100 unless an alternative plan is selected by the Customer. Calls are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.47 (C) of this Tariff.

Issued: April 2, 2004

Effective: May 3, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

				SECTION 3 - DESCRIPTION OF SERVICES	Missouri I Servico Con	Public	า
3.7	Custon	n Busi	ness S	ervices (continued)	REC'D MAY		Ν
	3.7.47	Valu	e Plan	s (continued)			1
		(H)	Rate	Options (continued)			
			.3	Business Long Distance Value 100			1
				Business Long Distance Value 100 is availabl			
				commit to an MMC of \$100 per month for a with the Company. Customers who cancel or	• •	-	ļ
				access line of a SBC Affiliate or whose servic	e is refused, cance	lled or	
				discontinued by an SBC Affiliate shall forfeit	eligibility for rates	s under	1
				this rate option and will be moved to Business	Long Distance 1	00 unless	l
				on alternative plan is calested by the Custome	- Calla are billed	:	1

an alternative plan is selected by the Customer. Calls are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.47 (C) of this Tariff.

### CANCELLED

MAY 0 3 2004 SHRS194.44 Public Service Commission MISSOURI

Issued: May 16, 2003

Effective: June 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FILED JUN 1 6 2003

Ν

- 3.7 Custom Business Services (continued)
  - 3.7.48 Business Unlimited Long Distance Plans
    - (A) Business Unlimited Long Distance Plans are custom combination switched TFS, outbound, and calling card optional pricing plans. For a monthly recurring charge, the Customer receives unlimited intrastate and interstate one plus (1+) Direct-Dialed MOU. Switched TFS calls and calls billed to a calling card are billed on a usage sensitive basis. For rates and charges, see Section 4.7.48 of this Tariff.
    - (B) Business Unlimited Long Distance Plans are available to new and existing Business Customers that:
      - .1 request to be provisioned under this optional pricing plan;
      - .2 utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS;
      - .3 subscribe to and maintain or currently subscribe to and maintain at least one but not more than ten (10) business access lines of an SBC Affiliate C that are associated with the qualifying BTN;

Issued: September 12, 2003

Effective: October 13, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

### Missouri Publie Service Commission SECTION 3 - DESCRIPTION OF SERVICES REAT JUN 12 2003 3.7 Custom Business Services (continued) 3.7.48 Business Unlimited Long Distance Plans Business Unlimited Long Distance Plans are custom combination switched (A) TFS, outbound, and calling card optional pricing plans. For a monthly recurring charge, the Customer receives unlimited intrastate and interstate one plus (1+) Direct-Dialed MOU. Switched TFS calls and calls billed to a calling card are billed on a usage sensitive basis. For rates and charges, see Section 4.7.48 of this Tariff. Business Unlimited Long Distance Plans are available to new and existing **(B)** Business Customers that: request to be provisioned under this optional pricing plan; .1 .2 utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS; subscribe to and maintain or currently subscribe to and maintain at least .3

one but not more than five business access lines of an SBC Affiliate that

### CANCELLED

Ν

Issued: June 12, 2003

Effective: July 17, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

are associated with the qualifying BTN;

Missouri Public Servisé Germission

FILED JUL 17 2003

- 3.7 Custom Business Services (continued)
  - 3.7.48 Business Unlimited Long Distance Plans (continued)
    - (B) (continued)
      - .4 subscribe to and maintain or currently subscribe to and maintain: a business access line with a SBC Affiliate.

Issued: June 15, 2004

Effective: July 16, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 C/D

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance 2nd Revised Sheet 194.146 Replacing 1st Revised Sheet 194.146

### SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

Missouri Public

3.7.48 Business Unlimited Long Distance Plans (continued)

Service Commission

FILED SEP 1 5 2003

(B) (continued)

.4

subscribe to and maintain or currently subscribe to and maintain: a business access line with term and Caller ID of a SBC Affiliate or, a business access line (with or without a term commitment) combined with any service or product of a SBC Affiliate listed below: (a) SimpleLink<sup>SM</sup> or (b) Pacific Bell Instant Office<sup>SM</sup> or (c) Business Solutions<sup>SM</sup> or (d) one Measured Rate Service plus The Works® or (e) one Measured Rate Service plus PowerPack or (f) one Measured Rate Service plus Business Preferred<sup>SM</sup> or (g) Power Office<sup>SM</sup> or (h) Centrex Service (1 to 10 stations lines only) or (i) multiline plus Caller ID and series hunting or (j) С Custom BizSaver<sup>SM</sup> or (k) Local Usage Saver<sup>SM</sup> or (l) Caller ID and any C two of the following features on a single business line: Call Waiting, Call Waiting ID, Call Forwarding, Busy Call Forwarding, Busy Call Forwarding-Extended, Delayed Call Forwarding, Select Call Forwarding, Remote Access Call Forwarding, Repeat Dialing<sup>™</sup>, Call Screen, Speed Calling 8<sup>™</sup>, 30 Member Speed Calling, Call Return, Three Way Calling, Priority Ringing<sup>™</sup>, PRIVACY MANAGER®, Call Transfer Disconnect, Auto Redial<sup>™</sup> or Priority Call or (m) Business Essentials<sup>™</sup>; and

### CANCELLED

JUL 1 6 2004 rd RS 194.146 Commission

Issued: September 15, 2003

Effective: October 15, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Servico Gemmicolen

Т

FILED OCT 15 2003

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance 1st Revised Sheet 194.146 Replacing Original Sheet 194.146

### SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

3.7.48 Business Unlimited Long Distance Plans (continued)

(B) (continued)

.4 subscribe to and maintain or currently subscribe to and maintain: a business access line with term and Caller ID of a SBC Affiliate or, a T/C business access line (with or without a term commitment) combined with С any service or product of a SBC Affiliate listed below: (a) SimpleLink<sup>SM</sup> C/T or (b) Pacific Bell Instant Office<sup>SM</sup> or (c) Business Solutions<sup>SM</sup> or (d) one Т Measured Rate Service plus The Works® or (e) one Measured Rate Service plus PowerPack or (f) one Measured Rate Service plus Business Preferred<sup>SM</sup> or (g) Power Office<sup>SM</sup> or (h) Centrex Service (1 to 5 stations lines only) or (i) multiline plus Caller ID and series hunting or (i) Caller ID and any two of the following features on a single business line: Call Waiting, Call Waiting ID, Call Forwarding, Busy Call Forwarding, Busy Call Forwarding-Extended, Delayed Call Forwarding, Select Call Forwarding, Remote Access Call Forwarding, Repeat Dialing<sup>™</sup>, Call Screen, Speed Calling 8<sup>™</sup>, 30 Member Speed Calling, Call Return, Three Way Calling, Priority Ringing<sup>™</sup>, PRIVACY MANAGER®, Call Transfer Disconnect, Auto Redial™ or Priority Call or (k) Business Essentials<sup>SM</sup>; and Т

CANCELLED

1 5 2003 Public Service Commission

Issued: July 18, 2003

Effective: August 17, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missourl Public Service Commission

FILED AUG 17 2003

Missouri Public Service Commissic.

RECTI JUL 18 20L

Misseuri Public Service Commission

REAT JUN 12 2013

#### SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
  - 3.7.48 Business Unlimited Long Distance Plans (continued)
    - (B) (continued)
      - .4 subscribe to and maintain or currently subscribe to and maintain: a business access line of a SBC Affiliate plus any feature package of a SBC Affiliate listed below or a business access line with term of a SBC Affiliate plus any feature package of a SBC Affiliate listed below: (a) SimpleLink<sup>™</sup> or (b) Pacific Bell Instant Office<sup>™</sup> or Business Essentials<sup>SM</sup> or (c) Business Solutions<sup>SM</sup> or (d) one Measured Rate Service plus The Works® or (e) one Measured Rate Service plus PowerPack or (f) one Measured Rate Service plus Business Preferred<sup>SM</sup> or (g) Power Office<sup>SM</sup> or (h) Centrex Service (1 to 5 stations lines only) or (i) multiline plus Caller ID and series hunting or (j) Caller ID and any two of the following features on a single business line: Call Waiting, Call Waiting ID, Call Forwarding, Busy Call Forwarding, Busy Call Forwarding-Extended, Delayed Call Forwarding, Select Call Forwarding, Remote Access Call Forwarding, Repeat Dialing<sup>™</sup>, Call Screen, Speed Calling 8<sup>™</sup>, 30 Member Speed Calling, Call Return, Three Way Calling, Priority Ringing<sup>™</sup>, PRIVACY MANAGER®, Call Transfer Disconnect, Auto Redial<sup>™</sup> or Priority Call; and

### CANCELLED

AUG 1 7 2003 L, I<sup>ST</sup> RS 194.146 Public Service Commission MISSOURI

Issued: June 12, 2003

Effective: July 17, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

Ν

FILED JUL 17 2003

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

### SECTION 3 - DESCRIPTION OF SERVICES

Custon	stom Business Services (continued)		N
	.7.48 Business Unlimited Long Distance Plans (continued)	1	
3.7.48		iness Unlimited Long Distance Plans (continued)	1
			1
	(B)	(continued)	
			1
		.5 Reserved for future use.	N
		3.7.48 Busi	

Issued: June 12, 2003

Effective: July 17, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Ν

Ν

#### SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

3.7.48 Business Unlimited Long Distance Plans (continued)

- (B) (continued)
  - .6 commit to a 1-year term plan for the Business Unlimited Long Distance Plan. If the Customer discontinues Service prior to the expiration of the Business Unlimited Long Distance term plan agreement, the early termination fee applies pursuant to Section 2.26 of this Tariff. The Customer may upgrade, or downgrade, to another Business Unlimited Long Distance Plan; and the Company will waive the early termination fee associated with the change in plans if the Customer commits to another 1-year term and the number of business access lines match the Business Unlimited Long Distance Plan described in Section 3.7.48 (l) of this Tariff.

Issued: June 12, 2003

Effective: July 17, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

#### 2nd Revised Sheet 194.149 Replacing 1st Revised Sheet 194.149

### SECTION 3 - DESCRIPTION OF SERVICES

### 3.7 Custom Business Services (continued)

- 3.7.48 Business Unlimited Long Distance Plans (continued)
  - (C) Except for Customers subscribing to one of the Company's High Volume
     Calling Plans, the Company will waive the early termination fee for existing
     T Business Customers cancelling their existing term plan agreement and
     committing to a 1-Year term plan for Business Unlimited Long Distance Plans.
  - (D) Business Unlimited Long Distance Plans are only available for a single BTN. A Customer as a single business entity with more than one BTN at that business entity's service location is not eligible for Business Unlimited Long Distance Plans. Exceptions to this requirements are:
    - .1 when subscription to an additional BTN is required for technical reasons by the SBC Affiliated LEC or Affiliated CLEC for DSL provisioning. The DSL service billed under the second BTN must be physically located at the same business premises as the original local service.
  - (E) The Customer may subscribe to Business Unlimited Long Distance Plans for outbound Service only or for both outbound and TFS for a single BTN. Business Customers subscribing to the Business Unlimited Long Distance Plans may also subscribe to the Calling Card - Option 2 at the rates described in Section 4.7.48 of this Tariff.

Issued: December 1, 2003

Effective: December 31, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED Missouri Public Service Commission

T/N

Ν

N

1st Revised Sheet 194.149 Replacing Original Sheet 194.149

			SECTION 3 - DESCRIPTION OF SERVICES	Missouri Futuit	
3.7	Custon	n Busi	ness Services (continued)	RECTO OCT 2 9 200	3
	3.7.48	Busi	ness Unlimited Long Distance Plans (continued)	Service Commissi	Ion
		(C)	Except for Customers subscribing to one of the Com Calling Plans, the Company will waive the early term Customers cancelling their existing term plan agreen Year term plan for Business Unlimited Long Distance	nination fee for Business ment and committing to a 1-	
		(D)	Business Unlimited Long Distance Plans are only av A Customer as a single business entity with more that business entity's service location is not eligible for B Distance Service.	an one BTN at that N	
		(E)	The Customer may subscribe to Business Unlimited	•	

outbound Service only or for both outbound and TFS for a single BTN.
Business Customers subscribing to the Business Unlimited Long Distance
Plans may also subscribe to the Calling Card - Option 2 at the rates described in
Section 4.7.48 of this Tariff.

### CANCELLED

DEC 3 1 2003 ₿¥ Public 8 annission

Issued: October 29, 2003

Effective: December 1, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Missouri Public Service Commission

FILED DEC 01 2003

Missouri Public Wing Commission

RECT JUN 12 2003

N

### SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
  - 3.7.48 Business Unlimited Long Distance Plans (continued)
    - (C) Except for Customers subscribing to one of the Company's High Volume Calling Plans, the Company will waive the early termination fee for Business Customers cancelling their existing term plan agreement and committing to a 1-Year term plan for Business Unlimited Long Distance Plans.
    - (D) Business Unlimited Long Distance Plans are only available for a single BTN.
    - (E) The Customer may subscribe to Business Unlimited Long Distance Plans for outbound Service only or for both outbound and TFS for a single BTN. Business Customers subscribing to the Business Unlimited Long Distance Plans may also subscribe to the Calling Card - Option 2 at the rates described in Section 4.7.48 of this Tariff.



Issued: June 12, 2003

Effective: July 17, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FILED JUL 17 2003

### **SECTION 3 - DESCRIPTION OF SERVICES** 3.7 Custom Business Services (continued) N 3.7.48 Business Unlimited Long Distance Plans (continued) **(F)** Customers may subscribe to the Business Unlimited Long Distance Plans for the provision of interstate and intrastate InterLATA service; interstate, intrastate InterLATA, and intrastate IntraLATA service; or intrastate IntraLATA service only. (G) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS. (H) For switched TFS calls and calls billed to the Calling Card - Option 2, calls are billed in increments of six (6) second subject to a minimum connect time (initial period) of thirty (30) seconds. Ν

Issued: June 12, 2003

Effective: July 17, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

- 3.7 Custom Business Services (continued)
  - 3.7.48 Business Unlimited Long Distance Plans (continued)
    - Customers with more than ten (10) business access lines are not eligible for this C plan. All business access lines under a participating BTN must be provisioned on this plan.

Customers who cancel or discontinue any of the qualifying products, services or features or whose Service is refused, cancelled or discontinued by the Company or those companies listed in Section 3.4.48 (B) .4 or .5 shall forfeit eligibility for rates under this Service. Customers continuing to presubscribe to the Company will be moved to Long Distance for Business.

Issued: September 12, 2003

Effective: October 13, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Original Sheet 194.151

### SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
  - 3.7.48 Business Unlimited Long Distance Plans (continued)
    - Customers with more than five (5) business access lines are not eligible for this plan. All business access lines under a participating BTN must be provisioned on this plan.

Customers who cancel or discontinue any of the qualifying products, services or features or whose Service is refused, cancelled or discontinued by the Company or those companies listed in Section 3.4.48 (B) .4 or .5 shall forfeit eligibility for rates under this Service. Customers continuing to presubscribe to the Company will be moved to Long Distance for Business.

### CANCELLED

OCT 1 3 2003 St RS 194.151 Public Service Commission MISSOURI

Issued: June 12, 2003

Effective: July 17, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission FILED JUL 17 2003

## RECT) JUN 12 2003

Ν

Missourt Public

**Bervice Commission** 

3.7 Custom Business Ser	vices (continued)
-------------------------	-------------------

- 3.7.48 Business Unlimited Long Distance Plans (continued)
  - (J) Every WTN within the participating BTN must be provisioned on this Service. Customers found in violation shall forfeit eligibility for rates under this plan and will be moved to Long Distance for Business unless an alternative plan is selected by the Customer. Early termination charges may apply as described in Section 2.26 of this Tariff. If the Customer is moved to an alternative Service and the Customer's term plan commitment is equal to or greater than the term plan commitment under Business Unlimited Long Distance Plans, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

Ν

Ν

Issued: June 12, 2003

Effective: July 17, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588
С

#### SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

з

- 3.7.48 Business Unlimited Long Distance Plans (continued)
  - (K) Certain restrictions apply. Business Unlimited Long Distance Plans may not be used in conjunction with the following: auto dialers, long distance Internet access, long distance Intranet access, call center applications including but not limited to autodialers, PBX trunks, ground start line or trunks, ISDN services, foreign exchange services, public telephone services, public access smart-pay phones, analog to digital conversion digital PBX services, WATS services, PBX/PABX/EABX services, non-square electronic key telephone systems, hybrid key telephone systems, predictive calling/dialing systems, automatic outbound dialing systems, any type of automatic call distribution system, or the functional equivalent of any such system listed above. If the Company determines that the Customer is in violation of above listed restrictions, the Customer shall forfeit eligibility for rates under this plan and will be moved to Long Distance for Business unless an alternative plan is selected by the Customer.

Issued: September 15, 2003

Effective: October 15, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Original Sheet 194.153 Missouri Public Service Commission

#### SECTION 3 - DESCRIPTION OF SERVICES

# RECT) JUN 12 2003

- 3.7 Custom Business Services (continued)
  - 3.7.48 Business Unlimited Long Distance Plans (continued)
    - (K) Certain restrictions apply. Business Unlimited Long Distance Plans may not be used in conjunction with the following: auto dialers, long distance Internet access, call center applications including but not limited to autodialers, PBX trunks, ground start line or trunks, ISDN services, foreign exchange services, public telephone services, public access smart-pay phones, analog to digital conversion digital PBX services, WATS services, PBX/PABX/EABX
      services, non-square electronic key telephone systems, hybrid key telephone systems, predictive calling/dialing systems, automatic outbound dialing systems, any type of automatic call distribution system, or the functional equivalent of any such system listed above. If the Company determines that the Customer is in violation of above listed restrictions, the Customer shall forfeit eligibility for rates under this plan and will be moved to Long Distance for Business unless an alternative plan is selected by the Customer.

# CANCELLED

OCT 1 5 2003 EV ST RS 194.153 Public Service Commission MISSOURI

Issued: June 12, 2003

Effective: July 17, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Publie Service Cemmicsien



- 3.7 Custom Business Services (continued)
  - 3.7.49 Business Long Distance Solutions
    - (A) Business Long Distance Solutions is a custom combination of switched TFS, outbound, and calling card Flat Rate optional pricing plan available to Business Customers that:
      - .1 request to be provisioned under this optional pricing plan;
      - .2 utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for switched TFS; and
      - .3 commit to a 1-year or 2-year term plan

For rules and regulations regarding the MMC and term plans, see Section 2.26 of this Tariff.

This Service is established at the BTN level and is only available for a single BTN/BAN.

(B) The Customer may subscribe to Business Long Distance Solutions for outbound Service only, switched TFS only or for both outbound and switched TFS for a single BTN/BAN. Customers may also subscribe to Calling Card – Option 2, category 11.

Issued: April 2, 2004

Effective: May 3, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

	stern Bell Com C Long Distan	•	C Mo No. 1 Original Sheet 194.154
		SECTION 3 - DESCRIPT	ION OF SERVICES Missouri Public
3.7 (	Custom Busin	ess Services (continued)	RECD FEB 23 2004
3	3.7.49 Busine	ss Long Distance Solutions	Service Commissior
	(A)	-	ations is a custom combination of switched ard Flat Rate optional pricing plan available
		.I request to be provisio	ned under this optional pricing plan;
		outbound calling and	ess to reach the long distance network for for utilize Switched Access to receive calls e network for switched TFS; and
		.3 commit to a 1-year	
NCELIEN		For rules and regulations reg 2.26 of this Tariff.	arding the MMC and term plans, see Section
IAY 032	004 04.154 000000000000000000000000000000000000	This Service is established at single BTN/BAN.	the BTN level and is only available for a
MISSO	(B)	outbound Service only, swite	to Business Long Distance Solutions for thed TFS only or for both outbound and TN/BAN. Customers may also subscribe to egory 11.

Issued: February 23, 2004

Public

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commicsion

FILED APR 01 2004

3.7	Custom Busin	ess Services (continued)	N
	3.7.49 Busine	ess Long Distance Solutions (continued)	 
	(C)	Switched TFS calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding switched TFS.	
	(D)	The Customer's usage rate is based on the MMC and the length of the term plan	
	(E)	Calls are billed in increments of one (1) seconds subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.49 of this Tariff.	     N

Issued: February 23, 2004

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Custom Bus	siness Se	rvices (	continued)	N
3.7.49	Busir	ness Lor	ng Distance Solutions (continued)	
	(F)	Rate	Options:	
		.1	Business Long Distance Solutions 15	
			In addition to the requirements in Section 3.7.49 (A) of this Tariff, Customers or Applicants subscribing to Business Long Distance Solutions 15 must commit to an MMC of \$15.	
			At the end of the initial term, the Customer will be moved to Business Domestic Saver 15 for the same term length as the original term, as described in Section 3.7.11 of this Tariff, unless otherwise specified by the Customer.	
		.2	Business Long Distance Solutions 50	
			In addition to the requirements in Section 3.7.49 (A) of this Tariff, Customers or Applicants subscribing to Business Long Distance Solutions 50 must commit to an MMC of \$50.	
			At the end of the initial term, the Customer will be moved to Business Long Distance 50 for the same term length as the original plan, as described in Section 3.7.9 of this Tariff, unless otherwise specified by the Customer.	     N
		3.7.49 Busir	3.7.49 Business Lor (F) Rate .1	<ul> <li>(F) Rate Options: <ol> <li>Business Long Distance Solutions 15</li> <li>In addition to the requirements in Section 3.7.49 (A) of this Tariff, Customers or Applicants subscribing to Business Long Distance Solutions 15 must commit to an MMC of \$15.</li> <li>At the end of the initial term, the Customer will be moved to Business Domestic Saver 15 for the same term length as the original term, as described in Section 3.7.11 of this Tariff, unless otherwise specified by the Customer.</li> <li>Business Long Distance Solutions 50</li> <li>In addition to the requirements in Section 3.7.49 (A) of this Tariff, Customers or Applicants subscribing to Business Long Distance Solutions 50</li> <li>In addition to the requirements in Section 3.7.49 (A) of this Tariff, Customers or Applicants subscribing to Business Long Distance Solutions 50 must commit to an MMC of \$50.</li> <li>At the end of the initial term, the Customer will be moved to Business Long Distance 50 for the same term length as the original plan, as described in Section 3.7.9 of this</li> </ol></li></ul>

Issued: February 23, 2004

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

3.7	Custom Busin	ness Sei	vices (c	continued)	N
	3.7.49	Busin	ess Lon	g Distance Solutions (continued)	
		(F)	Rate C	Options: (continued)	
			.3	Business Long Distance Solutions 100	
				In addition to the requirements in Section 3.7.49 (A) of this Tariff, Customers or Applicants subscribing to Business Long Distance Solutions 100 must commit to an MMC of \$100.	
				At the end of the initial term, the Customer will be moved to Business Long Distance 100 for the same term length as the original term, as described in Section 3.7.13 of this Tariff, unless otherwise specified by the Customer.	     N

Issued: February 23, 2004

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Ν

Ν

#### SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

3.7.50 Business Block of Time 5000

(A) Business Block of Time 5000 is a custom combination Business Optional Calling Plan for outbound, switched Toll Free Service, and Calling Card long distance calling. This Business Optional Calling Plan is established at the BTN level. Multiple BTN aggregation is not available with this Service. If the Customer or Applicant selects a different Business Optional Calling Plan for specific WTN(s), the Customer or Applicant is required to establish a separate BTN for each variation.

Outbound calls, TFS calls, and fully automated, Direct-Dialed calls billed to the Calling Card – Option 2, Category 11 are billed in increments of six (6) seconds subject to a minimum connect time (initial period) of eighteen (18) seconds.

Issued: December 10, 2004

Effective: January 10, 2005

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Ν

#### SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
  - 3.7.50 Business Block of Time 5000 (continued)
    - (B) For a specific MRC, the Customer receives a 5000 minute block of Direct-Dialed outbound, switched Toll Free Service, and Calling Card – Option 2, Category 11 interstate and/or intrastate MOU for all lines under the BTN that is presubscribed to the Company under this plan. The block of time may be used (1) for placing one plus (1+) Direct-Dialed outbound calls that originate from a line presubscribed to the Company and (2) for receiving Toll Free Services calls on the same presubscribed line. Direct-Dialed U.S. to International calls, Canada Toll Free Services calls and any calls with International origination or termination billed to the Calling Card – Option 2, Category 11, are not included in the Block of Time. All other MOU billed to Calling Card – Option 2 category 11 are included in the block.

The Customer is not required to sign a term plan agreement and no early termination fees will be assessed for the Business Block of Time 5000 plan.

All usage in excess of the 5000 minute block of time will be billed at a fixed rate per minute. See Section 4.7.51 of this Tariff for the per minute rate after the block of time has been used. Any minutes not used in a billing cycle will not be carried over to the next billing cycle. No credit will be given for any unused minutes.

(C) When ordering Service, the Customer or Applicant must specify if Business Block of Time 5000 is to be used for outgoing calls only, Toll Free Services, or both.

Ν

Issued: December 10, 2004

Effective: January 10, 2005

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

3.7	Custon	n Busi	ness Services (continued)	N
	3.7.50	Busi	ness Block of Time 5000 (continued)	
		(D)	The Business Block of Time 5000 is available to new and existing Customers or Applicants that:	
			.1 requests to be provisioned under the Business Block of Time 5000 plan for the purpose of placing interstate and intrastate long distance calls;	
			.2 utilize Switched Access to receive calls from the long distance network for Toll Free Services and/or to reach the long distance network for outbound calling;	
			.3 subscribe to no less than three (3) and no more than five (5) local business Access Lines under a term plan agreement from an Affiliated LEC or Affiliated CLEC.	
		(E)	The start of Service date for Business Block of Time 5000 may be on or after the installation date of the local business Access Line(s) as defined in Section 3.7.51 (D) .3 of this Tariff. The Company may require up to sixty (60) calendar days from Customer's request to provision the Block of Time 5000 Business Optional Calling Plan on the Customer's account.	
		(F)	Changes to the Business Block of Time 5000 plan will be effective on the day the Customer's change order is processed. If an existing Customer initially subscribes to the Business Block of Time 5000 in the middle of its billing cycle, the change will be effective on the first day after the Customer's change order is processed.	       N

Issued: December 10, 2004

Effective: January 10, 2005

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Ν

Ν

#### SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)
--

- 3.7.50 Business Block of Time 5000 (continued)
  - (G) If a Customer fails to maintain the products, services or features described in Section 3.7.50 (D) .3 of this Tariff that initially qualified the Customer for Business Block of Time 5000, the Customer will no longer qualify for Business Block of Time 5000. Unless the Customer selects an alternative Business Optional Calling Plan, the Customer will be moved to Long Distance for Business and the rates and charges in Section 4.7.7 of this Tariff will apply in lieu of the rates and charges in Section 4.7.50 of this Tariff.

Issued: December 10, 2004

Effective: January 10, 2005

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

PSC Mo. - No. 1

- 3.8 Custom Consumer Services
  - 3.8.1 General

For outbound Services provided via a Switched Access arrangement, Residential Customers may subscribe to any of the Company's outbound Service offerings for the provision of (1) intrastate InterLATA and intrastate IntraLATA calling; (2) intrastate InterLATA calling only and select another company for the provision of the Customer's intrastate IntraLATA calling; or (3) intrastate IntraLATA calling only and select another company for the provision of the Customer's intrastate InterLATA calling.

- 3.8.2 Block of Time: 300 Minutes
  - (A) Block of Time: 300 Minutes is a custom combination outbound and calling card long distance optional pricing plan. This optional calling plan is available to new and existing Residential Customers that use Switched Access to reach the long distance network and request to be provisioned under this optional pricing plan.
  - (B) Customers or End Users can access the Service by dialing 1 + the area code + the called telephone number.

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

RECD MAR 07 2001 Original Sheet 195 Service Commission

#### SECTION 3 - DESCRIPTION OF SERVICES

3.8 Custom Consumer Services

#### 3.8.1 General

For outbound Services provided via a Switched Access arrangement, Residential Customers may subscribe to any of the Company's outbound Service offerings for the provision of (1) intrastate InterLATA and intrastate IntraLATA calling; (2) intrastate InterLATA calling only and select another company for the provision of the Customer's intrastate IntraLATA calling; or (3) intrastate IntraLATA calling only and select another company for the provision of the Customers's intrastate IntraLATA calling; or (3) intrastate IntraLATA calling only and select another company for the provision of the Customers's intrastate IntraLATA calling.

- 3.8.2 Block of Time: 300 Minutes
  - (A) Block of Time: 300 Minutes is a custom combination outbound and calling card long distance optional pricing plan. This optional calling plan is available to new and existing Residential Customers that use Switched Access to reach the long distance network and request to be provisioned under this optional pricing plan.
  - (B) Customers or End Users can access the Service by dialing 1 + the area code + the called telephone number.

CANCELLED

FEB 07 2003 HARS 195 Public Service Commission MISSOUPI

Issued: March 7, 2001

Effective

Norm Descoteaux, Regulatory Manager DEC 0 7 2001 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

> FILED DEC 072001 01-475 Service Commission

- 3.8 Custom Consumer Services
  - 3.8.2 Block of Time: 300 Minutes
    - (C) For a specified monthly recurring charge, the Customer receives a specific amount (block) of time for placing one plus (1+) Direct-Dialed outbound calls that originate from a line presubscribed to the Company. For calling card calls billed to the Calling Card Option 1, the usage charges shown in Section 4.8.1 of this Tariff apply in lieu of the usage charges shown in Section 4.1.1 (B).2.b.i and Section 4.1.2 (A) of this Tariff. For per call charges, see Section 4.1.1 (B).2.a, Section 4.1.2 (B) and Section 4.1.2 (C) of this Tariff.
    - (D) For a monthly recurring charge, the Customer receives a 300 MOU (block) of intrastate and/or interstate one plus (1+) Direct-Dialed calling. All usage in excess of the selected block of time will be billed at a fixed rate per minute. See Section 4.8.1 of this Tariff for the per minute rate after the block of time has been used. The Customer may only subscribe to one block of time per BTN. Any minutes not used in a billing cycle will not be carried over to the next billing cycle. No credits will be given for any unused minutes.

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

#### 3.8 Custom Consumer Services

## Missouri Public

## REC'D DEC 07 2001

Service Commission

- 3.8.2 Block of Time: 300 Minutes
  - (C) For a specified monthly recurring charge, the Customer receives a specific T amount (block) of time for placing one plus (1+) Direct-Dialed outbound calls C that originate from a line presubscribed to the Company. For calling card calls | billed to the Proprietary Calling Card Option 1, the usage charges shown in | Section 4.8.1 of this Tariff apply in lieu of the usage charges shown in Section 4.1.1 (B).2.b.i and Section 4.1.2 (A) of this Tariff. For per call charges, see | Section 4.1.1 (B).2.a, Section 4.1.2 (B) and Section 4.1.2 (C) of this Tariff.
  - (D) For a monthly recurring charge, the Customer receives a 300 MOU (block) of intrastate and/or interstate one plus (1+) Direct-Dialed calling.

D

С

All usage in excess of the selected block of time will be billed at a fixed rate D per minute. See Section 4.8.1 of this Tariff for the per minute rate after the block of time has been used. The Customer may only subscribe to one block of time per BTN. Any minutes not used in a billing cycle will not be carried over to the next billing cycle. No credits will be given for any unused minutes.

# CANCELLED FED 07 2003 Public Service Commission

Issued: December 7, 2001

Effective: January 6, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

FILED JAN 06 2002

Missouri Public

Service Commission

### Missouri Public

RECD MAR 07 2001 Original Sheet 196 Service Commission

### SECTION 3 - DESCRIPTION OF SERVICES

3.8 Custom Consumer Services

I

- 3.8.2 Block of Time: 300 Minutes
  - (C) For a specified monthly recurring charge, the Customer receives a specific amount (block) of time for placing (1) one plus (1+) Direct-Dialed outbound calls that originate from a line presubscribed to the Company and (2) fully automated calling card calls and billing those calls to the Proprietary Calling Card Option 1. Operator Toll Assistance calls originating from presubscribed lines are not included in the block of time. Operator Toll Assistance calls, other than fully automated calls billed to the Proprietary Calling Card Option 1, are not included in the block of time. The per call charges shown in Section 4.1.1 (B).2.a of this Tariff apply to all fully automated calling card calls billed to the Proprietary Calling Card Option 1.
  - (D) For a monthly recurring charge, the Customer receives a 300 MOU (block) of intrastate and/or interstate one plus (1+) Direct-Dialed calling and usage generated from fully automated calling card calls billed to the Proprietary Calling Card Option 1. All usage in excess of the selected block of time will be billed at a fixed rate per minute. See Section 4.8.1 of this Tariff for the per minute rate after the block of time has been used. The Customer may only subscribe to one block of time per BTN. Any minutes not used in a billing cycle will not be carried over to the next billing cycle. No credits will be given for any unused minutes.



JAN 0 6 2002 By Stars 196 Public Cervice Commission MISSOULI

Issued: March 7, 2001

Effective No. 24 Pures

Norm Descoteaux, Regulatory Manager DEC 0 7 2001 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

> FILED DEC 07 2001 01-475 Service Commission

## Missouri Public

RECD MAR 07 2001 Original Sheet 197 Service Commission

#### SECTION 3 - DESCRIPTION OF SERVICES

- 3.8 Custom Consumer Services (continued)
  - 3.8.2 Block of Time: 300 Minutes (continued)
    - (E) Reserved for future use
    - (F) The block of time selected at the time the Customer's order is processed will remain in effect until cancelled or changed by the Customer. Changes to a block of time plan will be effective on the day the Customer's change order is processed. If an existing Customer initially subscribes to the Block of Time: 300 Minutes in the middle of its billing cycle, the change will be effective on the first day after the Customer's change order is processed.
    - (G) For all calls, the initial and additional periods are billed in increments of one (1) minute or a fraction thereof. This optional pricing plan is established at the BTN level. If a Customer selects a different price plan for specific WTN(s), the Customer is required to establish a separate BTN for each variation.

Issued: March 7, 2001

Effective April 20,200

Norm Descoteaux, Regulatory Manager DEC 0 7 2001 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

> FILED DEC 072001 01-475 Service Commission

## REC'D MAR 0 7 2001

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u> Original Sheet 198 Service Commission

### SECTION 3 - DESCRIPTION OF SERVICES

3.8 Custom Consumer Services (continued)

3.8.3 Reserved for future use

Issued: March 7, 2001

Effective

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588 DEC 0 7 2001 Missouri Public

FILED DEC 07 2001 01-475 Service Commission.

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance RECD MAR 0 7 2001 Original Sheet 199

Original Sheet 199 Service Commission

#### SECTION 3 - DESCRIPTION OF SERVICES

3.8 Custom Consumer Services (continued)

3.8.4 Reserved for future use

Issued: March 7, 2001

Effective to a state of the

Norm Descoteaux, Regulatory ManagerDEC 0 7 20015850 W. Las Positas Blvd., Pleasanton, California 94588Missouri Public

FILED DEC 072001

Service Commission

December 24, 2004 Missouri Public Service Commission TN-2005-0149

CANCELLED

Southwestern Bell Communications Services, Inc.PSC Mo. - No. 11 st Revised Sheet 200d/b/a SBC Long DistanceReplacing Original Sheet 200

#### SECTION 3 - DESCRIPTION OF SERVICES

## Missouri Public

3.9 Reserved for future use

# REC'D DEC 07 2001 T

Service Commission

Issued: December 7, 2001

Effective: January 6, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

٠

Missouri Public

FILED JAN 06 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Service Commission

### Missouri Public

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

RECT MAR 0 7 2001

Original Sheet 200 Service Commission

#### SECTION 3 - DESCRIPTION OF SERVICES

Reserved for future use

CANCELLED

JAN 06 2002 By ISTRS 200 Public Service Commission

Issued: March 7, 2001

Effective Apple 29, 200 M

DEC 072001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED DEC 07 2001 01-475 Service Commission

#### 3.10 Miscellaneous

#### 3.10.1 Account Codes

- (A) Account codes are an optional feature associated with outbound long distance Service that provides the Customer the ability to track usage by requiring the caller enter a string of digits ("digit string"). Account codes enable the Customer to obtain call detail from the Company which is sorted and summarized based on digits entered by the caller. The call detail is provided to the Customer without charge.
- (B) Account codes are available to Business Customers that (1) subscribe to one of the Company's outbound long distance service offerings described in Section 3.7 of this Tariff for the provision of interstate, intrastate InterLATA, and intrastate IntraLATA Service; (2) request the optional feature; and (3) are billed via a Company direct bill or a shared LEC or CLEC bill. Account codes are not available with SBC Long Distance Virtual Private Network Service, TFS or calling card Service. To subscribe to Account Codes for intrastate calling, the Customer must also subscribe to one of the Company's outbound Service offerings for the provision of interstate calling.
- (C) The length of the string digits available to a Customer is limited and the number of digits is based on whether the call originates at an On-Net or Off-Net location. The Company controls the length of the string digits. The caller will be allowed three (3) attempts to enter a valid account code before the call is terminated.
- (D) The account codes are available on a mandatory and non-mandatory basis. If the Customer subscribes to the mandatory feature, the caller must enter account codes for the call to complete. If the Customer subscribes to the non-mandatory feature, the caller may bypass entering the code by pressing the pound key (#) on the keypad.
- (E) The account codes are available on a validated and non-validated basis. If the Customer subscribes to the validated feature, the caller must enter specific account codes in order to complete the call. If the Customer subscribes to the nonvalidated feature, the caller may enter any digits as an account code as long as the string is the designated number of digits in length.

Issued: July 14, 2004

Effective: August 16, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Т

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

5th Revised Sheet 201 Replacing 4th Revised Sheet 201 VIISSOUTT UDHE

#### SECTION 3 - DESCRIPTION OF SERVICES

#### 3.10 Miscellaneous

3.10.1 Account Codes

# REC'D JUN 21 2002

Ν

Ν

- Service Commission
- (A) Account codes are an optional feature associated with outbound long distance Service that provides the Customer the ability to track usage by requiring the caller enter a string of digits ("digit string"). Account codes enable the Customer to obtain call detail from the Company which is sorted and summarized based on digits entered by the caller. The call detail is provided to the Customer without charge.
- (B) Account codes are available to Business Customers that (1) subscribe to one of the Company's outbound long distance service offerings described in Section 3.7 of this Tariff for the provision of interstate, intrastate InterLATA, and intrastate IntraLATA Service; (2) request the optional feature; and (3) are billed via a Company direct bill or a shared LEC or CLEC bill. Account codes are not available with SBC Long Distance Virtual Private Network Service, TFS or calling card Service. To subscribe to Account Codes, the Customer must also subscribe to one of the Company's outbound Service offerings for the provision of both intrastate InterLATA and intrastate IntraLATA calling.
- (C) The length of the string digits available to a Customer is limited and the number of digits is based on whether the call originates at an On-Net or Off-Net location. The Company controls the length of the string digits. The caller will be allowed three (3) attempts to enter a valid account code before the call is terminated.
- (D) The account codes are available on a mandatory and non-mandatory basis. If the Customer subscribes to the mandatory feature, the caller must enter account codes for the call to complete. If the Customer subscribes to the non-mandatory feature, the caller may bypass entering the code by pressing the pound key (#) on the keypad.
- (E) The account codes are available on a validated and non-validated basis. If the Customer subscribes to the validated feature, the caller must enter specific account codes in order to complete the call. If the Customer subscribes to the non-validated feature, the caller may enter any digits as an account code as long as the string is the designated number of digits in length.

AUG 1 6 2004 AUG 1 6 2004 By Low S 201 Effective: July 20, 2002 Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

FILED JUL 2 0 2002

Service Commission

Issued: June 20, 2002

#### Missouri Public SECTION 3 - DESCRIPTION OF SERVICES

#### 3.10 Miscellaneous

#### 3.10.1 Account Codes

### Service Commission

RFCID MAR 2 6 2002

- Account codes are an optional feature associated with outbound long distance (A) Service that provides the Customer the ability to track usage by requiring the caller enter a string of digits ("digit string"). Account codes enable the Customer to obtain call detail from the Company which is sorted and summarized based on digits entered by the caller. The call detail is provided to the Customer without charge.
- **(B)** Account codes are available to Business Customers that (1) subscribe to one of the Company's outbound long distance service offerings described in Section 3.7 of this Tariff for the provision of interstate, intrastate InterLATA, and intrastate IntraLATA Service; (2) request the optional feature; and (3) are billed via a Company direct bill or a shared LEC or CLEC bill. Account codes С are not available with SBC Long Distance Virtual Private Network Service, TFS or calling card Service.
- (C) The length of the string digits available to a Customer is limited and the number of digits is based on whether the call originates at an On-Net or Off-Net location. The Company controls the length of the string digits. The caller will be allowed three (3) attempts to enter a valid account code before the call is terminated.
- The account codes are available on a mandatory and non-mandatory basis. If (D) С the Customer subscribes to the mandatory feature, the caller must enter account codes for the call to complete. If the Customer subscribes to the non-Ν mandatory feature, the caller may bypass entering the code by pressing the pound key (#) on the keypad. Ν

The account codes are available on a validated and non-validated basis. If the Customer subscribes to the validated feature, the caller must enter specific account codes in order to complete the call. If the Customer subscribes to the inscittanon-validated feature, the caller may enter any digits as an account code as long as the string is the designated number of digits in length.

Issued: March 27, 2002

CANNELLED

2 0 2002

**(E)** 

Effective: May 1, 2002 Missouri Public

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

FILED MAY 01 2002

Service Commission

#### Missouri Public **SECTION 3 - DESCRIPTION OF SERVICES**

#### 3.10 **Miscellaneous**

#### 3.10.1 Account Codes

# Service Commission

REC'D FEB 21 2002

- (A) Account codes are an optional feature associated with outbound long distance Service that provides the Customer the ability to track usage by requiring the caller enter a string of digits ("digit string"). Account codes enable the Customer to obtain call detail from the Company which is sorted and summarized based on digits entered by the caller. The call detail is provided to the Customer without charge.
- **(B)** Account codes are available to Business Customers that (1) subscribe to one of the Company's outbound long distance service offerings described in Section 3.7 of this Tariff for the provision of interstate, intrastate InterLATA, and intrastate IntraLATA Service; (2) request the optional feature; and (3) are billed via a Company direct bill. Account codes are not available with SBC Long Distance Virtual Private Network Service, TFS or calling card Service.
- Т
- (C) The length of the string digits available to a Customer is limited and the number of digits is based on whether the call originates at an On-Net or Off-Net location. The Company controls the length of the string digits. The caller will be allowed three (3) attempts to enter a valid account code before the call is terminated.
- (D) The account codes are available on a mandatory basis. If the Customer subscribes to the mandatory feature, the caller must enter account codes for the call to complete.
- (E) The account codes are available on a validated and non-validated basis. If the Customer subscribes to the validated feature, the caller must enter specific account codes in order to complete the call. If the Customer subscribes to the non-validated feature, the caller may enter any digits as an account code as long as the string is the designated number of digits in length.

	CANCELLED	Missouri Public
	MAY 0 1 2002	FILED MAR 23 2002
	11.05.201	- Service Commission
Issued: February 21, 2002	Public Service Commission MISSOURI	Effective: March 23, 2002
	Norm Descoteaux, Regulatory Manage	

5850 W. Las Positas Blvd., Pleasanton, California 94588

#### 3.10 Miscellaneous

#### 3.10.1 Account Codes

Missouri Public REC'D DE C 2 6 2001

D Т

С

С

- Account codes are an optional feature associated with outbound long distance sion (A) Service that provides the Customer the ability to track usage by requiring the caller enter a string of digits ("digit string"). Account codes enable the Customer to obtain call detail from the Company which is sorted and summarized based on digits entered by the caller. The call detail is provided to the Customer without charge.
- Account codes are available to Business Customers that (1) subscribe to one **(B)** of the Company's outbound long distance service offerings described in Section 3.7 of this Tariff for the provision of interstate, intrastate InterLATA, and intrastate IntraLATA Service; (2) request the optional feature; and (3) are billed via a Company direct bill. Account codes are not available with VPN, TFS or calling card Serice.
- С (C) The length of the string digits available to a Customer is limited and the number of digits is based on whether the call originates at an On-Net or Off-Net location. The Company controls the length of the string digits. The caller will be allowed three (3) attempts to enter a valid account code before the call is terminated. С
- Т (D) The account codes are available on a mandatory basis. If the Customer Т subscribes to the mandatory feature, the caller must enter account codes for the call to complete.
- (E) The account codes are available on a validated and non-validated basis. If the Customer subscribes to the validated feature, the caller must enter specific account codes in order to complete the call. If the Customer subscribes to the non-validated feature, the caller may enter any digits as an account code as long as the string is the designated number of digits in length.





Effective: February 4, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED FEB 04 2002

Service Commission

1st Revised Sheet 201 Replacing Original Sheet 201

#### SECTION 3 - DESCRIPTION OF SERVICES

#### 3.10 Miscellaneous

#### 3.10.1 Account Codes

- (A) Account codes are an optional feature associated with outbound long distance Service that provides the Customer the ability to track usage by requiring the caller enter a string of digits ("digit string"). Account codes enable the Customer to obtain call detail from the Company which is sorted and summarized based on digits entered by the caller. The call detail is provided to the Customer without charge.
- (B) Account codes are available to Business Customers that (1) subscribe to one of the Company's outbound long distance service offerings described in Section 3.7 of this Tariff for the provision of interstate, intrastate InterLATA, and intrastate IntraLATA Service; (2) request the optional feature; and (3) are billed via a Company direct bill. Account codes are only available for On-Net call termination. Account codes are not available with VPN (when available).
- (C) The account codes are five (5) digits in length. Account codes of less than five
   (5) digits must be preceded by zeros. The caller will be allowed three (3) attempts to enter a valid account code before the call is terminated.
- (D) The account codes are available on a mandatory basis. The caller must enter account codes for the call to complete.
- (E) The account codes are available on a validated basis. The caller must enter specific account codes in order to complete the call.



CANCELLED

Issued: December 7, 2001

Effective: January 6, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

Ν

FILED JAN 06 2002

Service Commission

# REC'D DEC 07 2001

Service Commission

Missouri Public

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

RECD MAR 0 7 2001 Original Sheet 201

Service Commission

#### **SECTION 3 - DESCRIPTION OF SERVICES**

Reserved for future use

CANCELLED

JAN 06 2002 By ISTAS 201 Public Service Commission

Issued: March 7, 2001

Effect ..... ph. 20, 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

> FILED DEC 072001 01-475 Service Commission

DEC 0 7 2001

Missouri Public

## RECD MAR 0 7 2001

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u>

Original Sheet 202 Service Commission

1

## SECTION 3 - DESCRIPTION OF SERVICES

Reserved for future use

Issued: March 7, 2001

#### Effective April 23 200

FILED DEC 07 2001

01-475 Service Commission

Norm Descoteaux, Regulatory Manager DEC 0 7 2001 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

Original Sheet 203 Service Commission

Service Cammission

#### **SECTION 3 - DESCRIPTION OF SERVICES**

Reserved for future use

Effective Issued: March 7, 2001 DEC 072001 Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588 **Missouri Public** CANCELLED FILED DEC 072001 01-475 December 24, 2004 Missouri Public Service Commission

TN-2005-0149

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance RECD MAR 0 7 2001 Original Sheet 204

Service Commission

**SECTION 3 - DESCRIPTION OF SERVICES** 

Reserved for future use

Issued: March 7, 2001

Effective

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED DEC 072001

**Missouri Public** 

DEC 072001

Service Commission

Southwestern Bell Communications Services, Inc.	PSC Mo No. 1	1st Revised Sheet 205
d/b/a SBC Long Distance		Replacing Original Sheet 205

#### SECTION 4 - RATES AND CHARGES

4.1 Operator Toll Assistance Services

-----

•

- 4.1.1 Access Method Toll Free Access Number
  - (A) Reserved for future use

Issued: December 3, 2003

Effective: January 9, 2004

----

T/D

D

D

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

Service Commission

CD MAR 0 Ofiginal Sheet

#### SECTION 4 - RATES AND CHARGES

- 4.1 Operator Toll Assistance Services
  - 4.1.1 Access Method Toll Free Access Number
    - (A) Group 1 Toll Free Access Numbers

There are two rate elements. They include a usage charge and a per call charge. The usage charges and per call charges follow:

.1 Usage Rates

Call Type	Rate Per Minute
Fully Automated LEC Card	\$0.35
All Other Operator Toll Assistance	\$0.35
Services Calls	

.2 Person-to-Person Per Call Charge

Rate Per Call	\$4.90

Norm Descoteaux, Regulatory Manager

5850 W. Las Positas Blvd., Pleasanton, California 94588

## CANCELLED



Issued: March 7, 2001

Effective 23, 2001

DEC 0 7 2001

Missouri Public

FILED DEC 07 2001 01-475 Service Commission Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u>

٠.

•

1st Revised Sheet 206 Replacing Original Sheet 206

D

D

- -

#### **SECTION 4 - RATES AND CHARGES**

Issued: December 3, 2003

Effective: January 9, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

## Missouri Public

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance RECD MAR 07 2001 Original Sheet 206

Service Commission

#### SECTION 4 - RATES AND CHARGES

- 4.1 Operator Toll Assistance Services (continued)
  - 4.1.1 Access Method Toll Free Access Number (continued)
    - (A) Group 1 Toll Free Access Numbers (continued)
      - .3 Station-to-Station Per Call Charges Calling Card Option

Call Type	Rate Per Call
LEC Card	
Fully Automated	\$0.95
Operator Assisted	\$2.25
Operator Dialed	\$2.25

.4 Station-to-Station Per Call Charges - Collect and Third Party

Call Type	Rate Per Call	
Collect		
Fully Automated	\$2.45	
Operator Assisted	\$2.45	
Operator Dialed	\$3.45	
Third Party		
Fully Automated	\$2.45	
Operator Assisted	\$2.45	
Operator Dialed	\$3.45	

## CANCELLED



Issued: March 7, 2001

Effective:

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

DEC 0 7 2001

FILED DEC 07 2001 Service Commission Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u>

#### SECTION 4 - RATES AND CHARGES

- 4.1 Operator Toll Assistance Services (continued)
  - 4.1.1 Access Method Toll Free Access Number (continued)
    - (B) Group 2 Toll Free Access Numbers
      - .1 Billed To LEC Card

If charges are billed to a LEC Card, the usage rates and per call charges are the same as the usage rates and per call charges described in Section 4.1.2 (A), (B), and (C) of this Tariff.

- .2 Billed to Calling Card
  - .a Per Call Charges

For per call charges, see Section 4.1.2 (C) of this Tariff.

С

Issued: October 28, 2003

Effective: December 1, 2003

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588
Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

4.1	Opera	tor To	ll Ass	•	FION 4 - RATES AND CHARGES Services (continued)	ion
	4.1.1	Acce	ess Me	thod -	Toll Free Access Number (continued)	
		(B)	Grou	цр 2 То	oll Free Access Numbers	~
			.1	Bille	i To LEC Card	
				charg	arges are billed to a LEC Card, the usage rates and per call es are the same as the usage rates and per call charges ibed in Section 4.1.2 (A), (B), and (C) of this Tariff.	
			.2	Bille	I to Calling Card	D
				.a	Per Call Charges	
					- The per call charge for a call billed to the fully automated Calling Card - Option 1 is \$0.75 per call.	D
					- The per call charge for a call billed to the fully automated Calling Card - Option 2 is \$0.75 per call.	D
					MANIMEL I ER	

# CANCELLED

DEC 0 1 2003 LISHOS JO7 Public Service Commission MISSOUS

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

FILED FEB 07 2003

Missourt Public Service Commission

#### Missouri Public

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

RECT MAR 0.7 2001 Service Commission

#### SECTION 4 - RATES AND CHARGES

- 4.1 Operator Toll Assistance Services (continued)
  - 4.1.1 Access Method Toll Free Access Number (continued)
    - (B) Group 2 Toll Free Access Numbers
      - .1 Billed To LEC Card

If charges are billed to a LEC Card, the usage rates and per call charges are the same as the usage rates and per call charges described in Section 4.1.2 (A), (B), and (C) of this Tariff.

- .2 Billed to Proprietary Calling Card
  - .a Per Call Charges
    - The per call charge for a call billed to the fully automated Proprietary Calling Card - Option 1 is \$0.75 per call.
    - The per call charge for a call billed to the fully automated Proprietary Calling Card Option 2 is \$0.75 per call.

# CANCELLED

EB 07 2003 nssion

Issued: March 7, 2001

DEC 0 7 2001

Effective

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED DEC 07 2001 01 - 1.75 Service Commission Southwestern Bell Communications Services, Inc.PSC Mo. - No. 12nd Revised Sheet 208<u>d/b/a SBC Long Distance</u>Replacing 1st Revised Sheet 208

#### SECTION 4 - RATES AND CHARGES

- 4.1 Operator Toll Assistance Services (continued)
  - 4.1.1 Access Method Toll Free Access Number (continued)
    - (B) Group 2 Toll Free Access Numbers (continued)
      - .2 Billed to Calling Card (continued)
        - .a Per Call Charges (continued)
          - For fully automated calls billed to the Calling Card Option 3, a per call charge does not apply.
          - The per call charges for all other calls billed to the
            Calling Card are located in Section 4.1.2 (B) and 4.1.2 T
            (C) of this Tariff.

Issued: October 28, 2003

Effective: December 1, 2003

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

#### SECTION 4 - RATES AND CHARGES Missouri Public Service Commission 4.1 Operator Toll Assistance Services (continued) REGT JAN 0 7 2003 Access Method - Toll Free Access Number (continued) 4.1.1 Group 2 Toll Free Access Numbers (continued) (B) .2 Billed to Calling Card (continued) D Per Call Charges (continued) .a For fully automated calls billed to the Calling Card -D Option 3, a per call charge does not apply. The per call charges for all other calls billed to the Calling Card - Option 1, Option 2 or Option 3 are D

CANCELLED

located in Section 4.1.2 (B) and 4.1.2 (C) of this

DEC 01 2003 Public Service Commission MISSING

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Tariff.

Missouri Public Service Commission



--

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

RECD MAR 07 ZUU1 Original Sheet 208

Service Commission

#### SECTION 4 - RATES AND CHARGES

- 4.1 Operator Toll Assistance Services (continued)
  - 4.1.1 Access Method Toll Free Access Number (continued)
    - (B) Group 2 Toll Free Access Numbers (continued)
      - .2 Billed to Proprietary Calling Card (continued)
        - .a Per Call Charges (continued)
          - For fully automated calls billed to the Proprietary Calling Card - Option 3, a per call charge does not apply.
            - The per call charges for all other calls billed to the Proprietary Calling Card - Option 1, Option 2 or Option 3 are located in Section 4.1.2 (B) and 4.1.2 (C) of this Tariff.

# CANCELLED

FEB 07 2003 FEB 07 2003 Public Service Coontinussion

Issued: March 7, 2001

Effective April 25, 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED DEC 07 2001 01-475 Service Commission

DEC 07 2001

#### SECTION 4 - RATES AND CHARGES

4.1 Operator Toll Assistance Services (continued)

42

- 4.1.1 Access Method Toll Free Access Number (continued)
  - (B) Group 2 Toll Free Access Numbers (continued)
    - .2 Billed to Calling Card (continued)
      - .b Fully Automated Usage Charges
        - i Calling Card Option 1, Option 2, and Option 4, Value Card Plus

The usage rate may be found in Section 4.1.2 (A) of this Tariff.

plan selected by the Customer. Usage rates may be found

.ii Calling Card - Option 2 Categories

in Section 4 of this Tariff.

Calling card usage rates for Customers that subscribe to Calling Card - Option 2 Categories are the same as the T usage rates that apply to 1+ outbound calls originating via Switched Access and billed under the optional calling

Т

Issued: December 1, 2003

Effective: December 31, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

		cations	Services,	Inc. PSC Mo No. 1 2nd Revised Sheet 209 Replacing 1st Revised Sheet 209	
ton Tall				Missouri Pu	blic
or 101	ASSIS	stance s	bervices (		2003
Acce	ss Me	thod - '	Foll Free	Access Number (continued)	
(B)	Grou	up 2 To	oll Free A		ission
	.2	Bille	d to Call	ing Card (continued)	
		.b	Fully	Automated Usage Charges	
			.i	Calling Card - Option 1, Option 2, and Option 4, Value Card Plus	C 
				The usage rate may be found in Section 4.1.2 (A) of this Tariff.	
			.ii	Calling Card - Option 2, Category 11	
				Calling card usage rates for Customers that subscribe to Calling Card - Option 2, category 11, are the same as the usage rates that apply to 1+ outbound calls originating via Switched Access and billed under the optional calling plan selected by the Customer. Usage rates may be found in Section 4 of this Tariff.	Ì
	tor Toll	ig Distance tor Toll Assis Access Me (B) Grou	g Distance SEC tor Toll Assistance S Access Method - 7 (B) Group 2 To .2 Billeo	Ig Distance SECTION 4 tor Toll Assistance Services 6 Access Method - Toll Free (B) Group 2 Toll Free A .2 Billed to Call .b Fully .i	SECTION 4 - RATES AND CHARGES tor Toll Assistance Services (continued) Access Method - Toll Free Access Number (continued) (B) Group 2 Toll Free Access Numbers (continued) .2 Billed to Calling Card (continued) .b Fully Automated Usage Charges .i Calling Card - Option 1, Option 2, and Option 4, Value Card Plus The usage rate may be found in Section 4.1.2 (A) of this Tariff. .ii Calling Card - Option 2, Category 11 Calling card usage rates for Customers that subscribe to Calling Card - Option 2, category 11, are the same as the usage rates that apply to 1+ outbound calls originating via Switched Access and billed under the optional calling plan selected by the Customer. Usage rates may be found in Section 4 of this Tariff.



1ssued: October 28, 2003

Effective: December 1, 2003

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Missouri Public Sorvico Commission

FLED DEC 01 2003

	hwestern SBC Lo			ication	ns Services, In	c. PSC Mo No. 1	1st Revised Sheet 209 Replacing Original Sheet 209
4.1	Opera	tor Tol	ll Assi		CTION 4 - R. e Services (co	ATES AND CHARG	ES Missouri Public Service Commission RECD JAN 0 7 2003
	4.1.1	Acce	ss Me	thod ·	- Toll Free A	ccess Number (contin	nued)
		(B)	Grou	ıp 2 T	oll Free Acc	ess Numbers (continu	ed)
			.2	Bille	d to Calling	Card (continued)	D
				.b	Fully Auto	omated Usage Charges	s
					.i Cal	lling Card - Option 1	D
					Th	e usage rate is \$0.35 j	per minute.
					.ii Ca	lling Card - Option 2	D

The usage rate is \$0.35 per minute.

CANCELLED DEC 012003 Publi

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Missouri Public Service Commission

FILED FEB 07 2003

#### SECTION 4 - RATES AND CHARGES

- 4.1 Operator Toll Assistance Services (continued)
  - 4.1.1 Access Method Toll Free Access Number (continued)
    - (B) Group 2 Toll Free Access Numbers (continued)
      - .2 Billed to Proprietary Calling Card (continued)
        - .b Fully Automated Usage Charges
          - .i Proprietary Calling Card Option 1

The usage rate is \$0.35 per minute.

.ii Proprietary Calling Card - Option 2

The usage rate is \$0.35 per minute.

CANCELLEN

Roisen

Issued: March 7, 2001

Effective April 23, 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public FILED DEC 072001 01-475

DEC 0 7 2001

Service Commission

Т

#### SECTION 4 - RATES AND CHARGES

#### 4.1 Operator Toll Assistance Services (continued)

- 4.1.1 Access Method Toll Free Access Number (continued)
  - (B) . Group 2 Toll Free Access Numbers (continued)
    - .2 Billed to Calling Card (continued)
      - .b Fully Automated Usage Charges (continued)
        - .iii Calling Card Option 3, and Option 3 Categories

Calling card usage rates for Customers that subscribe to any of the Company's High Volume Calling plans are the same as the usage rates that apply to 1+ outbound calls originating via Switched Access and billed under the High Volume Calling optional calling plan selected by the Customer. Usage rates may be found in Section 4 of this Tariff.

Issued: October 28, 2003

Effective: December 1, 2003

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

2nd Revised Sheet 210 Replacing 1st Revised Sheet 210

RECT MAR 81 2003



Missouri Public Service Commission SECTION 4 - RATES AND CHARGES

- 4.1 Operator Toll Assistance Services (continued)
  - 4.1.1 Access Method Toll Free Access Number (continued)
    - (B) Group 2 Toll Free Access Numbers (continued)
      - .2 Billed to Calling Card (continued)
        - Fully Automated Usage Charges (continued) .b
          - Calling Card Option 3 .iii

Calling card usage rates for Customers that subscribe to any of С the Company's High Volume Calling plans are the same as the usage rates that apply to 1+ outbound calls originating via Switched Access and billed under the High Volume Calling optional calling plan selected by the Customer. Usage rates may be found in Section 4 of this Tariff.

# CANCELLED

DEC 0 1 2003

Issued: March 31, 2003

Effective: April 30, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission FILED APR 80 2003

С

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

						4 - RATES AND CHARGES	Missouri Public Service Commission
4.1	Operator 1	Foll A	Assista	s (continued)			
	4.1.1 Acce	ess M	lethod	- Toll	Free	Access Number (continued)	REGD JAN 0 7 2003
	(B)	Gro	up 2 T	Toll Fr	ree Ac	cess Numbers (continued)	
		.2	Bille	ed to (	Calling	g Card (continued)	D
			.b	Fully	y Auto	omated Usage Charges (continued)	
				.iii	Calli	ng Card - Option 3	D
					(1)	Calling Card Rates for Customers The High Volume Calling or High Volum Calling	
						The per minute usage rates are the sa 4.7.1 (A).1 of this Tariff.	ame as Section
					(2)	Calling Card Rates for Customers Th High Volume Dedicated Outbound C Volume Dedicated Toll Free Calling	
	CANC	FIL 30	ED 2003			The per minute usage rates are the sa 4.7.1 (A).1 of this Tariff.	ame as Section
	APR 2nd Public 20	RS	210 Coorn	nssio	n		

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588



#### Mis**souri** Public

RECD MAR 07 2001 Original Sheet 210 Service Commission

#### SECTION 4 - RATES AND CHARGES

- 4.1 Operator Toll Assistance Services (continued)
  - 4.1.1 Access Method Toll Free Access Number (continued)
    - (B) Group 2 Toll Free Access Numbers (continued)
      - .2 Billed to Proprietary Calling Card (continued)
        - .b Fully Automated Usage Charges (continued)
          - .iii Proprietary Calling Card Option 3
            - (1) Calling Card Rates for Customers That Subscribe To High Volume Calling or High Volume Toll Free Calling

The per minute usage rates are the same as Section 4.7.1 (A).1 of this Tariff.

(2) Calling Card Rates for Customers That Subscribe to High Volume Dedicated Outbound Calling or High Volume Dedicated Toll Free Calling

The per minute usage rates are the same as Section 4.7.1 (A).1 of this Tariff.

CANCELLED

FEB 07 2003 GAS 210 Public Service Commission MISSOURI

Issued: March 7, 2001

Effective April 23, 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

DEC 0 7 2001 Missouri Public

FILED DEC 072001 01-475 Service Commission Southwestern Bell Communications Services, Inc.PSC Mo. - No. 12nd Revised Sheet 210.1d/b/a SBC Long DistanceReplacing 1st Revised Sheet 210.1

#### SECTION 4 - RATES AND CHARGES

Issued: March 31, 2003

•

Effective: April 30, 2003

i

D

D

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Southwestern Bell Communications Services, Inc. d/b/a SBC Long Distance

Missouri Public Service Cemmissien

REGD JAN 07 2003

#### SECTION 4 - RATES AND CHARGES

### 4.1 Operator Toll Assistance Services (continued)

#### 4.1.1 Access Method - Toll Free Access Number (continued)

- (B) Group 2 Toll Free Access Numbers (continued)
  - .2 Billed to Calling Card (continued)
    - .b Fully Automated Usage Charges (continued)
      - .iii Calling Card Option 3 (continued)
        - (3) Calling Card Rates for Customers That Subscribe To High Volume Outbound Calling II or High Volume Toll Free Calling II

The per minute usage rates are the same as Section 4.7.2 (A).1 of this Tariff.

(4) Calling Card Rates for Customers That Subscribe to High Volume Dedicated Outbound Calling II or High Volume Dedicated Toll Free Calling II

The per minute usage rates are the same as Section 4.7.2 (A).1 of this Tariff.

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Missouri Public Service Commission FILED FEB 072003



D

D

.2

Original Sheet 210.1

### SECTION 4 - RATES AND CHARGES Missouri Public 4.1 Operator Toll Assistance Services (continued) Ν REC'D NOV 2 6 2002 4.1.1 Access Method - Toll Free Access Number (continued) (B) Group 2 Toll Free Access Numbers (continued) Billed to Proprietary Calling Card (continued) Fully Automated Usage Charges (continued) .b Proprietary Calling Card - Option 3 (continued) .111 Calling Card Rates for Customers That Subscribe To (3) High Volume Outbound Calling II or High Volume Toll Free Calling II The per minute usage rates are the same as Section 4.7.2 (A).1 of this Tariff. Calling Card Rates for Customers That Subscribe to (4) High Volume Dedicated Outbound Calling II or High Volume Dedicated Toll Free Calling II The per minute usage rates are the same as Section 4.7.2 (A).1 of this Tariff. Ν

# CANCELLED

FEB 0 7 2003 ៣៩នា០ក

Issued: November 26, 2002

Effective: December 27, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Serviso Commission

FILEB DEC 27 2067

#### Missouri Public

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance RECD MAR 07 2001

Service Commission

I.

#### SECTION 4 - RATES AND CHARGES

- 4.1 Operator Toll Assistance Services (continued)
  - 4.1.1 Access Method Toll Free Access Number (continued)
    - (B) Group 2 Toll Free Access Numbers (continued)
      - .3 All Other Operator Toll Assistance Billing Options

For all other Operator Toll Assistance Services calls completed via Group 2 Toll Free Access Numbers, the usage rates and per call charges are the same as the usage rates and per call charges described in Section 4.1.2 of this Tariff.

Issued: March 7, 2001

Effective April 23, 2001

DEC 072001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED DEC 072001 01-475 Service Commission

Missouri Public

#### SECTION 4 - RATES AND CHARGES

#### 4.1 Operator Toll Assistance Services (continued)

#### 4.1.2 All Other Access Methods

There are two rate elements. They include a usage charge and a per call charge. The usage charges and per call charges follow:

(A) Usage Rates

Call Type	Rate Per Minute
LEC Card	\$0.35
Calling Card - Option 1	\$0.75
Calling Card - Option 2	\$0.35
Calling Card - Option 3	See Section 4.1.1 (B).2.b.iii
Calling Card - Option 4	\$0.15
All Other Operator Toll Assistance Services Calls	
- Business	\$0.35
- Residential	\$0.35

(B) Person-to-Person Per Call Charge

	Rate Per Call	\$4.90
--	---------------	--------

Issued: January 26, 2005

Effective: February 12, 2005

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED Missouri Public Service Commission

I

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

4th Revised Sheet 212 Replacing 3rd Revised Sheet 212

#### SECTION 4 - RATES AND CHARGES

- 4.1 Operator Toll Assistance Services (continued)
  - 4.1.2 All Other Access Methods

There are two rate elements. They include a usage charge and a per call charge. The usage charges and per call charges follow:

(A) Usage Rates

Call Type	Rate Per Minute	
LEC Card	\$0.35	
Calling Card - Option 1	\$0.50	
Calling Card - Option 2	\$0.35	
Calling Card - Option 3	See Section 4.1.1 (B).2.b.iii	
Calling Card - Option 4	\$0.15	
All Other Operator Toll Assistance Services Calls		
- Business	\$0.35	
- Residential	\$0.35	

(B) Person-to-Person Per Call Charge



Issued: October 28, 2003

Effective: December 1, 2003

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Missouri Public Servico Germiosien

FILED DEC 01 2003

RECD 0CT 2 8 2003

Service Commission

Ι

Ν

Missouri Public

Southwestern Bell Communications Services, Inc. d/b/a SBC Long Distance

PSC Mo. - No. 1 3rd Revised Sheet 212 Replacing 2nd Revised Sheet 212

#### SECTION 4 - RATES AND CHARGES

- 4.1 Operator Toll Assistance Services (continued)
  - 4.1.2 All Other Access Methods

There are two rate elements. They include a usage charge and a per call charge. The usage charges and per call charges follow:

(A) Usage Rates

Call Type	Rate Per Minute	
LEC Card	\$0.35	
Calling Card - Option 1	\$0.35	D
Calling Card - Option 2	\$0.35	D
Calling Card - Option 3	See Section 4.1.1 (B).2.b.iii	D
All Other Operator Toll Assistance		
Services Calls		
- Business	\$0.35	
- Residential	\$0.35	

(B) Person-to-Person Per Call Charge

Rate Per Call	\$4.90

## CANCELLED



Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588





#### Missouri Public Service Commission

REGD JAN 0 7 2003

Southwestern Bell Communications Services, Inc.	PSC Mo No. 1	2nd Revised Sheet 212
d/b/a SBC Long Distance		Replacing 1st Revised Sheet 212

### SECTION 4 - RATES AND CHARGES Missouri Public

4.1 Operator Toll Assistance Services (continued)

REC'D FEB 1 4 2002

4.1.2 All Other Access Methods

Service Commission

There are two rate elements. They include a usage charge and a per call charge. The usage charges and per call charges follow:

Call Type	Rate Per Minute
LEC Card	\$0.35
Proprietary Calling Card - Option 1	\$0.35
Proprietary Calling Card - Option 2	\$0.35
Proprietary Calling Card - Option 3	See Section
	4.1.1 (B).2.b.iii
All Other Operator Toll Assistance	
Services Calls	
- Business	\$0.35
- Residential	\$0.35

(A) Usage Rates

(B) Person-to-Person Per Call Charge

Rate Per Call \$4.90

Missouri Public

Ι

# CANCELLED

FILED MAR 0 5 2002

Service Commission

FEB 0 7 2003

MISSION Publ

Issued: February 14, 2002

Effective: March 5, 2002

#### SECTION 4 - RATES AND CHARGES

4.1 Operator Toll Assistance Services (continued)

Usage Rates

4.1.2 All Other Access Methods

(A)

There are two rate elements. They include a usage charge and a per call charge. The usage charges and per call charges follow:

Call Type	Rate Per Minute	
LEC Card	\$0.35	
Proprietary Calling Card - Option 1	\$0.35	
Proprietary Calling Card - Option 2	\$0.35	
Proprietary Calling Card - Option 3	See Section	
	4.1.1 (B).2.b.iii	
All Other Operator Toll Assistance		
Services Calls		
- Business	\$0.25	
- Residential	\$0.35	

(B) Person-to-Person Per Call Charge

Rate Per Call	\$4.90

# CANCELLED

MAR 0 5 2002 ission PUDIIC

MISSOURI

Issued: December 7, 2001

Effective: January 6, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

FILED JAN 06 2002

Service Commission

Missouri Public

RECT DEC 07 2001

Service Commission

C C Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

RECD MAR 07 2001 Original Sheet 212 Service Commission

#### SECTION 4 - RATES AND CHARGES

- 4.1 Operator Toll Assistance Services (continued)
  - 4.1.2 All Other Access Methods

There are two rate elements. They include a usage charge and a per call charge. The usage charges and per call charges follow:

Call Type	Rate Per Minute
LEC Card	\$0.35
Proprietary Calling Card - Option 1	\$0.35
Proprietary Calling Card - Option 2	\$0.35
Proprietary Calling Card - Option 3	See Section
	4.1.1 (B).2.b.iii
All Other Operator Toll Assistance Services	
Calls	
- Business Origination	\$0.25
- Residential Origination	\$0.35

(A) Usage Rates

(B) Person-to-Person Per Call Charge



Issued: March 7, 2001

Effective: April 25, 2901

Norm Descoteaux, Regulatory Manager DEC 0 7 2001 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

> FILED DEC 07 2001 01 - 475 Service Commission

#### SECTION 4 - RATES AND CHARGES

#### 4.1 Operator Toll Assistance Services (continued)

#### 4.1.2 All Other Access Methods (continued)

(C) Station-to-Station Per Call Charges - Calling Card Option

Call Type	Rate Per Call
Calling Card	
LEC Card	
Fully Automated	\$0.95
Operator Assisted	\$2.25
Operator Dialed	\$2.25
Calling Card - Option 1	
Fully Automated	\$1.25
Operator Assisted	\$1.95
Operator Dialed	\$2.95
Calling Card - Option 2	
Fully Automated	\$0.95
Operator Assisted	\$1.95
Operator Dialed	\$2.95
Calling Card - Option 3	
Operator Assisted	\$1.00
Operator Dialed	\$2.00
Calling Card - Option 4, Value Card Plus	
Operator Assisted	\$1.00
Operator Dialed	\$1.50

Ι

Issued: January 26, 2005

Effective: February 12, 2005

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED Missouri Public Service Commission Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 2nd Revised Sheet 213 d/b/a SBC Long Distance Replacing 1st Revised Sheet 213

#### **Missouri** Public **SECTION 4 - RATES AND CHARGES**

4.1 Operator Toll Assistance Services (continued)

Issued: October 28, 2003

REC'D OCT 2 8 2003

All Other Access Methods (continued) 4.1.2

Service Commission

(C) Station-to-Station Per Call Charges - Calling Card Option

Call Type	Rate Per Call
Calling Card	
LEC Card	
Fully Automated	\$0.95
Operator Assisted	\$2.25
Operator Dialed	\$2.25
Calling Card - Option 1	
Fully Automated	\$0.95
Operator Assisted	\$1.95
Operator Dialed	\$2.95
Calling Card - Option 2	
Fully Automated	\$0.95
Operator Assisted	\$1.95
Operator Dialed	\$2.95
Calling Card - Option 3	
Operator Assisted	\$1.00
Operator Dialed	\$2.00
Calling Card - Option 4, Value Card Plus	
Operator Assisted	\$1.00
Operator Dialed	\$1.50

### CANCELLED

FEB 1 2 2005 Bv **Public Service Commission** MISSOURI

Effective: December 1, 2003

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Servico Commicolon

FILED DEC 01 2003

ces, Inc.

1st Revised Sheet 213 Replacing Original Sheet 213

#### SECTION 4 - RATES AND CHARGES

PSC Mo. - No. 1

- 4.1 Operator Toll Assistance Services (continued)
  - All Other Access Methods (continued) 4.1.2
    - (C) Station-to-Station Per Call Charges - Calling Card Option

Call Type	Rate Per Call
Calling Card	
LEC Card	
Fully Automated	\$0.95
Operator Assisted	\$2.25
Operator Dialed	\$2.25
Calling Card - Option 1	
Fully Automated	\$0.75
Operator Assisted	\$1.25
Operator Dialed	\$2.25
Calling Card - Option 2	
Fully Automated	\$0.75
Operator Assisted	\$1.25
Operator Dialed	\$2.25
Calling Card - Option 3	· · · _ · _ · _ · _ · _ · _
Operator Assisted	\$1.00
Operator Dialed	\$2.00

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FILED FEB 07 2003

È C L L L

Southwestern Bell Communications Ser	rv10
d/b/a SBC Long Distance	

Missouri Public Service Commission

REGD JAN 97 2003

RECTD MAR 07 2001

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

Original Sheet 213 Service Commission

#### **SECTION 4 - RATES AND CHARGES**

- 4.1 Operator Toll Assistance Services (continued)
  - 4.1.2 All Other Access Methods (continued)

(C) Station-to-Station Per Call Charges - Calling Card Option

Call Type	Rate Per Call
Calling Card	
LEC Card	
Fully Automated	\$0.95
Operator Assisted	\$2.25
Operator Dialed	\$2.25
Proprietary Calling Card - Option 1	
Fully Automated	\$0.75
Operator Assisted	\$1.25
Operator Dialed	\$2.25
Proprietary Calling Card - Option 2	
Fully Automated	\$0.75
Operator Assisted	\$1.25
Operator Dialed	\$2.25
Proprietary Calling Card - Option 3	· · · · · · · · · · · · · · · · · · ·
Operator Assisted	\$1.00
Operator Dialed	\$2.00

CANCELLED

07 2003 cine Pub

Issued: March 7, 2001

Effective April 23-2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missoury Fable

FILED DEC 072001 01-475 Service Commission Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u>

#### SECTION 4 - RATES AND CHARGES

- 4.1 Operator Toll Assistance Services (continued)
  - 4.1.2 All Other Access Methods (continued)
    - (C) Station-to-Station Per Call Charges Calling Card Option (continued)

Call Type	Rate Per Call
Calling Card - Option 2 Categories	
Category 11	
Fully Automated	\$0.75
Operator Assisted	\$1.25
Operator Dialed	\$2.25

Ν

N

Issued: October 28, 2003

Effective: December 1, 2003

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

#### Original Sheet 213.2

#### SECTION 4 - RATES AND CHARGES

#### 4.1 Operator Toll Assistance Services (continued)

- 4.1.2 All Other Access Methods (continued)
  - (C) Station-to-Station Per Call Charges Calling Card Option (continued)

Call Type	Rate Per Call
Calling Card - Option 3 Categories	
Category 21	
Fully Automated	\$0.50
Operator Assisted	\$1.00
Operator Dialed	\$2.00

N

Ν

Issued: October 28, 2003

Effective: December 1, 2003

#### SECTION 4 - RATES AND CHARGES

#### 4.1 Operator Toll Assistance Services (continued)

- 4.1.2 All Other Access Methods (continued)
  - (D) Station-to-Station Per Call Charges
    - Collect, Third Number, or Sent Paid

Call Type	Rate Per Call
Collect	
Fully Automated	\$2.25
Operator Assisted	\$2.25
Operator Dialed	\$3.25
Third Party	
Fully Automated	\$2.25
Operator Assisted	\$2.25
Operator Dialed	\$3.25
Sent Paid	
Operator Assisted	\$2.25
Operator Dialed	\$3.25

#### 4.1.3 MRC

(A) Calling Card - Option 4, Value Card Plus

The MRC is \$1.95.

Issued: October 28, 2003

Effective: December 1, 2003

Ν

Ν

### Missouri Public

# RECT MAR 07 2001

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

SerVige Cornalission

#### SECTION 4 - RATES AND CHARGES

- 4.1 Operator Toll Assistance Services (continued)
  - 4.1.2 All Other Access Methods (continued)
    - (D) Station-to-Station Per Call Charges
      - Collect, Third Number, or Sent Paid

Call Type	Rate Per Call
Collect	
Fully Automated	\$2.25
Operator Assisted	\$2.25
Operator Dialed	\$3.25
Third Party	
Fully Automated	\$2.25
Operator Assisted	\$2.25
Operator Dialed	\$3.25
Sent Paid	
Operator Assisted	\$2.25
Operator Dialed	\$3.25

# CANCELLED

DEC 01 2003 Pn

Issued: March 7, 2001

Effective April 23, 2004

Norm Descoteaux, Regulatory Manager DEC 0 7 2001 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

> FILED DEC 07 2001 01-475 Service Commission

#### SECTION 4 - RATES AND CHARGES

4.2 Directory Assistance Servi	ices
--------------------------------	------

•

4.2.1 Reserved for future use

T/D

. .....

4.2.2 Call Completed Via All Other Access Methods

The rate is \$1.25 per call.

4.2.3 Directory Assistance Call Completion

The rate is \$0.35 per completed call.

Issued: December 3, 2003

Effective: January 9, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

d/b/a SBC Long Distance		g Distance Replaci	Replacing 2nd Revised Sheet 215	
		SECTION 4 - RATES AND CHARGES	Missourl Public Service Commission	
4.2	Direct	ory Assistance Services	REC'D MAY 20 2003	
	4.2.1	Call Completed Via Group 1 Toll Free Numbers		
		The rate is \$1.00 per call.		
	4.2.2	Call Completed Via All Other Access Methods		
		The rate is \$1.25 per call.	Ι	
	4.2.3	Directory Assistance Call Completion		

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1

The rate is \$0.35 per completed call.

CANCELLED

3rd Revised Sheet 215

JAN 0 9 2004 By 4th RS 215 Public Service Commission MISSOURI

Issued: May 20, 2003

î

į

Effective: June 1, 2003

Tawnya Rechtin. Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission FILED JUN 01 2003 Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance 2nd Revised Sheet 215 Replacing 1st Revised Sheet 215

#### SECTION 4 - RATES AND CHARGES

- 4.2 Directory Assistance Services
  - 4.2.1 Call Completed Via Group 1 Toll Free Numbers The rate is \$1.00 per call.
  - 4.2.2 Call Completed Via All Other Access Methods

The rate is \$1.00 per call.

4.2.3 Directory Assistance Call Completion

The rate is \$0.35 per completed call.

Missouri Public

REC'D AUG 01 2002

# Service Commission

M

Μ

CANCELLED JUN 01 2003 CRS 215 CONTINUESION

M - Material moved to Original Sheet 215.1

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

FILED SEP 01 2002

Service Commission

	western Bell Communications Services, Inc. PSC Mo No. 1 SBC Long Distance	1st Revised Sheet 215 Replacing Original Sheet 215	
	SECTION 4 - RATES AND CHARGES	Missouri Public	
4.2	Directory Assistance Services	REC'D MAR 05 2002	
	4.2.1 Call Completed Via Group 1 Toll Free Numbers		

The rate is \$1.00 per call.

4.2.2 Call Completed Via All Other Access Methods

The rate is \$1.00 per call.

4.2.3 Directory Assistance Call Completion The rate is \$0.35 per completed call.

CANCELLED

SEP 0 1 2002 nssior Public

Service Commission

Т

Ν

4.3 Inmate Service

The usage rate is \$0.45 per minute. The per call service charge is \$3.00.

Missouri Public

FILED APR 04 2002

Service Commission

Issued: March 5, 2002

Effective: April 4, 2002

### Missourl Public

### RECTD MAR 07 2001

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance Serviem Damanission

#### SECTION 4 - RATES AND CHARGES

- 4.2 Directory Assistance Services
  - 4.2.1 Call Completed Via Group 1 Toll Free Numbers

The rate is \$1.00 per call.

4.2.2 Call Completed Via All Other Access Methods

The rate is \$1.00 per call.

4.2.3 Directory Assistance Call Completion

The rate is \$0.35 per completed call.

4.3 Reserved for future use

# CANCELLED



Issued: March 7, 2001

Effective: April 23, 2001 Norm Descoteaux, Regulatory Manager DEC 0 7 2001 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public FILED DEC 0 7 2001

01-475 Service Commission
#### 4.3 Inmate Service

REC'D AUG 01 2002

Missouri Publia

The usage rate is \$0.45 per minute. The per call service charge is \$3.0 Service Commission

M - Material moved from 1st Revised Sheet 215

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

#### 4.4 Outbound Services-Switched Access

#### 4.4.1 MTS

	Peak		Off-Peak		
	Initial Period	Add'l Period	Initial Period	Add'l Period	
Business	\$0.32	\$0.32	\$0.22	\$0.22	
Residential	\$0.32	\$0.32	\$0.22	\$0.22	

#### 4.4.2 Long Distance III, aka JustCallK Standard

The usage rate is \$0.10 per minute.

Issued: June 8, 2004

Effective: July 8, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Southwestern Bell Communications Services, Inc.	PSC Mo No. 1	3rd Revised Sheet 216
d/b/a SBC Long Distance		Replacing 2nd Revised Sheet 216
		Missouri Public

#### 4.4 Outbound Services-Switched Access

4.4.1 MTS

	Pe	Peak		f-Peak
	Initial Period	Add'l Period	Initial Period	Add'l Period
Business	\$0.32	\$0.32	\$0.22	\$0.22
Residential	\$0.32	\$0.32	\$0.22	\$0.22

4.4.2 Long Distance III

The usage rate is \$0.10 per minute.

R

Service Commission

REC'D MAY 28 2004

CANCELLED JUL 0 8 2004 ssion

Issued: May 28, 2004

Effective: June 7, 2004

Lisa Porterfield, Associate Director Regulatory M 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FILED JUN 07 2004



Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 2nd Revised Sheet 216 <u>d/b/a SBC Long Distance</u>
<u>Replacing 1st Revised Sheet 216</u>
<u>Wissourr</u> Public

#### SECTION 4 - RATES AND CHARGES

4.4 Outbound Services-Switched Access

4.4.1 MTS

REC'D APR 2 6 2004

Service Commission

		Peak		Off-Peak				
	Initial Period	[	Add'l Period		Initial Per	iod	Add'l Period	
Business	\$0.32	I	\$0.32	I	\$0.22	I	\$0.22	I
Residential	\$0.32	I	\$0.32	I	\$0.22	I	\$0.22	I

4.4.2 Long Distance III

The usage rate is \$0.50 per minute.

### CANCELLED

#### JUN 0 7 2004 By 3rd RS 2/6 Public Service Commission MISSOURI

Issued: April 26, 2004

Effective: May 17, 2004

Missouri Public Service Commission

FILED MAY 17 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

Replacing Original Sheet 216

#### SECTION 4 - RATES AND CHARGES

#### 4.4 Outbound Services-Switched Access

#### 4.4.1 MTS

· · · · · · · · · · · · · · · · · · ·		Peak		Off-Peak		
	Initial Period	Add'l Period	Initial Period	Add'l Period		
Business	\$0.27	\$0.27	\$0.17	\$0.17		
Residential	\$0.27	\$0.27	\$0.17	\$0.17		

4.4.2 Long Distance III

The usage rate is \$0.50 per minute.

## CANCELLED

MAY 1 7 2004 Dr 2004 RS214 Lic Service Commission MISSOURI

Issued: February 27, 2004

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FILED APR 01 2004

## REC'D FEB 27 2004

Miggouri Public

1st Revised Sheet 216

Service Commission

Т

Ν



#### Missouri Public

## RECT MAR 07 2001

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance Original Sheet 216 Service Commission

#### SECTION 4 - RATES AND CHARGES

4.4 Outbound Services-Switched Access

4.4.1 MTS

	Peak		Off-Peak		
	Initial Period	Add'l Period	Initial Period	Add'1 Period	
Business	\$0.27	\$0.27	\$0.17	\$0.17	
Residential	\$0.27	\$0.27	\$0.17	\$0.17	

4.4.2 Reserved for future use

## CANCELLED

#### APR 0 1 2004 By <sup>/S+</sup> RS 216 Public Service Commission MISSOURI

Issued: March 7, 2001	Effec	tive April 23, 2001
	Norm Descoteaux, Regulatory Manager	DEC 072001
585	50 W. Las Positas Blvd., Pleasanton, California 945	Missouri Public
		and the second
		FILED DEC 07 2001
		Bervice Commission

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

RECD MAR 07 2001 Original Sheet 217 Service Commission

i

#### SECTION 4 - RATES AND CHARGES

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services
    - (A) Reserved for future use
    - (B) Reserved for future use
    - (C) Reserved for future use
    - (D) Reserved for future use

Issued: March 7, 2001

Effective: April 23, 2001

DEC 072001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED DEC 072001 01-475 Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

4.4 Outbound Services-Switched Access (continued)

\_\_\_\_

- 4.4.3 Consumer Outbound Services
  - (E) Simple Solutions® II

Т

The usage rate is \$0.10 per minute for InterLATA calling and \$0.10 for IntraLATA calling.

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

			SECTION 4 - RATES AND CHARC	
4.4	Outbo	und Se	rvices-Switched Access (continued)	REC'D AUG 01 2002
	4.4.3	Cons	sumer Outbound Services	Service Commission
		(E)	Simple Solutions II	
			The usage rate is \$0.10 per minute for Int IntraLATA calling.	terLATA calling and \$0.10 for
				М
				 M
				M*
				M*
			CANCE	_LED
			FEB 0 By Sill R	Commission M**
				M***

M - Material moved to Original Sheet 218.1.1 M\* - Material moved to Original Sheet 218.1.2 M\*\* - Material moved to Original Sheet 218.1.3 M\*\*\* - Material moved to Original Sheet 218.1.4

Issued: August 1, 2002

I

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED SEP 01 2002

- 4.4.3 Consumer Outbound Services
  - (E) Simple Solutions II

The usage rate is \$0.10 per minute for InterLATA calling and \$0.10 for I IntraLATA calling.

0 1 2002

nssion

(F) Long Distance II

The usage rate is \$0.14 per minute for InterLATA calling and \$0.14 for IntraLATA calling.

(G) Domestic Saver

The usage rate is \$0.07 per minute. For Customers subscribing to Domestic Saver on a stand-alone basis for the provision of intrastate IntraLATA calling, the MRC is \$4.95.

(H) Long Distance Block of Time 500 Minutes

The monthly recurring charge is \$24.95 per BTN for a 500 minute block of time for intrastate and interstate calling. The rate is \$0.07 per minute for all outbound intrastate calls completed after the 500 minute block of time has been used.

(I) Simple Solutions Block of Time 100

The monthly recurring charge is \$6.00 per BTN for a 100 minute block of time for intrastate and interstate calling. The rate is \$0.08 per minute for all outbound intrastate calls completed after the 100 minute block of time has been used.

Issued: July 17, 2002

Effective: August 1, 2002

Missouri Publio

FILED AUG 01 2002

REC'D JUL 1 7 2002

Service Commission

I

Service Commission

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588





Issued: February 15, 2002

Effective: March 17, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

FILED MAR 1 7 2002

1st Revised Sheet 218 Replacing Original Sheet 218

#### SECTION 4 - RATES AND CHARGES

#### 4.4 Outbound Services-Switched Access (continued)

- 4.4.3 Consumer Outbound Services
  - (E) Simple Solutions  $\Pi$

The usage rate is \$0.08 per minute for InterLATA calling and \$0.08 for IntraLATA calling.

(F) Long Distance II

The usage rate is \$0.12 per minute for InterLATA calling and \$0.12 for IntraLATA calling.

(G) Domestic Saver

Issued: December 7, 2001

The monthly recurring charge is \$4.95. The usage rate is \$0.07 per minute. N

(H) Long Distance Block of Time 500 Minutes

The monthly recurring charge is \$24.95 per BTN for a 500 minute block of time for intrastate and interstate calling. The rate is \$0.07 per minute for all outbound intrastate calls completed after the 500 minute block of time has been used.

(I) Simple Solutions Block of Time 100

The monthly recurring charge is \$6.00 per BTN for a 100 minute block of time for intrastate and interstate calling. The rate is \$0.08 per minute for all outbound intrastate calls completed after the 100 minute block of time has been used.

ommission

Effective: January 6, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED JAN 06 2002

Service Commission

Missouri Public

REC'D DEC 07 2001

#### Missouri Public

RECTD MAR 07 2001

Service Commission

Original Sheet 218

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

SECTION 4 - RATES AND CHARGES CANCELLED

4.4.3 Consumer Outbound Services

4

(E) Simple Solutions II

JAN 06 2002 1925218 ervice Commission

The usage rate is \$0.08 per minute for InterLATA calling and \$0.08 for IntraLATA calling.

(F) Long Distance II

The usage rate is \$0.12 per minute for InterLATA calling and \$0.12 for IntraLATA calling.

(G) Domestic Saver

The usage rate is \$0.07 per minute.

(H) Long Distance Block of Time 500 Minutes

The monthly recurring charge is \$24.95 per BTN for a 500 minute block of time for intrastate and interstate calling. The rate is \$0.07 per minute for all outbound intrastate calls completed after the 500 minute block of time has been used.

(I) Simple Solutions Block of Time 100

The monthly recurring charge is \$6.00 per BTN for a 100 minute block of time for intrastate and interstate calling. The rate is \$0.08 per minute for all outbound intrastate calls completed after the 100 minute block of time has been used.

Issued: March 7, 2001

#### Effective April 23 2001

DEC 072001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED DEC 07 2001 01 - 475 Service Commission

SI	ECTION 4 - RAT	TES AND CHARGES	Missouri Pul	olic
			RECTO AUG 01	2002
			Service Commi	ssio
				M
				М
				ĺ
				 M
				М
				Μ
				ł
				M

M\* - Material moved to Original Sheet 218.1.6

 $M^{\boldsymbol{**}}$  - Material moved to Original Sheet 218.1.7

M\*\*\* - Material moved to Original Sheet 218.1.8

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

3rd Revised Sheet 218.1 Replacing 2nd Revised Sheet 218.1

#### **SECTION 4 - RATES AND CHARGES** Missouri Public Service Commission 4.4 Outbound Services-Switched Access (continued) REC'D APR 08 2002 4.4.3 Consumer Outbound Services (continued) **(J)** Reserved for future use Block of Time: 30 Minutes<sup>1</sup> (K) The monthly recurring charge is \$2.95 per BTN for a 30 minute block of time for intrastate and interstate calling. The rate is \$0.10 per minute for all outbound intrastate calls completed after the 30 minute block of time has been used. Reserved for future use (L)

(M) 150 Block of Time Gold

The monthly recurring charge is \$10.00 per BTN for a 150 minute block of time for intrastate and interstate calling as defined in Section 3.4.3 (M) of this Tariff. The rate is \$0.11 per minute for all outbound intrastate calls completed after the 150 minute block of time has been used.

<sup>1</sup> This Service is no longer available to new Customers effective May 8, 2002.

Issued: April 8, 2002

Effective: May 8, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588 Service Commission

FILED MAY 08 2002



Ν

2nd Revised Sheet 218.1 Replacing 1st Revised Sheet 218.1

#### SECTION 4 - RATES AND CHARGES

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services (continued)
    - (J) Reserved for future use
    - (K) Block of Time: 30 Minutes

The monthly recurring charge is \$2.95 per BTN for a 30 minute block of time for intrastate and interstate calling. The rate is \$0.10 per minute for all outbound intrastate calls completed after the 30 minute block of time has been used.

- (L) Reserved for future use
- (M) 150 Block of Time Gold

The monthly recurring charge is \$10.00 per BTN for a 150 minute block of time for intrastate and interstate calling as defined in Section 3.4.3 (M) of this Tariff. The rate is \$0.11 per minute for all outbound intrastate calls I completed after the 150 minute block of time has been used.

## CANCELLED

MAY 0 8 2002 STOP DS 218.1 Jublic Service Commission MISSOURI Missouri Public

FILED MAY 01 2002

Service Commission

Issued: April 19, 2002

Effective: May 1, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588



Missouri Public RECD APR 1 9 2002 Service Commission Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 1st Revised Sheet 218.1 d/b/a SBC Long Distance Replacing Original Sheet 218.1 **SECTION 4 - RATES AND CHARGES** Missouri Public 4.4 **Outbound Services-Switched Access (continued)** REC'D FEB 21 2002 Consumer Outbound Services (continued) 4.4.3 Service Commission Reserved for future use (J) Block of Time: 30 Minutes (K) The monthly recurring charge is \$2.95 per BTN for a 30 minute block of time for intrastate and interstate calling. The rate is \$0.10 per minute for all outbound intrastate calls completed after the 30 minute block of time has been used. (L) Reserved for future use Т (M) 150 Block of Time Gold Ν The monthly recurring charge is \$10.00 per BTN for a 150 minute block of time for intrastate and interstate calling as defined in Section 3.4.3 (M) of this Tariff. The rate is \$0.09 per minute for all outbound intrastate calls completed after the 150 minute block of time has been used. Ν

## CANCELLED

MAY 012002 MAY 012002 RS2(8.) Fublic Service Commission MISSOURI

Missouri Public

FILED MAR 23 2002

Service Commission

Issued: February 21, 2002

Effective: March 23, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

- Outbound Services-Switched Access (continued) 4.4
  - 4.4.3 **Consumer** Outbound Services
    - Reserved for future use (J)
    - Block of Time: 30 Minutes (K)

for intrastate and interstate calling. The rate is \$0.10 per minute for all outbound intrastate calls completed after the 30 minute block of time has been used.

Reserved for future use (L)

Issued: December 7, 2001

Effective: January 6, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED JAN 06 2002

Service Commission

Service Commission

REC'D DEC 07 2001 The monthly recurring charge is \$2.95 per BTN for a 30 minute block of time



Original Sheet 218.1

Missouri Public

Ν

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services
    - (F) Long Distance II<sup>1</sup>

С

The usage rate is \$0.16 per minute for InterLATA calling and \$0.16 for IntraLATA calling.

<sup>1</sup> This Service is no longer available to new Customers effective May 9, 2004.

Ν

Issued: April 9, 2004

Effective: May 9, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

	stern Bell Co C Long Dist	ommunications Services, Inc.		1st Revised Sheet 218.1.1 Dlacing Original Sheet 218.1.1
		SECTION 4 - RAT	TES AND CHARGES	Missouri Public
4.4 0	Outbound S	ervices-Switched Access (co	ontinued)	<b>RECT</b> MAR 1 0 2004
4	4.4.3 Con	sumer Outbound Services		Service Commission
	(F)	Long Distance II		

The usage rate is \$0.16 per minute for InterLATA calling and \$0.16 for I IntraLATA calling.

# CANCELLED

MAY 0 9 2004 By 2nd 12 219 Public Service Comments

Issued: March 10, 2004

I

Effective: April 8, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588



FILED APR 08 2004

	western SBC Lor		ommunications Services, Inc. PSC Mo No. 1	Original Sheet 218.1.1
<u>u 0/u</u>	<u>020 Bol</u>	<u>,                                    </u>	SECTION 4 - RATES AND CHARGES	Missouri Public
4.4	Outbo	und Se	ervices-Switched Access (continued)	RECT AUG 01 2002 $_{\rm M}$
	4.4.3	Cons	sumer Outbound Services	Service Commission
		(F)	Long Distance II	
			The usage rate is \$0.14 per minute for InterL IntraLATA calling.	ATA calling and \$0.14 for M

## CANCELLED

APR 0 8 2004 <sup>JST</sup> RS 2/8././ Public Service Commission MISSOURI

M - Material moved from 3rd Revised Sheet 218

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

FILED SEP 01 2002

	western SBC Lor		mmunications Services, Inc. PSC Mo No. 1 ance	Original Sheet 218.1.2	
			SECTION 4 - RATES AND CHARGES	Missouri Public	
4.4	Outbo	und Se	ervices-Switched Access (continued)	RECT AUG 01 2002M	
	4.4.3	Cons	sumer Outbound Services	Service Commission	
		(G)	Domestic Saver		
			The usage rate is \$0.07 per minute. For Custome Saver on a stand-alone basis for the provision of in the MRC is \$4.95.		

M - Material moved from 3rd Revised Sheet 218

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

ł

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u> Original Sheet 218.1.3

#### SECTION 4 - RATES AND CHARGES

4.4 Outbound Services-Switched Access (continued)

4.4.3 Consumer Outbound Services

(H) Long Distance Block of Time 500 Minutes

The monthly recurring charge is \$24.95 per BTN for a 500 minute block of time for intrastate and interstate calling. The rate is \$0.07 per minute for all outbound intrastate calls completed after the 500 minute block of time has been used.

M - Material moved from 3rd Revised Sheet 218

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Service Commission

REC'D AUG 01 2902

Missouri Public

Service Commission

Μ

4.4 Outbound Services-Switched Access (continued)

4.4.3 Consumer Outbound Services

(I) Simple Solutions Block of Time  $100^1$ 

Т

The monthly recurring charge is \$6.00 per BTN for a 100 minute block of time for intrastate and interstate calling. The rate is \$0.08 per minute for all outbound intrastate calls completed after the 100 minute block of time has been used.

<sup>1</sup> This Service is no longer available to new Customers effective October 30, 2002.

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 

 Southwestern Bell Communications Services, Inc.
 PSC Mo. - No. 1
 1st Revised Sheet 218.1.4

 d/b/a SBC Long Distance
 Replacing Original Sheet 218.1.4

#### SECTION 4 - RATES AND CHARGES

4.4 Outbound Services-Switched Access (continued)

4.4.3 Consumer Outbound Services

(I) Simple Solutions Block of Time  $100^1$ 

The monthly recurring charge is \$6.00 per BTN for a 100 minute block of time for intrastate and interstate calling. The rate is \$0.08 per minute for all outbound intrastate calls completed after the 100 minute block of time has been used.

CANCELLED

<sup>1</sup> This Service is no longer available to new Customers effective October 30, 2002.

Ν

Issued: September 27, 2002

Effective: October 30, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED OCT 3 0 2002

Service Commission

Missouri Public

RECD SEP 2 7 2002

Service Commission

С

	western SBC Lor		ommunications Services, Inc. PSC Mo No. 1	Original Sheet 218.1.4
<u>u o u</u>	<u>500 E0</u>	<u>15 0 15 (</u>	SECTION 4 - RATES AND CHARGES	Missouri Public
4.4	Outbo	und S	ervices-Switched Access (continued)	RECTO AUG 01 2002
	4.4.3	Con	sumer Outbound Services	Service Commission
		(I)	Simple Solutions Block of Time 100	
			The monthly recurring charge is \$6.00 per BTN for for intrastate and interstate calling. The rate is outbound intrastate calls completed after the 100 been used.	\$0.08 per minute for all

# CANCELLED

OCT 3 0 2002 SHRS 218.1.4 Public S String Commission

M - Material moved from 3rd Revised Sheet 218

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED SEP 01 2002

		Bell Communications Services, Inc. PSC Mo No. 1 ng Distance	Original Sheet 218.1.5 Missouri Public
		SECTION 4 - RATES AND CHARGES	RECTD AUG 01 2002
4.4	Outbo	und Services-Switched Access (continued)	M Service Commission
	4.4.3	Consumer Outbound Services (continued)	
		(J) Reserved for future use	 M

M - Material moved from 3rd Revised Sheet 218.1

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

-----

Missouri Public

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

1

	western SBC Lor		ommunications Services, Inc. PSC Mo No. 1	Original Sheet 218.1.6
<u>u/0/µ</u>		<u>15 Dist</u>		Missouri Public
			SECTION 4 - RATES AND CHARGES	DC00 1110 0 4 0000
4.4	Outbo	und Se	ervices-Switched Access (continued)	REC'D AUG 01 2002
	4.4.3	Cons	sumer Outbound Services (continued)	Service Commission
				l
		(K)	Block of Time: 30 Minutes <sup>1</sup>	
			The monthly recurring charge is \$2.95 per BTN for for intrastate and interstate calling. The rate is outbound intrastate calls completed after the 30 min used.	\$0.10 per minute for all

<sup>1</sup> This Service is no longer available to new Customers effective May 8, 2002.
 M - Material moved from 3rd Revised Sheet 218.1

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Southwestern Bel <u>d/b/a SBC Long I</u>	1 Communications Services, Inc. PSC Mo No. 1 Distance	Original Sheet 218.1.7	
	SECTION 4 - RATES AND CHARGES	REC'D AUG 01 2002	
4.4 Outbound	d Services-Switched Access (continued)	M	
		Service Commission	
4.4.3 0	Consumer Outbound Services (continued)	I	
(	L) Reserved for future use	М	

M - Material moved from 3rd Revised Sheet 218.1

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 FILED SEP 01 2002

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

#### SECTION 4 - RATES AND CHARGES

4.4 Outbound Services-Switched Access (continued)

÷

4.4.3 Consumer Outbound Services (continued)

(M) 150 Block of Time Gold<sup>1</sup>

The monthly recurring charge is \$10.00 per BTN for a 150 minute block of time for intrastate and interstate calling as defined in Section 3.4.3 (M) of this Tariff. The rate is \$0.11 per minute for all outbound intrastate calls completed after the 150 minute block of time has been used.

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations N effective July 9, 2003.

I

Issued: June 9, 2003

Effective: July 9, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

	western E SBC Long		·	PSC Mo No. 1	Original Sheet 218.1.8	
<u>u/0/u</u>	<u></u>	<u> 171010</u>	SECTION 4 - RATE	ES AND CHARGES	Missouri Public	
4.4	Outbound Services-Switched Access (continued) RECD AUG 01					
	4.4.3	Cons	umer Outbound Services (co	ntinued)	Service Commission	
		(M)	150 Block of Time Gold			
			time for intrastate and inters	state calling as defin	TN for a 150 minute block of   ed in Section 3.4.3 (M) of this   bund intrastate calls completed	

after the 150 minute block of time has been used.

CANCELLED

JUL 0 9 2003 By SRS 218.1.8 Public Service Commission MISSDUFT

M - Material moved from 3rd Revised Sheet 218.1

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

Μ

FILED SEP 01 2002

4.4 Outbound Services-Switched Access (continued)

> 4.4.3 Consumer Outbound Services

\$

150 Block of Time<sup>1</sup> (N)

Ç

The monthly recurring charge is \$12.50 per BTN for a 150 minute block of time for intrastate and interstate calling as defined in Section 3.4.3 (N) of this Tariff. The rate is \$0.11 per minute for all outbound intrastate calls completed after the 150 minute block of time has been used.

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective July 9, 2003.

Ν Ν

Issued: June 9, 2003

Effective: July 9, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

		g Distance Replac	C Mo No. 1 2nd Revised Sheet 218.2 Replacing 1st Revised Sheet 218.2	
		SECTION 4 - RATES AND CHARGES	Missouri Public	
4.4	Outbound Services-Switched Access (continued)		RECT) AUG 01 2002	
	4.4.3	Consumer Outbound Services	Service Commission	
		(N) 150 Block of Time		

The monthly recurring charge is \$12.50 per BTN for a 150 minute block of time for intrastate and interstate calling as defined in Section 3.4.3 (N) of this Tariff. The rate is \$0.11 per minute for all outbound intrastate calls completed after the 150 minute block of time has been used.



M - Material moved to Original Sheet 218.2.1 M\* - Material moved to Original Sheet 218.2.2

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

**Missouri** Public

Μ Μ

FILED SEP 01 2002

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u> 1st Revised Sheet 218.2 • Replacing Original Sheet 218.2

SECTION 4 - RATES AND CHARGES

REC'D APR 1 9 2002

Service Commission

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services
    - (N) 150 Block of Time

The monthly recurring charge is \$12.50 per BTN for a 150 minute block of time for intrastate and interstate calling as defined in Section 3.4.3 (N) of this Tariff. The rate is \$0.11 per minute for all outbound intrastate calls completed I after the 150 minute block of time has been used.

(O) 500 Block of Time Gold

The usage rate is \$.10 per minute.

(P) Domestic Saver Gold

The usage rate is \$0.07 per minute. For Customers subscribing to Domestic Saver Gold on a stand-alone basis for the provision of intrastate IntraLATA calling, the MRC is \$3.00.

viissouri Public

FILED MAY 01 2002

Service Commission

Issued: April 19, 2002

Effective: May 1, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588



CANCELLED



	western SBC Lor		ommunications Services, Inc. PSC Mo No. 1 ance	Original Sheet 218.2
<u></u>			SECTION 4 - RATES AND CHARGE	Missouri Public
4.4	Outbo	ound Se	ervices-Switched Access (continued)	REC'D FEB 21 2002 N
	4.4.3	Cons	sumer Outbound Services	Service Commission
		(N)	150 Block of Time	
			The monthly recurring charge is \$12.50 per B time for intrastate and interstate calling as define Tariff. The rate is \$0.09 per minute for all outbours after the 150 minute block of time has been us	ned in Section 3.4.3 (N) of this ound intrastate calls completed
		(0)	500 Block of Time Gold	
			The usage rate is \$.10 per minute.	
		(P)	Domestic Saver Gold	
			The usage rate is \$0.07 per minute. For Custo Saver Gold on a stand-alone basis for the pro- calling, the MRC is \$3.00.	2 ,

## CANCELLED

MAY 01 2002 /S4 RS 2(8.2 rublic Service Commission MISSOURI

Missouri Public

FILED MAR 23 2002

Service Commission

Issued: February 21, 2002

2

Effective: March 23, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

		Bell Communications Services, Inc. PSC Mo No. 1	Original Sheet 218.2.1	
		SECTION 4 - RATES AND CHARGES	Missouri Public	
4.4	Outbo	und Services-Switched Access (continued)	RECT) AUG 01 20024	
	4.4.3	Consumer Outbound Services	Service Commission	
		(O) 500 Block of Time Gold		
		The usage rate is \$.10 per minute.	l M	

M - Material moved from 1st Revised Sheet 218.2

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

~

Missouri Public

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149
	western i SBC Lor		ommunications Services, Inc. PSC Mo No. 1 ance	Original Sheet 218.2.2	
		-	SECTION 4 - RATES AND CHARGES	Missouri Public	
4.4	Outbo	und Se	ervices-Switched Access (continued)	REC'D AUG 01 2002 M	
	4.4.3	Con	sumer Outbound Services	Service Commission	ì
		(P)	Domestic Saver Gold		
			The usage rate is \$0.07 per minute. For Custome Saver Gold on a stand-alone basis for the provisi calling, the MRC is \$3.00.	e i	Ĺ

M - Material moved from 1st Revised Sheet 218.2

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Service Commission

### SECTION 4 - RATES AND CHARGES

4.4 Outbound Services-Switched Access (continued)

÷

- 4.4.3 Consumer Outbound Services (continued)
  - (Q) 200 Block of Time Gold<sup>1</sup>

The usage rate is \$.11 per minute.

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations N effective July 9, 2003.

Issued: June 9, 2003

Effective: July 9, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

		Bell Communications Services, Inc. PSC Mo No. 1 ng Distance	1st Revised Sheet 218.3 <u>Replacing Original Sheet 218.3</u>
		SECTION 4 - RATES AND CHARG	ES Missouri Public
4.4	Outbo	und Services-Switched Access (continued)	REC'D AUG 01 2002
	4.4.3	Consumer Outbound Services (continued)	Service Commission
		(Q) 200 Block of Time Gold	
		The usage rate is \$.11 per minute.	

M | | |

Μ

CANCELLED JUL 0 9 2003 JUL 0 9 2003 Public Service Conternacion

M - Material moved to Original Sheet 218.3.1

Issued: August 1, 2002

ł

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED SEP 01 2002

Service Commissio



after the 200 minute block of time has been used.



Missouri Public

Ν

FILED JUN 01 2002

Service Commission

Issued: May 1, 2002

Effective: June 1, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

### SECTION 4 - RATES AND CHARGES

4.4 Outbound Services-Switched Access (continued)

4.4.3 Consumer Outbound Services (continued)

(R) 200 Block of Time<sup>1</sup>

The monthly recurring charge is \$14.00 per BTN for a 200 minute block of time for intrastate and interstate calling as defined in Section 3.4.3 (R) of this Tariff. The rate is \$0.08 per minute for all outbound intrastate calls completed after the 200 minute block of time has been used.

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective September 15, 2003.

N N

С

Issued: August 14, 2003

Effective: September 15, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

	western SBC Lor		mmunications Services, Inc. PSC Mo No. 1	Original Sheet 218.3.1	
<u>u/ 0/ u</u>	<u></u>	<u>15 10 10 10 10 10 10 10 10 10 10 10 10 10 </u>	SECTION 4 - RATES AND CHARGES	Missouri Publi	ic
4.4	Outbo	und Se	ervices-Switched Access (continued)	RECTI AUG 01 20	024
	4.4.3	Cons	sumer Outbound Services (continued)	Servi <b>ce Commis</b> s	sipn
		(R)	200 Block of Time		
			The monthly recurring charge is \$14.00 per BTN time for intrastate and interstate calling as defined	in Section 3.4.3 (R) of this	   
			Tariff. The rate is \$0.08 per minute for all outbound	d intrastate calls completed	
			after the 200 minute block of time has been used.		M

. . . . . .

M - Material moved from Original Sheet 218.3

# CANCELLED

SEP 1 5 2003 Fot RS 218.3.1 Public Service Commission MISSDUPI

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED SEP 01 2002

Service Commissic

### SECTION 4 - RATES AND CHARGES

4.4.3 Consumer Outbound Services (continued)

(S) 100 Block of Time<sup>1</sup>

The monthly recurring charge is \$6.00 per BTN for a 100 minute block of time for intrastate and interstate calling. The rate is \$0.07 per minute for all outbound intrastate calls completed after the 100 minute block of time has been used.

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations N effective July 9, 2003.

Issued: June 9, 2003

Effective: July 9, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

<sup>4.4</sup> Outbound Services-Switched Access (continued)

2nd Revised Sheet 218.4 Replacing 1st Revised Sheet 218.4

### SECTION 4 - RATES AND CHARGES

Outbound Services-Switched Access (continued) 4.4

> Consumer Outbound Services (continued) 4.4.3

> > **(S)** 100 Block of Time

> > > The monthly recurring charge is \$6.00 per BTN for a 100 minute block of time for intrastate and interstate calling. The rate is \$0.07 per minute for all R outbound intrastate calls completed after the 100 minute block of time has been used.

> > > > CANCELLED

Issued: March 5, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Effective: March 15, 2003

Missourl Public Service Commission

FILED MAR 1 5 2003

Service Commissi REC'D MAR 0528

Missouri Puble

### Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance



	western SBC Lor		ommunications Services, Inc. PSC Mo No. 1 ance I	lst Revised Sheet 218.4 Replacing Original Sheet 218.4
			SECTION 4 - RATES AND CHARGES	Missouri Public
4.4	Outbo	und Se	ervices-Switched Access (continued)	REC'D SEP 272002
	4.4.3	Cons	sumer Outbound Services (continued)	Service Commission
		(S)	100 Block of Time	Т
			The monthly recurring charge is \$6.00 per BTN f	or a 100 minute block of time N

for intrastate and interstate calling. The rate is \$0.08 per minute for all | outbound intrastate calls completed after the 100 minute block of time has | been used.

CANCELLED

MAR 1 5 2003 2nd RS 218,4 Public Service Commission MISSOURI

Issued: September 27, 2002

Effective: October 30, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

.

FILED OCT 3 0 2002

Service Commission

	western Bell Communications Services, Inc. PSC Mo No. 1	Original Sheet 218.4
<u>d/b/a</u>	SBC Long Distance	Missouri Public
4.4	SECTION 4 - RATES AND CHARGES Outbound Services-Switched Access (continued)	RECT AUG 01 2002
4.4	Outobulid Services-Switched Access (continued)	Service Commission
	4.4.3 Consumer Outbound Services (continued)	
	(S) Reserved for future use.	Ň

CANCELLED

OCT 3 0 2002 Public Service Comanssien MSSOURI

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED SEP 01 2002

Service Commission

### SECTION 4 - RATES AND CHARGES

4:4 Outbound Services-Switched Access (continued)

.

4.4.3 Consumer Outbound Services (continued)

(T) 60 Block of Time<sup>1</sup>

The monthly recurring charge is \$4.00 per BTN for a 60 minute block of time for intrastate and interstate calling. The rate is \$0.09 per minute for all outbound intrastate calls completed after the 60 minute block of time has been used.

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations N effective September 15, 2003.

Issued: August 14, 2003

Effective: September 15, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

	western SBC Lor		mmunications Services, Inc. PSC Mo No. 1	Original Sheet 218.5
<u>drora</u>		<u>15 D 13</u> 10		
			SECTION 4 - RATES AND CHARGES	
4.4	Outbo	und Se	rvices-Switched Access (continued)	RECD AUG 01. 2002
				Service Commission
	4.4.3	Cons	umer Outbound Services (continued)	
		(T)	60 Block of Time	
			The monthly recurring charge is \$4.00 per BTN for	or a 60 minute block of time
			for intrastate and interstate calling. The rate i	s \$0.09 per minute for all

for intrastate and interstate calling. The rate is \$0.09 per minute for all | outbound intrastate calls completed after the 60 minute block of time has been | used. N

CANCELLED

SEP 1 5 2003 1St RS 218.5 Public Service Commission MISSOURI

Issued: August 1, 2002

ł

ł

Effective: September 1, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

FILED SEP 01 2002

Service Commission

### SECTION 4 - RATES AND CHARGES

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services (continued)
    - (U) Platinum Plans (continued)
      - .1 200 Block of Time Platinum and 1000 Minutes Nights & Weekends<sup>1</sup>

MRC	Anytime Minutes Allotted in MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$21.00	200	1000	\$0.08

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective July 9, 2003.

N N

С

Issued: June 9, 2003

Effective: July 9, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 Original Sheet 218.6 d/b/a SBC Long Distance Missouri Public **SECTION 4 - RATES AND CHARGES** Outbound Services-Switched Access (continued) REC'D NOV 0 8 2002 N 4.4 Service Commissioh Consumer Outbound Services (continued) 4.4.3 Platinum Plans (continued) (U) 200 Block of Time Platinum and 1000 Minutes Nights & Weekends .1 MRC Anytime Minutes Night/Weekend Additional Per Allotted in MRC or Off-Peak Minutes Minutes Charge Allotted in MRC Over Allotment

1000

200

Ν

# CANCELLED

\$0.08



Issued: November 8, 2002

\$21.00

Effective: December 9, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 DEC 2 3 2002

Missouri Public Service Commission

FILED DEC 23 2002

### SECTION 4 - RATES AND CHARGES

4.4 Outbound Services-Switched Access (continued)

÷

- 4.4.3 Consumer Outbound Services (continued)
  - (U) Platinum Plans (continued)
    - .2 200 Block of Time Platinum and 3000 Minutes Nights & Weekends<sup>1</sup>

MRC	Anytime Minutes Allotted in MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$23.00	200	3000	\$0.08

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations N effective July 9, 2003.

ί τ

С

Issued: June 9, 2003

Effective: July 9, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 Original Sheet 218.7 d/b/a SBC Long Distance Missouri Public **SECTION 4 - RATES AND CHARGES** 4.4 Outbound Services-Switched Access (continued) REC'D NOV 08 2002 Service Commission 4.4.3 Consumer Outbound Services (continued) Platinum Plans (continued) (U).2 200 Block of Time Platinum and 3000 Minutes Nights & Weekends Night/Weekend MRC Anytime Minutes Additional Per Allotted in MRC or Off-Peak Minutes Minutes Charge Allotted in MRC Over Allotment \$23.00 200 3000 \$0.08

Ν



Issued: November 8, 2002

Effective: December 9, 2002 DEC 2 3 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Missouri Public Service Commission

FILED DEC 23 2002



1

ī

### SECTION 4 - RATES AND CHARGES

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services (continued)
    - (U) Platinum Plans (continued)
      - .3 200 Block of Time Platinum and Unlimited Nights & Weekends<sup>1</sup>

С

MRC	Anytime Minutes Allotted in MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$24.00	200	Unlimited	\$0.08

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations
 N effective July 9, 2003.

Issued: June 9, 2003

Effective: July 9, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

		SECTION 4 - RATE	S AND CHARGES	Missouri Public
4.4	Outbound Service	es-Switched Access (con	tinued)	REC'D NOV 08 2002
	4.4.3 Consume	r Outbound Services (co	ntinued)	Service Commissic
		timum Dlang (continued)		
	(U) Pla .3	tinum Plans (continued) 200 Block of Time Pl	atinum and Unlimited I	Vights & Weekends
				Additional Per

JUL 0 9 2003 cn Pur

Issued: November 8, 2002

Effective: Desensor and the second

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

DEC 232002

Missouri Public Service Commission

FILED DEC 23 2002

### SECTION 4 - RATES AND CHARGES

4.4 Outbound Services-Switched Access (continued)

÷

- 4.4.3 Consumer Outbound Services (continued)
  - (U) Platinum Plans (continued)
    - .4 500 Block of Time Platinum and 1000 Minutes Nights & Weekends<sup>1</sup>

MRC	Anytime Minutes Allotted in MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$26.00	500	1000	\$0.07

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective July 9, 2003.

N N

Issued: June 9, 2003

Effective: July 9, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 Original Sheet 218.9 d/b/a SBC Long Distance\_ Missouri Public **SECTION 4 - RATES AND CHARGES** Outbound Services-Switched Access (continued) 4.4 REC'D NOV 0 8 2002 Service Commission Consumer Outbound Services (continued) 4.4.3 (U) Platinum Plans (continued) .4 500 Block of Time Platinum and 1000 Minutes Nights & Weekends MRC **Anytime Minutes** Night/Weekend Additional Per Allotted in MRC or Off-Peak Minutes Minutes Charge Allotted in MRC Over Allotment 500 1000 \$26.00 \$0.07

> CANCELLED JUL 0 9 2003

Issued: November 8, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 DEC 2 3 2002

Effective: December 9:2002

Missouri Public Service Commission

N

FILFD DEC 23 2002

### SECTION 4 - RATES AND CHARGES

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services (continued)
    - (U) Platinum Plans (continued)
      - .5 500 Block of Time Platinum and 3000 Minutes Nights & Weekends<sup>1</sup> C

MRC	Anytime Minutes Allotted in MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$28.00	500	3000	\$0.07

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations N effective July 9, 2003.

Issued: June 9, 2003

Effective: July 9, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

		Bell Communications Services, Inc. PSC Mo No. 1	Original Sheet 218.10
		SECTION 4 - RATES AND CHARGES	Missour Public
4.4	Outbo	und Services-Switched Access (continued)	REC'D NOV 08 2002
	4.4.3	Consumer Outbound Services (continued)	Service Commission
		(U) Platinum Plans (continued)	
		.5 500 Block of Time Platinum and 3000 Minu	utes Nights & Weekends
	<b></b>		╶╴╴╴╴

MRC	Anytime Minutes	Night/Weekend	Additional Per
1	Allotted in MRC	or Off-Peak Minutes	Minutes Charge
		Allotted in MRC	Over Allotment
\$28.00	500	3000	\$0.07

CANCELLED JUL 09 2003 JUL 0 9 218.10

Issued: November 8, 2002

Effective: December/9,2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Missouri Public Service Commission

DEC 2 3 2002

FILED DEC 23 2002



Ν

Ν

# SECTION 4 - RATES AND CHARGES 4.4 Outbound Services-Switched Access (continued) 4.4.3 Consumer Outbound Services (continued) (U) Platinum Plans (continued) .6 Intrastate Flat Rate 200 Plan The rate is \$0.11 per minute.

Issued: November 8, 2002

Effective: December 9, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

		Bell Communications Services, Inc. PSC Mo No. 1 Original S ng Distance	Sheet 218.12
		SECTION 4 - RATES AND CHARGES	
4.4	Outbo	ound Services-Switched Access (continued)	N
	4.4.3	Consumer Outbound Services (continued)	
		(U) Platinum Plans (continued)	
		.7 Intrastate Flat Rate 500 Plan	·
		The rate is \$0.10 per minute.	 N

Issued: November 8, 2002

.

•

Effective: December 9, 2002

.

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

### SECTION 4 - RATES AND CHARGES

- 4.4.3 Consumer Outbound Services (continued)
  - (U) Platinum Plans (continued)

### .8 500 Block of Time Platinum and Unlimited Nights & Weekends<sup>1</sup>

С

MRC	Anytime Minutes Allotted in MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$30.00	500	Unlimited	\$0.07

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective July 9, 2003.

N N

Issued: June 9, 2003

Effective: July 9, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

<sup>4.4</sup> Outbound Services-Switched Access (continued)

### SECTION 4 - RATES AND CHARGES

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services (continued)
    - (U) Platinum Plans (continued)

Missouri Public N RECD NOV 08 2002 Service Commission

.8 500 Block of Time Platinum and Unlimited Nights & Weekends

MRC	Anytime Minutes Allotted in MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$30.00	500	Unlimited	\$0.07

# CANCELLED

0 9 2003

Issued: November 8, 2002

Effective: Describer 0, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

DEC 232002

Missouri Public Service Commission

FLED DEC 23 2002

### 1st Revised Sheet 218.13.1 Replacing Original Sheet 218.13.1

### SECTION 4 - RATES AND CHARGES

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services (continued)
    - (U) Platinum Plans (continued)
      - .9 250 Block of Time Platinum and 1000 Minutes Nights & Weekends<sup>1</sup>

MRC	Anytime Minutes Allotted in MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$23.00	250	1000	\$0.08

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective July 9, 2003.

N N

С

Issued: June 9, 2003

Effective: July 9, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 Original Sheet 218.13.1 d/b/a SBC Long Distance SECTION 4 - RATES AND CHARGES Missouri Public

- 4.4 Outbound Services-Switched Access (continued) RECT NOV 2 6 2002 4.4.3 Consumer Outbound Services (continued) Service Commission Platinum Plans (continued)
  - .9 250 Block of Time Platinum and 1000 Minutes Nights & Weekends

MRC	Anytime Minutes Allotted in MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$23.00	250	1000	\$0.08

Ν

# CANCELLED

Issued: November 26, 2002

Effective: December 27, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Sorvice Commission

FILFD DEC 87 2002

(U)

1st Revised Sheet 218.13.2 Replacing Original Sheet 218.13.2

### SECTION 4 - RATES AND CHARGES

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services (continued)
    - (U) Platinum Plans (continued)
      - .10 400 Block of Time Platinum and 1000 Minutes Nights & Weekends<sup>1</sup>

MRC	Anytime Minutes Allotted in MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$25.00	400	1000	\$0.07

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective July 9, 2003.

N N

С

Issued: June 9, 2003

Effective: July 9, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

SECTION 4 - RATES AND CHARGES

- Outbound Services-Switched Access (continued) 4.4
  - 4.4.3 Consumer Outbound Services (continued)
    - (U)Platinum Plans (continued)
      - 400 Block of Time Platinum and 1000 Minutes Nights & Weekends .10

MRC	Anytime Minutes Allotted in MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$25.00	400	1000	\$0.07

Issued: November 26, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FILED DEC 27 201

Effective: December 27, 2002

JUL 0 9 2003 13.2

Original Sheet 218.13.2

Missouri Public

RECD NOV 2 6 2002

Service Commission

Ν

Ν



i

1st Revised Sheet 218.13.3 Replacing Original Sheet 218.13.3

### SECTION 4 - RATES AND CHARGES

- 4.4.3 Consumer Outbound Services (continued)
  - (U) Platinum Plans (continued)
    - .11 250 Block of Time Platinum and 3000 Minutes Nights & Weekends<sup>1</sup> C

MRC	Anytime Minutes Allotted in MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$24.00	250	3000	\$0.08

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations N effective July 9, 2003.

Issued: June 9, 2003

Effective: July 9, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

<sup>4.4</sup> Outbound Services-Switched Access (continued)

SECTION 4 - RATES AND CHARGES Missouri Public

- 4.4 Outbound Services-Switched Access (continued) RECD NOV 2 6 2002
  - 4.4.3 Consumer Outbound Services (continued)
    - (U) Platinum Plans (continued)
      - .11 250 Block of Time Platinum and 3000 Minutes Nights & Weekends

MRC	Anytime Minutes Allotted in MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$24.00	250	3000	\$0.08

CANCELLED

Issued: November 26, 2002

Effective: December 27, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missourl Public Service Commission

FILED DEC 27 2002

Service Commission

Ν

;

### SECTION 4 - RATES AND CHARGES

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services (continued)
    - (U) Platinum Plans (continued)
      - .12 400 Block of Time Platinum and 3000 Minutes Nights & Weekends<sup>1</sup>

MRC	Anytime Minutes Allotted in MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$27.00	400	3000	\$0.07

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective July 9, 2003.

N N

С

Issued: June 9, 2003

Effective: July 9, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Original Sheet 218.13.4

Missouri Public

RECT NOV 2 6 2002

Service Commission

### SECTION 4 - RATES AND CHARGES

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services (continued)
    - (U) Platinum Plans (continued)
      - .12 400 Block of Time Platinum and 3000 Minutes Nights & Weekends

MRC	Anytime Minutes Allotted in MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$27.00	400	3000	\$0.07

CANCELLED JUL 0 9 2003

1ssued: November 26, 2002

Effective: December 27, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FLED DEC 27 200.

Ν

ĩ

### SECTION 4 - RATES AND CHARGES

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services (continued)
    - (U) Platinum Plans (continued)

## .13 400 Block of Time Platinum and Unlimited Nights & Weekends<sup>1</sup>

С

MRC	Anytime Minutes Allotted in MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$29.00	400	Unlimited	\$0.07

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective July 9, 2003.

N N

Issued: June 9, 2003

Effective: July 9, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 Original Sheet 218.13.5 d/b/a SBC Long Distance SECTION 4 - RATES AND CHARGES **Missouri Public** 4.4 Outbound Services-Switched Access (continued) N RECD NOV 2 6 2002 4.4.3 Consumer Outbound Services (continued) Service Commission (U) Platinum Plans (continued) .13 400 Block of Time Platinum and Unlimited Nights & Weekends

MRC	Anytime Minutes Allotted in MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$29.00	400	Unlimited	\$0.07

N



Issued: November 26, 2002

Effective: December 27, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FILED DEC 27 2000
Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

;

## SECTION 4 - RATES AND CHARGES

- 4.4.3 Consumer Outbound Services (continued)
  - (U) Platinum Plans (continued)

## .14 250 Block of Time Platinum and Unlimited Nights & Weekends<sup>1</sup>

С

MRC	Anytime Minutes Allotted in MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$25.00	250	Unlimited	\$0.08

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective July 9, 2003.

N N

Issued: June 9, 2003

Effective: July 9, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

<sup>4.4</sup> Outbound Services-Switched Access (continued)

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

Original Sheet 218.13.6

Service Commission

## SECTION 4 - RATES AND CHARGES

## Missouri Public Outbound Services-Switched Access (continued) 4.4 RECT NOV 2 6 2002

- 4.4.3 Consumer Outbound Services (continued)
  - Platinum Plans (continued) (U)
    - 250 Block of Time Platinum and Unlimited Nights & Weekends .14

MRC	Anytime Minutes Allotted in MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$25.00	250	Unlimited	\$0.08

N

Ν



Issued: November 26, 2002

Effective: December 27, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public Sorvice Commission

FILED DEC 27 2002

Southwestern Bell Communications Services, Inc.	PSC Mo No. 1	1st Revised Sheet 218.13.7
d/b/a SBC Long Distance	<u> </u>	Replacing Original Sheet 218.13.7

•

- 4.4.3 Consumer Outbound Services (continued)
  - (U) Platinum Plans (continued)
    - .15 Intrastate Flat Rate 250 Plan<sup>1</sup>

The rate is \$0.11 per minute.

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations	Ν
effective July 9, 2003.	Ν

Issued: June 9, 2003

Effective: July 9, 2003

\_\_\_

С

ł

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 Original Sheet 218.13.7 d/b/a SBC Long Distance

#### SECTION 4 - RATES AND CHARGES

 4.4
 Outbound Services-Switched Access (continued)
 Missouri Public N

 4.4.3
 Consumer Outbound Services (continued)
 RECD NOV 2 6 2002

 (U)
 Platinum Plans (continued)
 Service Commissic

 .15
 Intrastate Flat Rate 250 Plan

The rate is \$0.11 per minute.

Issued: November 26, 2002

Effective: December 27, 2002

CANCELLED

JUL 0 9 2003

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Servicé Commission

FILED DEC 27 2002

N

Southwestern Bell Communications Services, Inc.	PSC Mo No. 1	1st Revised Sheet 218.13.8
d/b/a SBC Long Distance		Replacing Original Sheet 218.13.8

4.4 Outbound Services-Switched Access (continued)

-- --

-

÷

- 4.4.3 Consumer Outbound Services (continued)
  - (U) Platinum Plans (continued)
    - .16 Intrastate Flat Rate 400 Plan<sup>1</sup>

The rate is \$0.10 per minute.

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations	Ν
effective July 9, 2003.	Ν

Issued: June 9, 2003

Effective: July 9, 2003

С

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

## SECTION 4 - RATES AND CHARGES

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services (continued)
    - (U) Platinum Plans (continued)

.16 Intrastate Flat Rate 400 Plan

The rate is \$0.10 per minute.





Issued: November 26, 2002

Effective: December 27, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FILED DEC 27 2002

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

#### 1st Revised Sheet 218.13.9 Replacing Original Sheet 218.13.9

## SECTION 4 - RATES AND CHARGES

÷

- 4.4.3 Consumer Outbound Services (continued)
  - (U) Platinum Plans (continued)
    - .17 1000 Block of Time Nights and Weekends<sup>1</sup>

	С

MRC	Night/Weekend or Off-Peak Minutes	Additional Per Minutes Charge
	Allotted in MRC	Over Allotment
\$17.00	1000	\$0.07

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective July 9, 2003.

N N

Issued: June 9, 2003

Effective: July 9, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

<sup>4.4</sup> Outbound Services-Switched Access (continued)

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 Original Sheet 218.13.9 d/b/a SBC Long Distance

\$17.00

## SECTION 4 - RATES AND CHARGES

4.4 Outbo		und Se	ervices-Switched Access	(continued)	Missouri	Public <sup>N</sup>
	4.4.3	Cons	sumer Outbound Service	s (continued)	RECTINOV 2	6 2002
J)			Platinum Plans (contin	ued)	Service Com	mission
		.17 1000 Block of T	ime Nights and Weekend	ls		
			MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment	

1000

CANCELLED JUL 092003 JUL 092003 Public Services Construction

\$0.07

Issued: November 26, 2002

1

ł

Effective: December 27, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

N

FILED DEC 27 2002

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance 1st Revised Sheet 218.13.10 Replacing Original Sheet 218.13.10

## SECTION 4 - RATES AND CHARGES

÷

- 4.4.3 Consumer Outbound Services (continued)
  - (U) Platinum Plans (continued)

#### .18 3000 Block of Time Nights and Weekends1

MRCNight/Weekend or<br/>Off-Peak MinutesAdditional Per<br/>Minutes Charge<br/>Over Allotment\$18.503000\$0.07

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective July 9, 2003.

N N

Issued: June 9, 2003

Effective: July 9, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

<sup>4.4</sup> Outbound Services-Switched Access (continued)

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 Original Sheet 218.13.10 d/b/a SBC Long Distance

## SECTION 4 - RATES AND CHARGES

Missouri PublicN 4.4 Outbound Services-Switched Access (continued) RECT NOV 2 6 2002 4.4.3 Consumer Outbound Services (continued) Platinum Plans (continued) Service Commission (U) 3000 Block of Time Nights and Weekends .18 MRC Night/Weekend or Additional Per **Off-Peak Minutes** Minutes Charge Over Allotment

CANCELLED JUL 092003

\$0.07

Issued: November 26, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588



Effective: December 27, 2002

FILED DEC 27 2002

Allotted in MRC

3000

\$18.50



N

\_

:

### SECTION 4 - RATES AND CHARGES

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services (continued)
    - (U) Platinum Plans (continued)
      - .19 Unlimited Nights and Weekends<sup>1</sup>

MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$20.00	Unlimited	\$0.07

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective July 9, 2003.

N N

С

Issued: June 9, 2003

Effective: July 9, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

Original Sheet 218.13.11

Service Commisside

SECTION 4 - RATES AND CHARGES

- 4.4 Outbound Services-Switched Access (continued) Missouri Public N 4.4.3 Consumer Outbound Services (continued) RECD NOV 2 6 2002
  - 4.4.3 Consumer Outbound Services (continued)
    - (U) Platinum Plans (continued)
      - .19 Unlimited Nights and Weekends

MRC	Night/Weekend or Off-Peak Minutes Allotted in MRC	Additional Per Minutes Charge Over Allotment
\$20.00	Unlimited	\$0.07

Ν

CANCELLED 218.13.11

Issued: November 26, 2002

Effective: December 27, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FILED DE.C 2 7 2002

Southwestern Bell Communications Services, Inc.PSC Mo. - No. 11st Revised Sheet 218.14d/b/a SBC Long Distance\_\_\_\_\_\_\_\_\_\_\_\_Replacing Original Sheet 218.14

. . . . . .

## SECTION 4 - RATES AND CHARGES

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services (continued)
    - (V) ValueSaver

Peak			Off-Peak				
Initial		Additional		Initial Additional			
Period		Period		Period		Period	
\$0.07	R	\$0.07	R	\$0.07	R	\$0.07	R

Issued: February 25, 2003

Effective: March 30, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

4.4.3

Outbound Services-Switched Access (continued)

ValueSaver (V)

Southwestern Bell Communications Services, Inc.

4.4

i

REC'D NOV 08 2002 ervice Commission

Ν

N

Consumer Outbound Services (continued)	1
	Se

	Peak	Off-Peak		
Initial Period	Additional Period	Initial Period	Additional Period	
\$0.27	\$0.27	\$0.17	\$0.17	

PSC Mo. - No. 1

CANCELLED

MAR 3 0 2003 ISSION

Issued: November 8, 2002

Effective: Engennes 7, 20020

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 DEC 2 3 2002

Missouri Public Service Commission

FILED DEC 23 2002



Southwestern Bell Communications Services, Inc.	PSC Mo No. 1	2nd Revised Sheet 218.15
d/b/a SBC Long Distance		Replacing 1st Revised Sheet 218.15

4.4 Outbound Services-Switched Access (continued)

.

.

4.4.3 Consumer Outbound Services (continued)

(W)	250 Block of Time	and 400 Block of Time <sup>1</sup>	C

.1 Intrastate Flat Rate 250 Gold Plan<sup>1</sup>

The rate is \$0.11 per minute.

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations	Ν
effective July 9, 2003.	Ν

Issued: June 9, 2003

Effective: July 9, 2003

 $\mathbf{C}$ 

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc.PSC Mo. - No. 11st Revised Sheet 218.15d/b/a SBC Long DistanceReplacing Original Sheet 218.15

## SECTION 4 - RATES AND CHARGES

4.4	Outbo	und Services-Switched Access (continued)	Missouri Public Service Commission		
	4.4.3	Consumer Outbound Services (continued)	RECT) APR 15 2003		
		(W) 250 Block of Time and 400 Block of Time	Т		

The rate is \$0.11 per minute.

.1

Intrastate Flat Rate 250 Gold Plan

CANCELLED

JUL 0 9 2003

Public Service Construction

Issued: April 15, 2003

Effective: May 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Missouri Public Service Commission

FILED MAY 16 2003

(W) Gold Plans

.1 Intrastate Flat Rate 250 Gold Plan

The rate is \$0.11 per minute.

CANCELLED

MAY 16 2003 Public Service Commission

Issued: November 26, 2002

Effective: December 27, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission FILFD DEC 27 2002

N

Ν

Service Commission

Southwestern Bell Communications Services, Inc.	PSC Mo No. 1	2nd Revised Sheet 218.16
d/b/a SBC Long Distance		Replacing 1st Revised Sheet 218.16

4.4 Outbound Services-Switched Access (continued)

تىر

7

4.4.3 Consumer Outbound Services (continued)

(W) 250 Block of Time<sup>1</sup> and 400 Block of Time<sup>1</sup> (continued) C

.2 Intrastate Flat Rate 400 Gold Plan<sup>1</sup>

The rate is \$0.10 per minute.

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations N effective July 9, 2003.

Issued: June 9, 2003

Effective: July 9, 2003

С

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

d/b/a SBC Long Dista	nce	Replacing Original Sheet 218.16
	SECTION 4 - RATES AND CHARGI	ES Missouri Public Service Commission
4.4 Outbound Ser	rvices-Switched Access (continued)	RECT) APR 15 2003
4.4.3 Consu	umer Outbound Services (continued)	
(W)	250 Block of Time and 400 Block of Time (c	continued) T
	.2 Intrastate Flat Rate 400 Gold Plan	
	The rate is \$0.10 per minute.	

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1

CANCELLED JUL 0 9 2003 JUL 0 9 2003 JUL 0 9 2003 Public Providence

Issued: April 15, 2003

i

Effective: May 16, 2003

1st Revised Sheet 218.16

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Comm**ission** 

FILED MAY 16 2003

d/b/a	SBC Long Distance	
	SECTION 4 - RATES AND CHARGES	Ata and Rublin
4.4	Outbound Services-Switched Access (continued)	Missouri Public
	4.4.3 Consumer Outbound Services (continued)	RECD NOV 2 6 2002
	(W) Gold Plans (continued)	Service Commissid
	.2 Intrastate Flat Rate 400 Gold Plan	1
	The rate is \$0.10 per minute.	: 1

PSC Mo. - No. 1

Southwestern Bell Communications Services, Inc.



Issued: November 26, 2002

ł

Effective: December 27, 2002

Original Sheet 218.16

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FLED DEC 27 2002

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services (continued)
    - (X) Connections Services
      - .1 National Connections

The monthly recurring charge is \$20.00 for unlimited interstate and intrastate MOU as defined in Section 3.4.3 (X).4.a of this Tariff.

- .2 Reserved for future use
- .3 National Connections Plus

The monthly recurring charge is \$20.00 for unlimited interstate and intrastate MOU as defined in Section 3.4.3 (X).4.c of this Tariff.

.4 National Connections II

The monthly recurring charge is \$30.00 for unlimited interstate and intrastate MOU as defined in Section 3.4.3 (X).4.d of this Tariff.

.5 National Connections Preferred<sup>1</sup>

The monthly recurring charge is \$20.00 for unlimited interstate and intrastate MOU, as defined in Section 3.4.3 (X).4.e of this Tariff. Every third bill cycle after the first full bill cycle, the MRC will be billed at a reduced rate of \$15.00, for the first 24 months the customer continuously subscribes to this calling plan.

This Service is no longer available to new Customers or to existing Customers at new locations effective January 24, 2005.

С

N N

Issued: December 23, 2004

Effective: January 24, 2005

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

1

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services (continued)
    - (X) Connections Services
      - .1 National Connections

JAN 2 4 2005 By GUNRS 218.17 Public Service Commission

CANCELLED

The monthly recurring charge is \$20.00 for unlimited interstate and intrastate MOU as defined in Section 3.4.3 (X).4.a of this Tariff.

- .2 Reserved for future use
- .3 National Connections Plus

The monthly recurring charge is \$20.00 for unlimited interstate and intrastate MOU as defined in Section 3.4.3 (X).4.c of this Tariff.

.4 National Connections II

The monthly recurring charge is \$30.00 for unlimited interstate and intrastate MOU as defined in Section 3.4.3 (X).4.d of this Tariff.

.5 National Connections Preferred

The monthly recurring charge is \$20.00 for unlimited interstate and intrastate MOU, as defined in Section 3.4.3 (X).4.e of this Tariff. Every third bill cycle after the first full bill cycle, the MRC will be billed at a reduced rate of \$15.00, for the first 24 months the customer continuously subscribes to this calling plan.

Issued: November 3, 2004

Effective: December 6, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

December 10, 2004

Ν

Ν

.1+

				SECTION 4 - RATES AND CHARGES	Missouri Public
4.4	Outbo	und Se	rvices	-Switched Access (continued)	
	4.4.3	Cons	umer	Outbound Services (continued)	REC'D NOV 1 8 2003 Service Commission
		(X)	Con	nections Services	Service Commission
			.1	National Connections	
				The monthly recurring charge is \$20.00 for unline intrastate MOU as defined in Section 3.4.3 (X).4.	
			.2	Reserved for future use	
			.3	National Connections Plus	
				The monthly recurring charge is \$20.00 for unlin intrastate MOU as defined in Section 3.4.3 (X).4	
			.4	National Connections II	
				The monthly recurring charge is \$30.00 for unlin intrastate MOU as defined in Section 3.4.3 (X).4	
				CANCEL	LED D
				DEC 1 0 By Surres Public Service Co MISSOU	2004 218.17

Issued: November 18, 2003

Effective: December 18, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FILED DEC 18 2003

			cations Services, Inc. PSC Mo No. 1 3rd Revised Sheet 218.17 Replacing 2nd Revised Sheet 218.17	
			SECTION 4 - RATES AND CHARGES	
Outbou	ind Se	rvices		
			RECD OCT 1 4 2003	
4.4.3	Cons	sumer	•	
	(X)	Con		I
		.1	National Connections	
			The monthly recurring charge is \$20.00 for unlimited interstate and intrastate MOU as defined in Section 3.4.3 (X).4.a of this Tariff.	
		.2	Reserved for future use	
		.3	National Connections Plus	
			The monthly recurring charge is \$20.00 for unlimited interstate and intrastate MOU as defined in Section 3.4.3 (X).4.c of this Tariff.	
		.4	National Connections II	
			The monthly recurring charge is \$30.00 for unlimited interstate and intrastate MOU as defined in Section 3.4.3 (X).4.d of this Tariff.	
Amita	FA	.5	JustCall <sup>sM</sup> Worldwide	N
			The monthly recurring charge is \$22.95 for unlimited interstate and intrastate MOU and International rates to 200 countries as defined in Section 3.4.3 (X).4.e of this Tariff.	     1
	Outbou 4.4.3	Outbound Se 4.4.3 Cons	<ul> <li>4.4.3 Consumer</li> <li>(X) Con</li> <li>.1</li> <li>.2</li> <li>.3</li> <li>.4</li> </ul>	SECTION 4 - RATES AND CHARGES Outbound Services-Switched Access (continued) 4.4.3 Consumer Outbound Services (continued) 4.4.3 Consumer Outbound Services (continued) (X) Connections Services 1 National Connections The monthly recurring charge is \$20.00 for unlimited interstate and intrastate MOU as defined in Section 3.4.3 (X).4.a of this Tariff. 2 Reserved for future use 3 National Connections Plus The monthly recurring charge is \$20.00 for unlimited interstate and intrastate MOU as defined in Section 3.4.3 (X).4.c of this Tariff. 4 National Connections II The monthly recurring charge is \$30.00 for unlimited interstate and intrastate MOU as defined in Section 3.4.3 (X).4.d of this Tariff. 5 JustCall <sup>SM</sup> Worldwide The monthly recurring charge is \$22.95 for unlimited interstate and intrastate MOU and International rates to 200 countries as defined in

Issued: October 14, 2003

ł

2

i

ļ

Effective: November 15, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Missouri Public Service Commission

FILED NOV 1 5 2003

<u>d/b/a (</u>	SBC Loi	ng Dist	ance	Replaci	ng 1st Revised Sheet 218.17	<u>-</u>
				SECTION 4 - RATES AND CHARGES	و و و و مرم مرم مرم و	
4.4	Outbo	und Se	ervice	s-Switched Access (continued)	RECTD JUL 282	003
	4.4.3	Cons	sumer	Outbound Services (continued)	Service Commis	ssion
		(X)	Con	nections Services		
			.1	National Connections		
				The monthly recurring charge is \$20.00 for intrastate MOU as defined in Section 3.4.3		
			.2	Reserved for future use		
			.3	National Connections Plus		
				The monthly recurring charge is \$20.00 for intrastate MOU as defined in Section 3.4.3		
			.4	National Connections II		N
				The monthly recurring charge is \$30.00 for intrastate MOU as defined in Section 3.4.3		 N
					Missouri	Public
				CANCELLED	FILED SEP	<b>0</b> 1 2003
	·			NOV 1 5 2003 By 3rd RS 218.17	Service Cor	nmissior -
Issued	i: July 2	.8, 200	3	Public Service Loganassion MISSINIPI	ffective: September 1, 2003	-

ł

ĺ

I

.

ì

.

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 d/b/a\_SBC Long Distance Replacing Original Sheet 218.17 SECTION 4 - RATES AND CHARGES Missouri Public Service Commission Outbound Services-Switched Access (continued) 4.4RECTD MAY 1 6 2003 Consumer Outbound Services (continued) 4.4.3 **Connections Services**  $(\mathbf{X})$ .1 National Connections The monthly recurring charge is \$20.00 for unlimited interstate and intrastate MOU as defined in Section 3.4.3 (X).4.a of this Tariff. .2 Reserved for future use N National Connections Plus .3 The monthly recurring charge is \$20.00 for unlimited interstate and intrastate MOU as defined in Section 3.4.3 (X).4.c of this Tariff. Ν

PSC Mo. - No. 1

Southwestern Bell Communications Services, Inc.

# CANCELLED

SEP 01 2003 Public Service Commission MISSOURI

Issued: May 16, 2003

Effective: June 16, 2003

1st Revised Sheet 218.17

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FILED JUN 16 2003

Southwester d/b/a SBC L	ell Communications Services, Inc. PSC Mo No. 1 Original Sheet 218.17 Distance	
	SECTION 4 - RATES AND CHARGES Missouri Public Service Commission RECD FEB 28 201	
4.4.3	Consumer Outbound Services (continued)	
	(X) Connections Services	
	.1 National Connections	ļ
	The monthly recurring charge is \$20.00 for unlimited interstate and intrastate MOU as defined in Section 3.4.3 (X).4.a of this Tariff.	   N

## CANCELLED



Issued: February 28, 2003

ţ

Effective: March 30, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FILED MAR 30 2003

### 4.4 Outbound Services-Switched Access (continued)

- 4.4.3 Consumer Outbound Services (continued)
  - (Y) Value Plus<sup>1</sup>
    - .1 Flat Rate Options
      - .a Value Plus Flat Rate

The rate is \$0.07 per minute.

For Customers subscribing to Value Plus Flat Rate, the Customers will pay a Monthly Recurring Charge of \$2.00.

.b Value Plus 60

The rate is \$0.07 per minute.

.c Value Plus 200

The rate is \$0.07 per minute.

.d Value Plus 500

The rate is \$0.07 per minute.

<sup>1</sup> This Service is not longer available to new Customers or to existing Customers at new N locations effective August 1, 2004. N

Issued: July 1, 2004

Effective: August 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 С

	western SBC Lor			nications	Services, Inc.	PSC Mo No. 1	1st Revised Sheet 218.18 Replacing Original Sheet 218.18
_,				SEC	TION 4 - RA'	TES AND CHAR	Missouri Public
4.4	Outbo	und Se	ervice	s-Switcl	hed Access (c	ontinued)	REC'D MAY 06 2004
	4.4.3	Cons	sumer	Outbou	und Services (	continued)	
		(Y)	Val	ue Plus			
			.1	Flat R	late Options		
				.a	Value Plus	Flat Rate	
					The rate is S	\$0.07 per minute.	
						-	Value Plus Flat Rate, the Recurring Charge of \$2.00.
				.b	Value Plus	60	
					The rate is !	\$0.07 per minute.	
				.c	Value Plus	200	
					The rate is !	\$0.07 per minute.	CANCELLED
				.d	Value Plus	500	AUG 0 1 2004 Hy 2rd RS 218.18
					The rate is	\$0.07 per minute.	Public Service Commission MISSOURI

Issued: May 6, 2004

Effective: June 8, 2004

Missouri Public Service Commission

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

FILED JUN 08 2004

	western H SBC Lon			cations S	Services, Inc.	PSC Mo No. 1	<u>-</u>	Original She	et 218.18	
				SEC	TION 4 - RA	ATES AND CHARC	GES	Missouri Pu	ıblic	
4.4	Outbo	und Se	rvices	-Switch	ed Access (co	ontinued)		RECTO SEP 08	2003	N
	4.4.3	Cons	sumer	Outbou	nd Services (	continued)	Se	ervice Comm	nission	ł
		(Y)	Valu	e Plus						(
			.1	Flat R	ate Options					}
				.a	Value Plus	Flat Rate				}
					The rate is	\$0.07 per minute.				
				.b	Value Plus	60				
					The rate is	\$0.07 per minute.				{
				.c	Value Plus	200				}
					The rate is	\$0.07 per minute.				 
				.d	Value Plus	: 500				
					The rate is	\$0.07 per minute.				[ N

CANCELLED

JUN 0 8 2004 By <sup>/S+</sup> RS 2/8-18 Public Service Commission

Issued: September 8, 2003

e

1

T

Effective: October 8, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Servico Commission

FILED OCT 08 2003

## 4.4 Outbound Services-Switched Access (continued)

- 4.4.3 Consumer Outbound Services
  - (Z) Block of Time II
    - .1 60 Block of Time II

The MRC is \$5.00 per BTN. The rate is \$0.07 per minute for all 1+ Direct-Dialed outbound intrastate calls completed after the 60 minute block of time has been used.

.2 200 Block of Time II

The MRC is \$12.00 per BTN. The rate is \$0.07 per minute for all 1+ Direct-Dialed outbound intrastate calls completed after the 200 minute block of time has been used.

.3	500 Block of Time II	
----	----------------------	--

The rate is \$0.07 per minute.

Issued: October 2, 2003

Effective: November 1, 2003

Т

Ν

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Ν

N

SECTION 4 - RATES AND CHARGES Missouri Public
4.4 Outbound Services-Switched Access (continued) RECD AUG 1 4 2003
4.4.3 Consumer Outbound Services (continued) Service Commission
(Z) Block of Time II
.1 60 Block of Time II
The MRC is \$5.00 per BTN. The rate is \$0.07 per minute for all 1+ Direct-Dialed outbound intrastate calls completed after the 60 minute

.2 200 Block of Time II

block of time has been used.

The MRC is \$12.00 per BTN. The rate is \$0.07 per minute for all 1+ Direct-Dialed outbound intrastate calls completed after the 200 minute block of time has been used.

.3 Reserved for future use.

## CANCELLED

NOV 0 1 2003 by IST RS 218.19 Public Service Comunission MISSOURI

Issued: August 14, 2003

Effective: September 15, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Sorvice Commission

FILED SEP 1 5 2003

Southwestern Bell Communications Services, Inc.	PSC Mo No. 1	2nd Revised Sheet 218.20
d/b/a SBC Long Distance		Replacing 1st Revised Sheet 218.20

4.4	Outbound Services-Switched Access (continued)						
			}				
	4.4.3	Consumer Outbound Services (continued)	{				
			ļ				
		(AA) Reserved for future use	N				

Issued: December 16, 2003

Effective: January 15, 2004

l

1

ì

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance 1st Revised Sheet 218.20 Replacing Original Sheet 218.20

## SECTION 4 - RATES AND CHARGES

RECT NOV 85 2003

D

Service Commission

# CANCELLED

JAN 1 5 2004 By AN RS 218.20 Public Service Commission MISSOURI

Issued: November 5, 2003

Effective: December 5, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Missouri Public Servico Commission

D

FILED DEC 05 2003

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance Original Sheet 218.20

SECTION 4 - RATES AND CHARGES							
4.4	Outbo	und Services	s-Switched Access (continued)	Missouri Public N			
	4.4.3	Consumer	Outbound Services (continued)	RECD AUG 1 4 2003			
		(AA) Min	uteShare	Service Commission			
		.1	MinuteShare Home 300/5000				
			The MRC is \$31.99. The rate is \$0. the block of time has been exhausted	- ,			
		.2	MinuteShare Home 400/5000				
			The MRC is \$40.99. The rate is \$0. the block of time has been exhausted				
		.3	MinuteShare Home 600/5000				
			The MRC is \$49.99. The rate is \$0. the block of time has been exhausted	· · · · · · · · · · · · · · · · · · ·			
		.4	MinuteShare Home 900/5000				
			The MRC is \$67.99. The rate is \$0. the block of time has been exhausted				
		.5	MinuteShare Home 1400/5000				
			The MRC is \$94.99. The rate is \$0. the block of time has been exhausted	-			
			CANCELLED				
	DEC 0 5 2003 Missouri Public Sorvico Commission						
			Public Service Construction	FLED SEP 13 2003			

Issued: August 14, 2003

Effective: September 13, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services (continued)

(AB) Just Call 3 cents<sup>1</sup>

The usage rate is \$0.03 per minute.

<sup>1</sup> This Service is no longer available to new Customers effective April 12, 2004.

N

С

Issued: March 10, 2004

Effective: April 12, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588
Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u> SECTION 4 - RATES AND CHARGES 4.4 Outbound Services-Switched Access (continued) 4.4.3 Consumer Outbound Services (continued) (AB) Just Call 3 cents The usage rate is \$0.03 per minute.

## CANCELLED

N

1

N

APR 1 2 2004 By 3rd RS 2/8.2/ Public Service Commission MISSOURI

Issued: December 16, 2003

Effective: January 15, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missourl Public Service Commission





Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance 1st Revised Sheet 218.21 Replacing Original Sheet 218.21

#### SECTION 4 - RATES AND CHARGES

Missouri Public D

REC'D NOV 05 2003

Service Commission

## CANCELLED

JAN 1 5 2004 By JN4 LS 218.21 Public Service Commission MISSOURI

Issued: November 5, 2003

Effective: December 5, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Public Service Commission

FILED DEC 0 5 2003



		Bell Commun	ications Services, Inc. PSC Mo No.	1 Original Sheet 218.21
			SECTION 4 - RATES AND CHA	RGES
4.4	Outbo	und Services	-Switched Access (continued)	Missouri Public
	4.4.3	Consumer	Outbound Services (continued)	RECTD AUG 1 4 2003
		(AA) Min	uteShare (continued)	Service Commission
		.6	Reserved for future use	
		.7	Reserved for future use	
		.8	Reserved for future use	
		.9	Reserved for future use	
		.10	Reserved for future use	
		.11	Reserved for future use	

CANCELLED

DEC 0 5 2003 SICR **b**hp<sub>l</sub>

Missouri Public Sorvico Commission

FILED SEP 13 2003

Issued: August 14, 2003

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Effective: September 13, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc.	PSC Mo No. 1	2nd Revised Sheet 218.22
d/b/a SBC Long Distance		Replacing 1st Revised Sheet 218.22

• • • • • • • • •

- ----

### SECTION 4 - RATES AND CHARGES

Outbound Services-Switched Access (continued)						
		i				
4.4.3	Consumer Outbound Services (continued)					
	(AC) Reserved for future use	N				
		4.4.3 Consumer Outbound Services (continued)				

Issued: December 11, 2003

- ----

Effective: January 12, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public D

REC'D NOV 05 2003

Service Commission

### CANCELLED

JAN 1 2 2004 By 2<sup>nd</sup> RS 218.22 Public Service Commission MISSOURI

Issued: November 5, 2003

Effective: December 5, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Public Service Commission

FILED DEC 0 5 2003

D

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

			SECTION 4 - RATES AND C	HARGES	
4.4	Outbo	und Services	s-Switched Access (continued)	Missouri Public	N
	4.4.3	Consumer	Outbound Services (continued)	RECTD AUG 1 4 2003	
		(AA) Minu	uteShare (continued)	Service Commission	
		.12	MinuteShare CPN 250/5000		
			The MRC is \$33.49. The rate is the block of time has been exha	s \$0.45 per minute for all MOU after usted.	
		.13	MinuteShare CPN 350/5000		
			The MRC is \$42.99. The rate is the block of time has been exha	s \$0.45 per minute for all MOU after usted.	   
		.14	MinuteShare CPN 500/5000		}
			.a Option 1		
				e rate is \$0.39 per minute for all time has been exhausted.	   N

## CANCELLED

DEC 0 5 2003 Ly ISTUS 219.82 Public Service Commission MISSOURI

> Missouri Public Service Commission

FILED SEP 13 2003

Issued: August 14, 2003

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Effective: September 13, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

Ν

Ν

#### SECTION 4 - RATES AND CHARGES

4.4 Outbound Services-Switched Access (continued)

......

4.4.3 Consumer Outbound Services (continued)

(AC) Simply Talk<sup>SM</sup> 5 Cents

The usage rate is \$0.05 per minute. For the interstate MRC, see Section 4.4.3 (AC) of the Company's interstate Voice Product Reference and Pricing Guide which may be found at www.sbc.com.

Issued: December 12, 2003

Effective: January 15, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

4.4 Outbound Services-Switched Access (continued)

> Consumer Outbound Services (continued) 4.4.3

> > (AD) JustCall<sup>SM</sup> Unlimited Weekends<sup>1</sup>

С

The usage rate for peak rate period MOU is \$0.07 per minute. The MRC is \$14.95 for an unlimited block of interstate and intrastate off-peak period MOU as defined in Section 3.4.3 (AD).4 of this Tariff.

1 This Service is not longer available to new Customers or to existing Customers at new locations effective August 1, 2004. Ν

Ν

Issued: July 1, 2004

Effective: August 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

		Bell Communications Services, Inc.		3rd Revised Sheet 218.23 ing 2nd Revised Sheet 218.23
		SECTION 4 - RAT	TES AND CHARGES	Missouri Public
4.4	Outbo	und Services-Switched Access (co	ontinued)	RECD FEB 03 2004
	4.4.3	Consumer Outbound Services (o		Service Commission
		(AD) JustCallK Unlimited Wee		

The usage rate for peak rate period MOU is \$0.07 per minute. The MRC is \$14.95 for an unlimited block of interstate and intrastate off-peak period T MOU as defined in Section 3.4.3 (AD).4 of this Tariff.

## CANCELLED

AUG 0.1 2004 By 44425 218.23 Public Service Commission MISSOURI

Issued: February 3, 2004

Effective: March 5, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Public Service Commission



Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance 2nd Revised Sheet 218.23 Relacing 1st Revised Sheet 218.23

### **SECTION 4 - RATES AND CHARGES**

4.4	Outbo	utbound Services-Switched Access (continued) Missouri Publ			
	4.4.3	Consumer Outbound Services (continued)	<b>RECD DEC</b> 11 2003	{	
		(AD) JustCall <sup>sM</sup> Unlimited Weekends	Service Commission		
		The usage rate for peak rate period MOU	J is \$0.07 per minute. The MRC is	1	
		\$14.95 for an unlimited block of intersta	te and intrastate peak period MOU as	]	
		defined in Section 3.4.3 (AD).4 of this T	ariff.	Ν	

### CANCELLED

### MAR 0 5 2004 3ª RS 218.23 Public Service Commission MISSOURI

Issued: December 11, 2003

Effective: January 12, 2004

į.

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED JAN 12 2004

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance 1st Revised Sheet 218.23 Replacing Original Sheet 218.23

#### SECTION 4 - RATES AND CHARGES

Missouri Public D RECD NOV 05 2003

### CANCELLED

JAN 1 2 2004 By 2nd RS 218 23 Public Service Commission MISSOURI

Issued: November 5, 2003

Effective: December 5, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Missouri Public Service Commission

FILED DEC 05 2003

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u>

Original Sheet 218.23

			SECTION 4 - RATES AND CH		
4.4	Outbo	und Services	s-Switched Access (continued)	Missouri Public	N
	4.4.3	Consumer	Outbound Services (continued)	RECT AUG 1 4 2003	
		(AA) Min	uteShare (continued)	Service Commission	
		.15	Reserved for future use		
		.16	MinuteShare CPN 900/5000		
	The MRC is \$71.49. The rate is \$0.39 per minute for all MOU after the block of time has been exhausted.				
		.17	Reserved for future use	<i>,</i>	l N



Missouri Public Sarvico Commission

FILED SEP 13 2003

Issued: August 14, 2003

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Effective: September 13, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

4.4	Outbo	tbound Services-Switched Access (continued)			
	4.4.3	Consumer	Outbo	und Services	
		(AE) Just	Call <sup>sm</sup>		
		.1	Rese	erved For Future Use	
		.2	Rese	erved For Future Use	
		.3	Rese	erved For Future Use	
		.4	Rese	erved For Future Use	
		.5	Just	Call <sup>sm</sup> Standard Options	
			.a	JustCall <sup>sm</sup> 5 Cents Standard	
				The usage rate is \$0.05 per minute. For the interstate MRC, see Section 4.4.3 (AE) of the Company 's interstate Voice Product Reference and Pricing Guide, which may be found at www.sbc.com.	
			.b	JustCall <sup>sm</sup> 9 Cents Standard	
				The usage rate is \$0.09 per minute. For the interstate MRC, see Section 4.4.3 (AE) of the Company's interstate Voice Product Reference and Pricing Guide, which may be found at www.sbc.com.	     N

Issued: February 27, 2004

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance 1st Revised Sheet 218.24 Replacing Original Sheet 218.24

#### **SECTION 4 - RATES AND CHARGES**



CANCELLED

APR 0 1 2004 By 2<sup>nd</sup> RS 218 24 Public Service Commission MISSOURI

Issued: November 5, 2003

Effective: December 5, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED DEC 0 5 2003

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u>

Original Sheet 218.24

			SECTION 4 - RATES AND CHA	ARGES Missouri Public	
4.4	Outbo	und Services	s-Switched Access (continued)		Ν
	4.4.3	Consumer	Outbound Services (continued)	REC'D AUG 14 2003	{
				Service Commission	ļ
		(AA) Min	inssion		
		.18	MinuteShare CPN 1200/5000		
					Ì
	The MRC is \$99.99. The rate is \$0.39 per minute for all MOU at the block of time has been exhausted.			•	
					Ì
		.19	Reserved for future use		Ν



Missouri Public Service Commission

FILED SEP 13 2003

Issued: August 14, 2003

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Effective: September 13, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

			SEC	TION 4 - RATES AND CHARGES			
4.4	Outbou	und Service	s-Switcl	hed Access (continued)	Ν		
	4.4.3	Consumer	Outbou	and Services (continued)			
		(AE) Just	Call <sup>sm</sup>	Call <sup>sm</sup>			
		.6	JustC	all <sup>sm</sup> Standard Block of Time Options			
			.a	JustCall <sup>sm</sup> 60 Standard			
				The MRC is \$4.00 per BTN for a 60 minute block of time as defined in Section 3.4.3 (AE).6.a of this Tariff. The rate is \$0.07 per minute for all outbound intrastate calls completed after the 60 minute block of time has been used.			
			.b	JustCall <sup>sm</sup> 200 Standard			
				The MRC is \$10.00 per BTN for a 200 minute block of time as defined in Section 3.4.3 (AE).6.b of this Tariff. The rate is \$0.07 per minute for all outbound intrastate calls completed after the 200 minute block of time has been used.			
			.c	JustCall <sup>sm</sup> 400 Standard			
				The MRC is \$15.00 per BTN for a 400 minute block of time as defined in Section 3.4.3 (AE).6.c of this Tariff. The rate is \$0.07 per minute for all outbound intrastate calls completed after the 400 minute block of time has been used.	     N		

Issued: February 27, 2004

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

			SEC	CTION 4 - RATES AND CHARGES	
4.4	Outbo	und Service	es-Switc	hed Access (continued)	N
	4.4.3	Consume	r Outbo	und Services (continued)	
		(AE) Jus	tCall <sup>sm</sup>		
		.7	JustC	Call <sup>sm</sup> Standard II Options	
			a.	JustCall <sup>sm</sup> Standard II	
				The intrastate usage rate is \$0.21 per minute. For the interstate usage rate and interstate MRC, see Section 4.4.3 (AE) of the Company 's interstate Voice Product Reference and Pricing Guide, which may be found at www.sbc.com.	
		.8	JustC	Call <sup>sm</sup> Standard II Block of Time Options	
			.a	JustCall <sup>sm</sup> 60 Standard II	
				The MRC is \$3.00 per BTN for a 60 minute block of time as defined in Section 3.4.3 (AE).8.a of this Tariff. The rate is \$0.21 per minute for all outbound intrastate calls completed after the 60 minute block of time has been used.	
			.b	JustCall <sup>sm</sup> 200 Standard II	
				The MRC is \$8.00 per BTN for a 200 minute block of time as defined in Section 3.4.3 (AE).8.b of this Tariff. The rate is \$0.21 per minute for all outbound intrastate calls completed after the 200 minute block of time has been used.	     N

Issued: February 27, 2004

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

### **SECTION 4 - RATES AND CHARGES** 4.4 Outbound Services-Switched Access (continued) Ν 4.4.3 Consumer Outbound Services (continued) (AE) JustCall<sup>sm</sup> .8 JustCall<sup>sm</sup> Standard II Block of Time Options (continued) JustCall<sup>sm</sup> 400 Standard II .c The MRC is \$12.00 per BTN for a 400 minute block of time as defined in Section 3.4.3 (AE).8.c of this Tariff. The rate is \$0.21 per minute for all outbound intrastate calls completed after the 400 minute block of time has been used. Ν

Issued: February 27, 2004

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

		SEC	CTION 4 - RATES AND CHARGES			
Outbo	und Service	s-Swite	ched Access (continued)	Ν		
4.4.3	Consumer Outbound Services					
	(AE) JustCall <sup>sm</sup>					
	.9	JustC	Call <sup>sm</sup> Preferred Options			
		.a	JustCall <sup>sm</sup> 3 Cents Preferred			
			The usage rate is \$0.03 per minute. For the interstate MRC, see Section 4.4.3 (AE) of the Company 's interstate Voice Product Reference and Pricing Guide, which may be found at www.sbc.com.			
		.b	JustCall <sup>sm</sup> 7 Cents Preferred			
			The usage rate is \$0.07 per minute. For the interstate MRC, see Section 4.4.3 (AE) of the Company 's interstate Voice Product Reference and Pricing Guide, which may be found at www.sbc.com.	     N		

Issued: February 27, 2004

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

4.4

SECTION 4 - RATES AND CHARGES						
Outbound Services-Switched Access (continued) N						
4.4.3 Consum	4.4.3 Consumer Outbound Services					
(AE) Ju	ıstCall	sm				
.1	0 Ju	stCall <sup>sm</sup> Preferred Block of Time Options				
	.a	JustCall <sup>sm</sup> 60 Preferred				
		The MRC is \$3.00 per BTN for a 60 minute block of time as defined in Section 3.4.3 (AE).10.a of this Tariff. The rate is \$0.07 per minute for all outbound intrastate calls completed after the 60 minute block of time has been used.				
	.b	JustCall <sup>sm</sup> 200 Preferred				
		The MRC is \$8.00 per BTN for a 200 minute block of time as defined in Section 3.4.3 (AE).10.b of this Tariff. The rate is \$0.07 per minute for all outbound intrastate calls completed after the 200 minute block of time has been used.				
	.c	JustCall <sup>sm</sup> 400 Preferred				
		The MRC is \$12.00 per BTN for a 400 minute block of time as defined in Section 3.4.3 (AE).10.c of this Tariff. The rate is \$0.07 per minute for all outbound intrastate calls completed after the 400 minute block of time has been used.	     N			

Issued: February 27, 2004

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

4.4

### **SECTION 4 - RATES AND CHARGES** 4.4 Outbound Services-Switched Access (continued) Ν **Consumer Outbound Services** 4.4.3(AE) JustCall<sup>sm</sup> .11 JustCall<sup>sm</sup> Options JustCall<sup>sm</sup> 5 Cents .a The usage rate is \$0.05 per minute. For the interstate MRC, see Section 4.4.3 (AE) of the Company's interstate Voice Product Reference and Pricing Guide, which may be found at www.sbc.com. JustCall<sup>sm</sup> 7 Cents .b The usage rate is \$0.07 per minute. For the interstate MRC, see Section 4.4.3 (AE) of the Company's interstate Voice Product Reference and Pricing Guide, which may be found at www.sbc.com. JustCall<sup>sm</sup> Block of Time Options .12 .a JustCall<sup>sm</sup> 60 The MRC is \$3.00 per BTN for a 60 minute block of time as defined in Section 3.4.3 (AE).12.a of this Tariff. The rate is \$0.07 per minute for all outbound intrastate calls completed after the 60 minute block of time has been used. Ν

Issued: February 27, 2004

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.3 Consumer Outbound Services (continued)
    - (AE) JustCall<sup>sm</sup>
      - .12 JustCall<sup>sm</sup> Block of Time Options (continued)
        - .b JustCall<sup>sm</sup> 200

The MRC is \$8.00 per BTN for a 200 minute block of time as defined in Section 3.4.3 (AE).12.b of this Tariff. The rate is \$0.07 per minute for all outbound intrastate calls completed after the 200 minute block of time has been used.

.c JustCall<sup>sm</sup> 400

The MRC is \$12.00 per BTN for a 400 minute block of time as defined in Section 3.4.3 (AE).12.c of this Tariff. The rate is \$0.07 per minute for all outbound intrastate calls completed after the 400 minute block of time has been used.

Т

Issued: April 2, 2004

Effective: May 3, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc.	PSC Mo No. 1
d/b/a SBC Long Distance	

-- -

Original Sheet 218.31

			SEC	CTION 4 - RATES AND CHA	RGES	Micsouri Public	
4.4	Outbo	und Services	-Switc	hed Access (continued)	i	RECTIFEB 27 2004	N
	4.4.3	Consumer	Outbo	und Services (continued)			
		(AE) Just	Call™		Se	rvice Commission	
		.12	JustC	Call <sup>®</sup> Block of Time Options (a	continu	ed)	
			.b	JustCall <sup>™</sup> 200			ļ
				The MRC is \$8.00 per BTN defined in Section 3.4.3 (AI \$0.07 per minute for all out after the 200 minute block of	E).12.b bound i	of this Tariff. The rate is intrastate calls completed	
			.c	JustCall <sup>sm</sup> 400			
				The MRC is \$12.00 per BT as defined in Section 3.4.3 ( is \$0.07 per minute for all o after the 60 minute block of	(AE).12 outbour	2.c of this Tariff. The rate d intrastate calls completed	N

# CANCELLED

MAY 0 3 2004 By SHR SAIR, 31 Public Service Commission MISSOURI

Issued: February 27, 2004

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Missouri Public Service Commission

FILED APR 01 2004

4.4	Outbo	ound Services-Switched Access (continued)				
	4.4.3	Consumer Outbound Services (continued)				
		(AF) Simply Talk <sup>sm</sup>				
		The usage rate is \$0.21 per minute.				
		(AG) FallBack				
		The rates and charges for this optional calling plan are the same as shown in Section 4.4.3 (AF) of this Tariff.	   N			

Issued: March 10, 2004

Effective: April 12, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

REC'D MAR 07 2001

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance Original Sheet 219 Service Commission

### **SECTION 4 - RATES AND CHARGES**

- 4.4 Outbound Services-Switched Access (continued)
  - 4.4.4 Business Outbound Services

Reserved for future use.

Issued: March 7, 2001

Effective 73 pr123, 2001

Norm Descoteaux, Regulatory Manager DEC 0 7 2001 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED DEC 07 2001 01-475 Service Commission

## REC'D MAR 07 2001

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u>

Service Commission

### SECTION 4 - RATES AND CHARGES

4.5 Outbound Services-Dedicated Access

Reserved for future use.

Issued: March 7, 2001

Effective April 23, 2001

Norm Descoteaux, Regulatory ManagerDEC 07 20015850 W. Las Positas Blvd., Pleasanton, California 94588Missouri Public

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 FLED DEC 07 2001 Service 1 Commission

Southwestern Bell Communications S	ervices, Inc. F	PSC Mo No. 1
d/b/a SBC Long Distance		

### 4.6 Toll Free Services

.

Ş

N
1
]
1
}
N

Issued: October 24, 2003

Effective: November 24, 2003

Lisa Poterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

### Missouri Public

### REC'D MAR 07 2001

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

,ť

Service Continuission

#### **SECTION 4 - RATES AND CHARGES**

#### **Toll Free Services** 4.6

4.6.1 Reserved for future use.

NOV 2 4 2003 ssion PI

Issued: March 7, 2001

Effective. April 23, 2001 DEC 07 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Missouri Public

FILED DEC 07 2001 01-475 Service Commission Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u>

# REC'D MAR 07 2001

Service Commission

### SECTION 4 - RATES AND CHARGES

4.6 Toll Free Services (continued)

4.6.2 Reserved for future use.

Issued: March 7, 2001

Effective April 23, 2001 5

Norm Descoteaux, Regulatory Manager DEC 0 7 2001 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED DEC 072001 01-475 Service Commission

Ĺ

4.6 Toll Free Services (continued)

4.6.3 Optional Feature Charges

The description and rates for available optional features may be found in the Company's interstate Voice Product Reference and Pricing Guide which may be found at www.sbc.com.

D

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

2nd Revised Sheet 223 Replacing 1st Revised Sheet 223

### SECTION 4 - RATES AND CHARGES

#### Toll Free Services (continued) 4.6

4.6.3 **Optional Feature Charges** 

> The description and rates for available optional features may be found in the Company's interstate Voice Product Reference and Pricing Guide which may be found at www.sbc.com/public\_affairs/.

CANCELLED

FEB 07 2003 mssior

Issued: September 27, 2002

Effective: October 30, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED OCT 3 0 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Service Commission

Service Commission

Т

Missouri Public REC'D SEP 2 7 2002 Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

### 1st Revised Sheet 223 Replacing Original Sheet 223

### SECTION 4 - RATES AND CHARGES

4.6 Toll Free Services (continued)

4.6.3 Optional Feature Charges

The description and rates for available optional features may be found in the Company's interstate Voice Product Reference and Pricing Guide which may be T found at www.sbc.com/PublicAffairs/LDByState/. T

# CANCELLED

Issued: December 7, 2001

Effective: January 6, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED JAN 06 2002

Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

### HARGES Missouri Public

MOELLED



Service Commission

A dia a sunt fitulati

RECTD DEC 07 2001

### Missouri Public

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance RECD MAR 0 7 2001 Original Sheet 223 Service Commission

SECTION 4 - RATES AND CHARGES

4.6 Toll Free Services (continued)

4.6.3 Optional Feature Charges

The description and rates for available optional features may be found in the Company's interstate tariff.

JAN 0 6 2002

Issued: March 7, 2001

Effective: April 23, 2001 DEC 0 7 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED DEC 07 2001 01-475 Service Commission

Missouri Public

REC'D MAR 0 7 2001

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance Original Sheet 224 Service Commission

### SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services
  - 4.7.1 High Volume Calling
    - (A) Outbound Calls
      - .1 High Volume Outbound Calling

The per minute usage rates for InterLATA calls are as follows:

		Per Minute Rate			
MMC	MAC	MTM	1 Year Term Plan	2 Year Term Plan	3 Year Term Plan
\$50	\$600	\$0.1100	\$0.0990	\$0.0970	\$0.0940
\$200	\$2,400	\$0.1100	\$0.0980	\$0.0960	\$0.0930
\$500	\$6,000	\$0.1100	\$0.0970	\$0.0950	\$0.0920
\$1,000	\$12,000	\$0.1080	\$0.0960	\$0.0940	\$0.0910
\$2,500	\$30,000	\$0.1060	\$0.0950	\$0.0930	\$0.0900
\$5,000	\$60,000	\$0.1040	\$0.0940	\$0.0920	\$0.0890
\$10,000	\$120,000	\$0.1020	\$0.0930	\$0.0910	\$0.0880
\$15,000	\$180,000	\$0.1000	\$0.0920	\$0.0900	\$0.0870
\$20,000	\$240,000	\$0.0980	\$0.0910	\$0.0890	\$0.0860

Issued: March 7, 2001

Effectiver April 23, 2001

DEC 0 7 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

.

Missouri Public

FILED DEC 072001 01-475 Service Commission

4.7 Custom Business Services (continued)

i

- 4.7.1 High Volume Calling<sup>1</sup> (continued)
  - (A) Outbound Calls (continued)
    - .1 High Volume Outbound Calling (continued)

The per minute usage rates for IntraLATA calls are as follows:

MMC	MAC	Per Minute Rate			
		MTM	1 Year Term Plan	2 Year Term Plan	3 Year Term Plan
\$50	\$600	\$0.1100	\$0.0990	\$0.0970	\$0.0940
\$200	\$2,400	\$0.1100	\$0.0980	\$0.0960	\$0.0930
\$500	\$6,000	\$0.1100	\$0.0970	\$0.0950	\$0.0920
\$1,000	\$12,000	\$0.1080	\$0.0960	\$0.0940	\$0.0910
\$2,500	\$30,000	\$0.1060	\$0.0950	\$0.0930	\$0.0900
\$5,000	\$60,000	\$0.1040	\$0.0940	\$0.0920	\$0.0890
\$10,000	\$120,000	\$0.1020	\$0.0930	\$0.0910	\$0.0880
\$15,000	\$180,000	\$0.1000	\$0.0920	\$0.0900	\$0.0870
\$20,000	\$240,000	\$0.0980	\$0.0910	\$0.0890	\$0.0860

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective August 1, 2003.

Issued: June 25, 2003

Effective: August 1, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

### Missouri Public

RECD MAR 0 7 2001 Original Sheet 225 Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 ...

Service Commission

530B

### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

d/b/a SBC Long Distance

- 4.7.1 High Volume Calling (continued)
  - Outbound Calls (continued) (A)
    - .1 High Volume Outbound Calling (continued)

The per minute usage rates for IntraLATA calls are as follows:

	<u></u>	Per Minute Rate			
MMC	MAC	MTM	1 Year	2 Year	3 Year
			Term Plan	Term Plan	Term Plan
\$50	\$600	\$0.1100	\$0.0990	\$0.0970	\$0.0940
\$200	\$2,400	\$0.1100	\$0.0980	\$0.0960	\$0.0930
\$500	\$6,000	\$0.1100	\$0.0970	\$0.0950	\$0.0920
\$1,000	\$12,000	\$0.1080	\$0.0960	\$0.0940	\$0.0910
\$2,500	\$30,000	\$0.1060	\$0.0950	\$0.0930	\$0.0900
\$5,000	\$60,000	\$0.1040	\$0.0940	\$0.0920	\$0.0890
\$10,000	\$120,000	\$0.1020	\$0.0930	\$0.0910	\$0.0880
\$15,000	\$180,000	\$0.1000	\$0.0920	\$0.0900	\$0.0870
\$20,000	\$240,000	\$0.0980	\$0.0910	\$0.0890	\$0.0860

Issued: March 7, 2001

Effective: April 23, 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

DEC 0 7 2001 Missouri Public

07 2001 ñ1-475 Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

CANCELLED

BA

200

AUG\_0 1 2003
4.7 Custom Business Services (continued)

. . . . . . . . . . .

:

- 4.7.1 High Volume Calling<sup>1</sup> (continued)
  - (A) Outbound Calls (continued)
    - .2 High Volume Dedicated Outbound Calling

The per minute usage rates for InterLATA calls are as follows.

		Per Minute Rate					
MMC	MAC	MTM	1 Year Term Plan	2 Year Term Plan	3 Year Term Plan		
\$50	\$600	\$0.0820	\$0.0720	\$0.0680	\$0.0640		
\$200	\$2,400	\$0.0800	\$0.0700	\$0.0660	\$0.0620		
\$500	\$6,000	\$0.0780	\$0.0680	\$0.0640	\$0.0600		
\$1,000	\$12,000	\$0.0760	\$0.0660	\$0.0620	\$0.0580		
\$2,500	\$30,000	\$0.0740	\$0.0630	\$0.0590	\$0.0550		
\$5,000	\$60,000	\$0.0720	\$0.0600	\$0.0570	\$0.0530		
\$10,000	\$120,000	\$0.0700	\$0.0580	\$0.0560	\$0.0520		
\$15,000	\$180,000	\$0.0680	\$0.0570	\$0.0550	\$0.0510		
\$20,000	\$240,000	\$0.0660	\$0.0560	\$0.0540	\$0.0500		

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective August 1, 2003.

N N

Issued: June 25, 2003

N

Effective: August 1, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

 Southwestern Bell Communications Services, Inc.
 PSC Mo. - No. 1
 2nd Revised Sheet 226

 d/b/a SBC Long Distance
 Replacing 1st Revised Sheet 226

#### SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services (continued)
  - 4.7.1 High Volume Calling (continued)
    - (A) Outbound Calls (continued)
      - .2 High Volume Dedicated Outbound Calling

The per minute usage rates for InterLATA calls are as follows.

· · <b>_</b> ·			Per Minute Rate						
MMC	MAC	МТМ	1 Year Term Plan	2 Year Term Plan	3 Year Term Plan				
\$50	\$600	\$0.0820	\$0.0720	\$0.0680	\$0.0640				
\$200	\$2,400	\$0.0800	\$0.0700	\$0.0660	\$0.0620				
\$500	\$6,000	\$0.0780	\$0.0680	\$0.0640	\$0.0600				
\$1,000	\$12,000	\$0.0760	\$0.0660	\$0.0620	\$0.0580				
\$2,500	\$30,000	\$0.0740	\$0.0630	\$0.0590	\$0.0550				
\$5,000	\$60,000	\$0.0720	\$0.0600	\$0.0570	\$0.0530				
\$10,000	\$120,000	\$0.0700	\$0.0580	\$0.0560	\$0.0520				
	J								

\$0.0570

\$0.0560

\$0.0680

\$0.0660

R R

CANCELLED

UG 01 2003

\$0.0510

\$0.0500

\$0.0550

\$0.0540

Issued: October 16, 2002

Effective: November 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 ~

qud

## Missouri Public

FILED NOV 01 2002

Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

\$15,000

\$20,000

\$180,000

\$240,000

# Missouri Public

REC'D OCT 1 6 2002

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

1st Revised Sheet 226 **Replacing Original Sheet 226** 

#### **SECTION 4 - RATES AND CHARGES**

4.7 Custom Business Services (continued)

- High Volume Calling (continued) 4.7.1
  - Service Commission Outbound Calls (continued) (A)
    - .2 High Volume Dedicated Outbound Calling

The per minute usage rates for InterLATA calls are as follows.

	1		Per Minute Rate						
MMC	MAC	MTM	1 Year Term Plan	2 Year Term Plan	3 Year Term Plan				
\$50	\$600	\$0.0910	\$0.0880	\$0.0820	\$0.0760				
\$200	\$2,400	\$0.0910	\$0.0860	\$0.0800	\$0.0740				
\$500	\$6,000	\$0.0780	\$0.0680	\$0.0640	\$0.0600				
\$1,000	\$12,000	\$0.0760	\$0.0660	\$0.0620	\$0.0580				
\$2,500	\$30,000	\$0.0740	\$0.0630	\$0.0590	\$0.0550				
\$5,000	\$60,000	\$0.0720	\$0.0600	\$0.0570	\$0.0530				
\$10,000	\$120,000	\$0.0700	\$0.0580	\$0.0560	\$0.0520				
\$15,000	\$180,000	\$0.0680	\$0.0570	\$0.0550	\$0.0510				
\$20,000	\$240,000	\$0.0660	\$0.0560	\$0.0540	\$0.0500				



FILED MAR 04 2002

Service Commission

NOV 0 1 2002 Pupilo

CANCELLED

Issued: February 19, 2002

Effective: March 4, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Missouri Public

RECD FEB 1 9 2002

### Missourl Public

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

Original Sheet 226

RECTD MAR 0 7 2001

#### SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services (continued)
  - 4.7.1 High Volume Calling (continued)
    - (A) Outbound Calls (continued)
      - .2 High Volume Dedicated Outbound Calling

The per minute usage rates for InterLATA calls are as follows.

		Per Minute Rate						
MMC	MAC	MTM	1 Year Term Plan	2 Year Term Plan	3 Year Term Plan			
\$50	\$600	\$0.0910	\$0.0880	\$0.0820	\$0.0760			
\$200	\$2,400	\$0.0910	\$0.0860	\$0.0800	\$0.0740			
\$500	\$6,000	\$0.0880	\$0.0780	\$0.0740	\$0.0700			
\$1,000	\$12,000	\$0.0860	\$0.0760	\$0.0720	\$0.0680			
\$2,500	\$30,000	\$0.0840	\$0.0730	\$0.0690	\$0.0650			
\$5,000	\$60,000	\$0.0820	\$0.0700	\$0.0670	\$0.0630			
\$10,000	\$120,000	\$0.0800	\$0.0680	\$0.0660	\$0.0620			
\$15,000	\$180,000	\$0.0780	\$0.0670	\$0.0650	\$0.0610			
\$20,000	\$240,000	\$0.0760	\$0.0660	\$0.0640	\$0.0600			

CANCELLED

MAR 0 4 2002 MAR 0 4 2002 MAR 0 4 2002 Public Service Commission MISSOURI

Issued: March 7, 2001

Effective April 23, 2001 CH

DEC 0 7 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

.

Missouri Public

FILED DEC 072001 01-475 Service Commission

4.7 Custom Business Services (continued)

----

:

÷r

- 4.7.1 High Volume Calling<sup>1</sup> (continued)
  - (A) Outbound Calls (continued)
    - .2 High Volume Dedicated Outbound Calling (continued)

The per minute usage rates for IntraLATA calls are as follows.

		Per Minute Rate					
MMC	MAC	MTM	l Year Term Plan	2 Year Term Plan	3 Year Term Plan		
\$50	\$600	\$0.0820	\$0.0720	\$0.0680	\$0.0640		
\$200	\$2,400	\$0.0800	\$0.0700	\$0.0660	\$0.0620		
\$500	\$6,000	\$0.0780	\$0.0680	\$0.0640	\$0.0600		
\$1,000	\$12,000	\$0.0760	\$0.0660	\$0.0620	\$0.0580		
\$2,500	\$30,000	\$0.0740	\$0.0630	\$0.0590	\$0.0550		
\$5,000	\$60,000	\$0.0720	\$0.0600	\$0.0570	\$0.0530		
\$10,000	\$120,000	\$0.0700	\$0.0580	\$0.0560	\$0.0520		
\$15,000	\$180,000	\$0.0680	\$0.0570	\$0.0550	\$0.0510		
\$20,000	\$240,000	\$0.0660	\$0.0560	\$0.0540	\$0.0500		

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective August 1, 2003.

N N

C

Issued: June 25, 2003

Effective: August 1, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance 2nd Revised Sheet 227 Replacing 1st Revised Sheet 227

#### **SECTION 4 - RATES AND CHARGES**

- 4.7 Custom Business Services (continued)
  - 4.7.1 High Volume Calling (continued)
    - (A) Outbound Calls (continued)
      - .2 High Volume Dedicated Outbound Calling (continued)

Per Minute Rate 2 Year MMC MTM 1 Year 3 Year MAC Term Plan Term Plan Term Plan \$0.0820 \$0.0680 \$50 \$600 \$0.0720 \$0.0640 \$200 \$2,400 \$0.0800 \$0.0700 \$0.0660 \$0.0620 \$500 \$6,000 \$0.0780 \$0.0680 \$0.0640 \$0.0600 \$1,000 \$12,000 \$0.0760 \$0.0660 \$0.0620 \$0.0580 \$2,500 \$30,000 \$0.0740 \$0.0630 \$0.0590 \$0.0550 \$5,000 \$60,000 \$0.0720 \$0.0600 \$0.0570 \$0.0530 \$10,000 \$120,000 \$0.0700 \$0.0580 \$0.0560 \$0.0520 \$15,000 \$180,000 \$0.0680 \$0.0570 \$0.0550 \$0.0510 \$20,000 \$240,000 \$0.0660 \$0.0560 \$0.0540 \$0.0500

The per minute usage rates for IntraLATA calls are as follows.

CANCELLED

Issued: October 16, 2002

Effective: November 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

FILED NOV 01 2002

Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

# Missouri Public

REC'D OCT 1 6 2002

Service Commission

R

R

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

#### **SECTION 4 - RATES AND CHARGES**

4.7 Custom Business Services (continued)

4.7.1

Missouri Public

(A) Outbound Calls (continued)

High Volume Calling (continued)

REC'D FEB 1 9 2002

.2 High Volume Dedicated Outbound Calling (continued)

The per minute usage rates for IntraLATA calls are as follows.

ММС	MAC	MTM	1 Year Term Plan	2 Year Term Plan	3 Year Term Plan	
\$50	\$600	\$0.0910	\$0.0880	\$0.0820	\$0.0760	
\$200	\$2,400	\$0.0910	\$0.0860	\$0.0800	\$0.0740	
\$500	\$6,000	\$0.0780	\$0.0680	\$0.0640	\$0.0600	R
\$1,000	\$12,000	\$0.0760	\$0.0660	\$0.0620	\$0.0580	
\$2,500	\$30,000	\$0.0740	\$0.0630	\$0.0590	\$0.0550	
\$5,000	\$60,000	\$0.0720	\$0.0600	\$0.0570	\$0.0530	
\$10,000	\$120,000	\$0.0700	\$0.0580	\$0.0560	\$0.0520	
\$15,000	\$180,000	\$0.0680	\$0.0570	\$0.0550	\$0.0510	i l
\$20,000	\$240,000	\$0.0660	\$0.0560	\$0.0540	\$0.0500	R

Missouri Public

FILED MAR 04 2002

Service Commission

CANCELLED 0 1 200%

INS STOR PU

Issued: February 19, 2002

Effective: March 4, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

## Missouri Public

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance RECD MAR 07 2001 Original Sheet 227 Service Commission

#### SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services (continued)
  - 4.7.1 High Volume Calling (continued)
    - (A) Outbound Calls (continued)
      - .2 High Volume Dedicated Outbound Calling (continued)

The per minute usage rates for IntraLATA calls are as follows.

		Per Minute Rate						
MMC	MAC	MTM	1 Year Term Plan	2 Year Term Plan	3 Year Term Plan			
\$50	\$600	\$0.0910	\$0.0880	\$0.0820	\$0.0760			
\$200	\$2,400	\$0.0910	\$0.0860	\$0.0800	\$0.0740			
\$500	\$6,000	\$0.0880	\$0.0780	\$0.0740	\$0.0700			
\$1,000	\$12,000	\$0.0860	\$0.0760	\$0.0720	\$0.0680			
\$2,500	\$30,000	\$0.0840	\$0.0730	\$0.0690	\$0.0650			
\$5,000	\$60,000	\$0.0820	\$0.0700	\$0.0670	\$0.0630			
\$10,000	\$120,000	\$0.0800	\$0.0680	\$0.0660	\$0.0620			
\$15,000	\$180,000	\$0.0780	\$0.0670	\$0.0650	\$0.0610			
\$20,000	\$240,000	\$0.0760	\$0.0660	\$0.0640	\$0.0600			

# CANCELLED

MAR 04 2002

Public Service Commission MISSOURI

Issued: March 7, 2001

Effective: April 23, 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

DEC 072001

FILED DEC 072001 01-475 Service Commission

4.7 Custom Business Services (continued)

;

- 4.7.1 High Volume Calling<sup>1</sup> (continued)
  - (B) Inbound Toll Free Calls
    - .1 High Volume Toll Free Calling Usage Rates
      - .a Without CMR

The per minute usage rates are the same as Section 4.7.1 (A).1 of this Tariff.

.b With CMR

The rate is \$.03 per minute which applies in addition to the per minute usage rates in Section 4.7.1 (A).1 of this Tariff.

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective August 1, 2003.

Issued: June 25, 2003

Effective: August 1, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 N N

	western 1 SBC Lon			nications	Services, Inc. PSC Mo No. 1 Replacing 2nd Revised Sheet 228 3rd Revised Sheet 228	
				SEC	TION 4 - RATES AND CHARGES Missouri Public	
4.7	Custor	m Bus	iness S	Services	RECD AUG 01 2002	)
	4.7.1	Higł	n Volu	ime Cal	ling (continued) Service Commission	on
		(B)	Inbo	ound To	ll Free Calls	
			1	High	Volume Toll Free Calling - Usage Rates	
				.a	Without CMR	
					The per minute usage rates are the same as Section 4.7.1 (A).1 of this Tariff.	
				.b	With CMR	
					The rate is \$.03 per minute which applies in addition to the per minute usage rates in Section 4.7.1 (A).I of this Tariff.	
						M   

CANCELLED

0 1 2003

**MOISSING** 

M - Material moved to Original Sheet 228.1

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

21

Missouri Public

Μ

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

		western I SBC_Lon			cations	Services, Inc.	PSC Mo No. 1	2nd Revised Sheet 228 Replacing 1st Revised Sheet 228	
	47	Gusta		0			TES AND CHAR	GES Missouri Public	
	4.7	Custor	n Busii	iess 5	ervices	(continued)			
		4.7.1	High	Volur	ne Calli	ing (continue	d)	REC'D MAY 3 0 2002	
			(B)	Inbou	und Tol	l Free Calls		Service Commissior	ì
				.1	High V	Volume Toll I	Free Calling - Us	age Rates	
		IFN			.a	Without CM	ſR		Ν
CÞ	NO.FO	1 FN 1 2002 822	)			The per min of this Tarif		re the same as Section 4.7.1 (A).1	
c c		SP	y missio	A	.b	With CMR			Ν
Pupilc Fy	S MIS	JUA					•	which applies in addition to the per a 4.7.1 (A).1 of this Tariff.	   N
_				.2	High V	Volume Dedi	cated Toll Free C	alling - Usage Rates	
					.a	Without CM	ſR		Ν
						The per min of this Tarif	-	re the same as Section 4.7.1 (A).2	
					.b	With CMR			N
								which applies in addition to the per 1 4.7.1 (A).2 of this Tariff.	N
				.3	Option	nal Feature C	harges		
					in the	Company's V	Voice Product Re	ble optional features may be found eference and Pricing Guide which licAffairs/LDByState/.	

Issued: May 30, 2002

4

Effective: July 1, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED JUL 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

.

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance **Replacing Original Sheet 228** Missouri Public **SECTION 4 - RATES AND CHARGES** RECT DEC 07 2001 4.7 Custom Business Services (continued) Service Commission High Volume Calling (continued) 4.7.1 **(B)** Inbound Toll Free Calls High Volume Toll Free Calling - Usage Rates .1 The per minute usage rates are the same as Section 4.7.1 (A).1 of this Tariff. .2 High Volume Dedicated Toll Free Calling - Usage Rates The per minute usage rates are the same as Section 4.7.1 (A).2 of this Tariff. .3 **Optional Feature Charges** The description and rates for available optional features may be found in the Company's Voice Product Reference and Pricing Guide which Т may be found at www.sbc.com/PublicAffairs/LDByState/. Т

# CANCELLED

JUI 0 1 2002 228 Public Service Commission MISSOURI

Issued: December 7, 2001

ų,

Effective: January 6, 2002

1st Revised Sheet 228

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

# Missouri Public

FILED JAN 06 2002

CANCELLED December 24, 2004 **Missouri Public** Service Commission TN-2005-0149

## Missouri Public

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance RECT) MAR 0 7 2001 Original Sheet 228 Service Commission

#### SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services (continued)
  - 4.7.1 High Volume Calling (continued)
    - (B) Inbound Toll Free Calls
      - .1 High Volume Toll Free Calling Usage Rates

The per minute usage rates are the same as Section 4.7.1 (A).1 of this Tariff.

.2 High Volume Dedicated Toll Free Calling - Usage Rates

The per minute usage rates are the same as Section 4.7.1 (A).2 of this Tariff.

.3 Optional Feature Charges

The description and rates for available optional features may be found in the Company's interstate tariff.



Issued: March 7, 2001

Effective April 23,2901-

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

DEC 0 7 2001 Missouri Public

FILED DEC 07 2001 01-475 Service Commission

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 3rd Revised Sheet 228.1 d/b/a SBC Long Distance Replacing 2nd Revised Sheet 228.1

#### **SECTION 4 - RATES AND CHARGES**

4.7 Custom Business Services (continued)

•

- High Volume Calling<sup>1</sup> (continued) 4.7.1
  - Inbound Toll Free Calls (continued) **(B)** 
    - .2 High Volume Dedicated Toll Free Calling - Usage Rates
      - Without CMR .a

The per minute usage rates are the same as Section 4.7.1 (A).2 of this Tariff.

.b With CMR

> The rate is \$.03 per minute which applies in addition to the per minute usage rates in Section 4.7.1 (A).2 of this Tariff.

.3 **Optional Feature Charges** 

> The description and rates for available optional features may be found in the Company's Voice Product Reference and Pricing Guide which may be found at www.sbc.com.

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations Ν effective August 1, 2003.

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Issued: June 25, 2003

С

Ν

Effective: August 1, 2003

PSC Mo. - No. 1 2nd Revised Sheet 228.1 Southwestern Bell Communications Services, Inc. d/b/a SBC Long Distance Replacing 1st Revised Sheet 228.1

#### SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services (continued)
  - 4.7.1 High Volume Calling (continued)
    - Inbound Toll Free Calls (continued) (B)
      - .2 High Volume Dedicated Toll Free Calling - Usage Rates
        - Without CMR. .a

The per minute usage rates are the same as Section 4.7.1 (A).2 of this Tariff.

With CMR .b

> The rate is \$.03 per minute which applies in addition to the per minute usage rates in Section 4.7.1 (A).2 of this Tariff.

.3 **Optional** Feature Charges

> The description and rates for available optional features may be found in the Company's Voice Product Reference and Pricing Guide which may be found at www.sbc.com.

# CANCELLED

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 **Missouri Public** Service Commission TN-2005-0149

Issued: January 7, 2003

Effective: February 7, 2003

Missouri Public Sorvice Commission

# FILED FEB. 07 2003

REGA JAN 0 7 2003



D

Missourt Public Service Commission

	western H SBC Lo <u>n</u>			ications	Services, Inc. PSC Mo No. 1 1st Revised Sheet 228.1 Replacing Original Sheet 328.1			
				SEC	TION 4 - RATES AND CHARGES Service Commissio	5		
4.7	Custor	n Busi	ness S	ervices	(continued)	**		
	4.7.1	High	Volu	me Call	ing (continued)			
	(B) Inbound Toll Free Calls (continued) N							
			.2	High `	Volume Dedicated Toll Free Calling - Usage Rates			
				.a	Without CMR			
					The per minute usage rates are the same as Section 4.7.1 (A).2 of this Tariff.			
				.b	With CMR			
					The rate is \$.03 per minute which applies in addition to the per minute usage rates in Section 4.7.1 (A).2 of this Tariff.			
			.3	Optio	nal Feature Charges			
				in the	escription and rates for available optional features may be found Company's Voice Product Reference and Pricing Guide which be found at www.sbc.com/public_affairs/.	Т		
					CANCELLED			

0 7 2003

Publ

Issued: September 27, 2002

Ĵ

Effective: October 30, 2002

Missouri Public

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

**Missouri** Public

÷

FILED OCT 3 0 2002

		Bell Commun	ications	Services, Inc.	PSC Mo No. 1	Original Sheet 228.1	
		<u> </u>	SEC.	TION 4 - RAT	TES AND CHARGES	Missouri Public	;
4.7	Custor	n Business S	Services	(continued)		REC'D AUG 01 200	2м
	4.7.1	High Volu	me Call	ling (continued	d)	Service Commiss	ί <mark>φ</mark> η
		.2	High	Volume Dedic	cated Toll Free Calling -	Usage Rates	
			.a	Without CM	IR		{
				The per min of this Tarifi	e	ame as Section 4.7.1 (A).2	
			.b	With CMR			
					0.03 per minute which age rates in Section 4.7.1	oplies in addition to the per (A).2 of this Tariff.	
		.3	Optio	nal Feature Cl	harges		
			in the	e Company's V	-	onal features may be found e and Pricing Guide which rs/LDByState/.	   M

# CANCELLED

OCT 3 0 2002 STRS 228.1 Public Scrift Continues on MISSOURI

M - Material moved from 2nd Revised Sheet 228

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

PSC Mo. - No. 1

- 4.7 Custom Business Services (continued)
  - 4.7.2 High Volume Calling II
    - (A) Outbound Calls
      - .1 High Volume Outbound Calling II

The per minute usage rates for intrastate InterLATA calls are as follows:

		Per Minute Rate						
MMC	MAC	MTM	1 Year	2 Year	3 Year			
			Term Plan	Term Plan	Term Plan			
\$50	\$600	\$0.1100	\$0.0990	\$0.0970	\$0.0940			
\$200	\$2,400	\$0.1100	\$0.0980	\$0.0960	\$0.0930			
\$500	\$6,000	\$0.1100	\$0.0970	\$0.0950	\$0.0920			
\$1,000	\$12,000	\$0.1080	\$0.0960	\$0.0940	\$0.0910			
\$2,500	\$30,000	\$0.1060	\$0.0950	\$0.0930	\$0.0900			
\$5,000	\$60,000	\$0.1040	\$0.0940	\$0.0920	\$0.0890			
\$10,000	\$120,000	\$0.1020	\$0.0930	\$0.0910	\$0.0880			
\$15,000	\$180,000	\$0.1000	\$0.0920	\$0.0900	\$0.0870			
\$20,000	\$240,000	\$0.0980	\$0.0910	\$0.0890	\$0.0860			

Ν

Issued: November 26, 2002

Effective: December 27, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 N

Missouri Public

RECT MAR 07 2001

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance SerVice Commission

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.2 Reserved for future use.

CANCELLED

DEC 27 2002 St RS 229 Aubito Service Commission Missouri

Issued: March 7, 2001

Effective April 25 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588 DEC 0 7 2001 Missouri Public

FILED DEC 07 2001 01-475 Service Commission

- 4.7 Custom Business Services (continued)
  - 4.7.2 High Volume Calling II (continued)
    - (A) Outbound Calls (continued)
      - .1 High Volume Outbound Calling II (continued)

The per minute usage rates for intrastate IntraLATA calls are as follows:

			Per Minute Rate				
MMC	MAC	MTM	1 Year	2 Year	3 Year		
			Term Plan	Term Plan	Term Plan		
\$50	\$600	\$0.1100	\$0.0990	\$0.0970	\$0.0940		
\$200	\$2,400	\$0.1100	\$0.0980	\$0.0960	\$0.0930		
\$500	\$6,000	\$0.1100	\$0.0970	\$0.0950	\$0.0920		
\$1,000	\$12,000	\$0.1080	\$0.0960	\$0.0940	\$0.0910		
\$2,500	\$30,000	\$0.1060	\$0.0950	\$0.0930	\$0.0900		
\$5,000	\$60,000	\$0.1040	\$0.0940	\$0.0920	\$0.0890		
\$10,000	\$120,000	\$0.1020	\$0.0930	\$0.0910	\$0.0880		
\$15,000	\$180,000	\$0.1000	\$0.0920	\$0.0900	\$0.0870		
\$20,000	\$240,000	\$0.0980	\$0.0910	\$0.0890	\$0.0860		

Ν

Issued: November 26, 2002

Effective: December 27, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

- 4.7 Custom Business Services (continued)
  - 4.7.2 High Volume Calling II (continued)
    - (A) Outbound Calls (continued)
      - .2 High Volume Dedicated Outbound Calling II

The per minute usage rates for InterLATA calls are as follows.

		Per Minute Rate				
MMC	MAC	MTM	1 Year	2 Year	3 Year	
			Term Plan	Term Plan	Term	
					Plan	
\$50	\$600	\$0.0820	\$0.0720	\$0.0680	\$0.0640	
\$200	\$2,400	\$0.0800	\$0.0700	\$0.0660	\$0.0620	
\$500	\$6,000	\$0.0780	\$0.0680	\$0.0640	\$0.0600	
\$1,000	\$12,000	\$0.0760	\$0.0660	\$0.0620	\$0.0580	
\$2,500	\$30,000	\$0.0740	\$0.0630	\$0.0590	\$0.0550	
\$5,000	\$60,000	\$0.0720	\$0.0600	\$0.0570	\$0.0530	
\$10,000	\$120,000	\$0.0700	\$0.0580	\$0.0560	\$0.0520	
\$15,000	\$180,000	\$0.0680	\$0.0570	\$0.0550	\$0.0510	
\$20,000	\$240,000	\$0.0660	\$0.0560	\$0.0540	\$0.0500	

N

Issued: November 26, 2002

Effective: December 27, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

4.7 Custom Business Services (continued)

•

- 4.7.2 High Volume Calling II (continued)
  - (A) Outbound Calls (continued)
    - .2 High Volume Dedicated Outbound Calling II (continued)

The per minute usage rates for IntraLATA calls are as follows.

MMC	MAC	MTM	1 Year	2 Year	3 Year
			Term Plan	Term Plan	Term Plan
\$50	\$600	\$0.0820	\$0.0720	\$0.0680	\$0.0640
\$200	\$2,400	\$0.0800	\$0.0700	\$0.0660	\$0.0620
\$500	\$6,000	\$0.0780	\$0.0680	\$0.0640	\$0.0600
\$1,000	\$12,000	\$0.0760	\$0.0660	\$0.0620	\$0.0580
\$2,500	\$30,000	\$0.0740	\$0.0630	\$0.0590	\$0.0550
\$5,000	\$60,000	\$0.0720	\$0.0600	\$0.0570	\$0.0530
\$10,000	\$120,000	\$0.0700	\$0.0580	\$0.0560	\$0.0520
\$15,000	\$180,000	\$0.0680	\$0.0570	\$0.0550	\$0.0510
\$20,000	\$240,000	\$0.0660	\$0.0560	\$0.0540	\$0.0500

Ņ

Issued: November 26, 2002

Effective: December 27, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

# **SECTION 4 - RATES AND CHARGES** Custom Business Services (continued) Ν 1 4.7.2 High Volume Calling II (continued) **(B)** Inbound Toll Free Calls .1 High Volume Toll Free Calling II - Usage Rates Without CMR .a The per minute usage rates are the same as Section 4.7.2 (A).1 of this Tariff. .b With CMR The rate is \$.03 per minute which applies in addition to the per minute usage rates in Section 4.7.2 (A).1 of this Tariff.

Ν

Issued: November 26, 2002

Effective: December 27, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

4.7

Custor	Custom Business Services (continued)						
4.7.2	High Volume Calling II (continued)						
	(B)	Inbo	ound To	ll Free Calls (continued)	t 1 1		
		.2	High	Volume Dedicated Toll Free Calling II - Usage Rates	1 1		
			.a	Without CMR			
				The per minute usage rates are the same as Section 4.7.2 (A).2 of this Tariff.			
			.b	With CMR			
				The rate is \$.03 per minute which applies in addition to the per minute usage rates in Section 4.7.2 (A).2 of this Tariff.			
		.3	Optional Feature Charges The description and rates for available optional features may b found in the Company's interstate Voice Product Reference an				

Pricing Guide which may be found at www.sbc.com/public\_affairs/.

Issued: November 26, 2002

Effective: December 27, 2002

Ν

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

4.7

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

1

1

1st Revised Sheet 229.6 Replacing Original Sheet 229.6

> D I

> D

#### SECTION 4 - RATES AND CHARGES

Issued: June 10, 2003

Effective: July 10, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

#### Original Sheet 229.6

N

Ν

#### **SECTION 4 - RATES AND CHARGES**

- 4.7 Custom Business Services (continued)
  - High Volume Calling II (continued) 4.7.2
    - (C)

# Missouri Public Servico Commission

Sign On Bonus

**RECT DEC 18** 2002

Customers that subscribe to High Volume Outbound Calling II and/or High Volume Dedicated Outbound Calling II under a term plan agreement for the provision of their IntraLATA or their IntraLATA and InterLATA long distance intrastate calling and sign up for or already subscribe to CompleteLink service from an Affiliated LEC or Affiliated CLEC will receive a one-time bonus to be fulfilled to the Customer in the form of a check within ninety (90) days of the provisioning of their High Volume Calling II. The amount of the bonus is shown in the following chart:

MAC	<u>Term</u>	<u>Bonus Amount</u>
\$600	1 Year	\$6.00
\$2,400	1 Year	\$24.00
\$6,000	1 Year	\$60.00
\$12,000	1 Year	\$120.00
\$30,000 -		
\$240,000	1 Year	\$300.00

#### Missouri Public

FLED JAN 19 2003

#### <del>Service Commissio</del>n

Effective: January 19, 2003

Issued: December 18, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

#### CANCELLED

**Missouri Public** Service Commission R″ |^ F€ÊG€€H

SBC Long Distance, Inc.

PSC Mo. - No. 1

2nd Revised Sheet 230 Replacing 1st Revised Sheet 230

#### **SECTION 4 - RATES AND CHARGES**

#### 4.7 Custom Business Services (continued)

- 4.7.3 High Volume Calling Connections I<sup>1</sup>
  - (A) Outbound Calls
    - .1 High Volume Outbound Calling Connections I

The per minute usage rates for intrastate InterLATA calls are as follows:

МАС	l Year Term Plan
\$600	\$0.0970
\$2,400	\$0.0960
\$6,000	\$0.0950
\$12,000	\$0.0940
\$30,000	\$0.0930
\$60,000	\$0.0930
\$120,000	\$0.0930
\$180,000	\$0.0920
\$240,000	\$0.0910

This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2005.

N N

С

Issued: February 28, 2005

Effective: April 1, 2005

Janet Vader, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

1

FILED MO PSC .

#### SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services (continued)
  - 4.7.3 High Volume Calling Connections I
    - (A) Outbound Calls
      - .1 High Volume Outbound Calling Connections I

The per minute usage rates for intrastate InterLATA calls are as follows:

	1 Year
MAC	Term Plan
\$600	\$0.0970
\$2,400	\$0.0960
\$6,000	\$0.0950
\$12,000	\$0.0940
\$30,000	\$0.0930
\$60,000	\$0.0930
\$120,000	\$0.0930
\$180,000	\$0.0920
\$240,000	\$0.0910

N

Issued: February 28, 2003

Effective: March 31, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 T N

### Missouri Public

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance RECD MAR 0 7 2001 Original Sheet 230 Service Commission

#### **SECTION 4 - RATES AND CHARGES**

4.7 Custom Business Services (continued)

4.7.3 Reserved for future use.

CANCELLED SSION

Issued: March 7, 2001

Effective April 23 2001

Norm Descoteaux, Regulatory Manager DEU 5850 W. Las Positas Blvd., Pleasanton, California 94588

DEC 072001

Missouri Public

FILED DEC 072001 01-475 Service Commission

- 4.7 Custom Business Services (continued)
  - 4.7.3 High Volume Calling Connections I (continued)
    - Outbound Calls (continued) (A)

I

High Volume Outbound Calling Connections I (continued) .1

1 Year

The per minute usage rates for intrastate IntraLATA calls are as follows:

MAC	Term Plan
\$600	\$0.0970
\$2,400	\$0.0960
\$6,000	\$0.0950
\$12,000	\$0.0940
\$30,000	\$0.0930
\$60,000	\$0.0930
\$120,000	\$0.0930
\$180,000	\$0.0920
\$240,000	\$0.0910

Issued: February 28, 2003

Effective: March 31, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Ν

Ν

#### Original Sheet 230.2

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

-- --

- 4.7.3 High Volume Calling Connections I (continued)
  - (A) Outbound Calls (continued)
    - .2 High Volume Dedicated Outbound Calling Connections I

The per minute usage rates for InterLATA calls are as follows.

MAC	1 Year
	Term Plan
\$600	\$0.0720
\$2,400	\$0.0700
\$6,000	\$0.0680
\$12,000	\$0.0660
\$30,000	\$0.0630
\$60,000	\$0.0600
\$120,000	\$0.0580
\$180,000	\$0.0570
\$240,000	\$0.0560

Ν

Issued: February 28, 2003

Effective: March 31, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Ν

#### Original Sheet 230.3

#### SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services (continued)
  - 4.7.3 High Volume Calling Connections I (continued)
    - (A) Outbound Calls (continued)
      - .2 High Volume Dedicated Outbound Calling Connections I (continued)

The per minute usage rates for IntraLATA calls are as follows.

MAC	1 Year Term Plan
\$600	\$0.0720
\$2,400	\$0.0700
\$6,000	\$0.0680
\$12,000	\$0.0660
\$30,000	\$0.0630
\$60,000	\$0.0600
\$120,000	\$0.0580
\$180,000	\$0.0570
\$240,000	\$0.0560

Ň

Ň

Issued: February 28, 2003

Effective: March 31, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

---

٠

				SEC	TION 4 - RATES AND CHARGES			
4.7	Custor	Custom Business Services (continued)						
	4.7.3	High Volume Calling Connections I (continued)		ling Connections I (continued)				
		(B)	Inbo	ound To	oll Free Calls	i		
			.1	High	Volume Toll Free Calling Connections I - Usage Rates	1 1 1 1		
				.a	Without CMR	) } 		
					The per minute usage rates are the same as Section 4.7.3 (A).1 of this Tariff.	, , , , , ,		
				.b	With CMR			
					The rate is \$.03 per minute which applies in addition to the per minute usage rates in Section 4.7.3 (A).1 of this Tariff.	i   N		

Issued: February 28, 2003

Effective: March 31, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Custor	n Busi	ness	Services	(continued)	N			
4.7.3	High	ligh Volume Calling Connections I (continued)						
	(B)	Inbo	ound To	und Toll Free Calls (continued)				
		.2	High Rates	Volume Dedicated Toll Free Calling Connections I - Usage				
	.a Without CMR							
		The per minute usage rates are the same as Section 4 (A).2 of this Tariff.						
	.b With CMR							
				The rate is \$.03 per minute which applies in addition to the per minute usage rates in Section 4.7.3 (A).2 of this Tariff.				
		.3	Option	nal Feature Charges				
				lescription and rates for available optional features may be in the Company's interstate Voice Product Reference and				

Pricing Guide which may be found at www.sbc.com.

Issued: February 28, 2003

Effective: March 31, 2003

Ν

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

4.7

SBC Long Distance, Inc.

PSC Mo. - No. 1

2nd Revised Sheet 231 Replacing 1st Revised Sheet 231

#### **SECTION 4 - RATES AND CHARGES**

- 4.7 Custom Business Services (continued)
  - 4.7.4 High Volume Calling Connections II<sup>1</sup>
    - (A) Outbound Calls
      - .1 High Volume Outbound Calling Connections II

The per minute usage rates for intrastate InterLATA calls are as follows:

MAC	1 Year Term Plan
\$600	\$0.0950
\$2,400	\$0.0940
\$6,000	\$0.0930
\$12,000	\$0.0920
\$30,000	\$0.0910
\$60,000	\$0.0910
\$120,000	\$0.0910
\$180,000	\$0.0910
\$240,000	\$0.0910

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2005.

N N

С

Issued: February 28, 2005

Effective: April 1, 2005

Janet Vader, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 FILED MO PSC Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

.....

.....

#### SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services (continued)
  - 4.7.4 High Volume Calling Connections II
    - (A) Outbound Calls
      - .1 High Volume Outbound Calling Connections II

The per minute usage rates for intrastate InterLATA calls are as follows:

MAC	1 Year Term Plan
\$600	\$0.0950
\$2,400	\$0.0940
\$6,000	\$0.0930
\$12,000	\$0.0920
\$30,000	\$0.0910
\$60,000	\$0.0910
\$120,000	\$0.0910
\$180,000	\$0.0910
\$240,000	\$0.0910

Issued: February 28, 2003

Effective: March 31, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Ν
### Missouri Public

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance RECD MAR 0 7 2001 Original Sheet 231 Service Commission

### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.4 Reserved for future use.

CANCELLED mssion

Issued: March 7, 2001

Effective: April 25-2001.5

Norm Descoteaux, Regulatory Manager DEC 0 7 2001 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED DEC 07 2001 01-475 Service Commission

4.7 Custom Business Services (continued)

---

- 4.7.4 High Volume Calling Connections II (continued)
  - (A) Outbound Calls (continued)
    - .1 High Volume Outbound Calling Connections II (continued)

The per minute usage rates for intrastate IntraLATA calls are as follows:

	1 Year
MAC	Term Plan
\$600	\$0.0950
\$2,400	\$0.0940
\$6,000	\$0.0930
\$12,000	\$0.0920
\$30,000	\$0.0910
\$60,000	\$0.0910
\$120,000	\$0.0910
\$180,000	\$0.0910
\$240,000	\$0.0910

Issued: February 28, 2003

Effective: March 31, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Ν

- 4.7 Custom Business Services (continued)
  - 4.7.4 High Volume Calling Connections II (continued)
    - (A) Outbound Calls (continued)
      - .2 High Volume Dedicated Outbound Calling Connections II

The per minute usage rates for InterLATA calls are as follows.

МАС	1 Year Term Plan
\$600	\$0.0720
\$2,400	\$0.0700
\$6,000	\$0.0680
\$12,000	\$0.0660
\$30,000	\$0.0630
\$60,000	\$0.0600
\$120,000	\$0.0580
\$180,000	\$0.0570
\$240,000	\$0.0560

Ν

Ν

Issued: February 28, 2003

Effective: March 31, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

- 4.7 Custom Business Services (continued)
  - 4.7.4 High Volume Calling Connections II (continued)
    - (A) Outbound Calls (continued)

١Ċ

.2 High Volume Dedicated Outbound Calling Connections II (continued)

The per minute usage rates for IntraLATA calls are as follows.

MAC	1 Year
	Term Plan
\$600	\$0.0720
\$2,400	\$0.0700
\$6,000	\$0.0680
\$12,000	\$0.0660
\$30,000	\$0.0630
\$60,000	\$0.0600
\$120,000	\$0.0580
\$180,000	\$0.0570
\$240,000	\$0.0560

Issued: February 28, 2003

Effective: March 31, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 N

.....

•

- -

				SEC	TION 4 - RATES AND CHARGES	
4.7	Custor	n Bus	iness	Service	es (continued)	N
	4.7.4	High	n Voli	ıme Ca	lling Connections II (continued)	
		(B)	Inbo	ound Te	oll Free Calls	
			.1	High	Volume Toll Free Calling Connections II - Usage Rates	
				.a	Without CMR	
					The per minute usage rates are the same as Section 4.7.4 (A).1 of this Tariff.	1 1 1 1
				.b	With CMR	     
					The rate is \$.03 per minute which applies in addition to the per minute usage rates in Section 4.7.4 (A).1 of this Tariff.	i N

Issued: February 28, 2003

Effective: March 31, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Ν

i

Ν

#### SECTION 4 - RATES AND CHARGES

Custom Business Services (continued)					
4.7.4	High	Volu	olume Calling Connections II (continued)		
	(B)	Inbo	und To	ll Free Calls (continued)	
		.2	High Rates	Volume Dedicated Toll Free Calling Connections II - Usage	
			.a	Without CMR	
				The per minute usage rates are the same as Section 4.7.4 (A).2 of this Tariff.	
			.b	With CMR	
				The rate is \$.03 per minute which applies in addition to the per minute usage rates in Section 4.7.4 (A).2 of this Tariff.	
		.3	Optio	nal Feature Charges	
				description and rates for available optional features may be in the Company's interstate Voice Product Reference and	

Pricing Guide which may be found at www.sbc.com.

Issued: February 28, 2003

Effective: March 31, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

í

τ,

4.7

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

RECT MAR 07 2001 Original Sheet 232 Service Commission

#### **SECTION 4 - RATES AND CHARGES**

Norm Descoteaux, Regulatory Manager

4.7 Custom Business Services (continued)

> 4.7.5 Reserved for future use.

Issued: March 7, 2001

Effective part 2001 - 5

DEC 072001 5850 W. Las Positas Blvd., Pleasanton, California 94588 **Missouri** Public

> FILED DEC 072001 Service Commission

#### 4.7 Custom Business Services (continued)

4.7.6 Business Long Distance 200<sup>1</sup>

The per minute usage rates for outbound and TFS calls are as follows:

Ν

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.100
1 Year Term Plan	\$0.098

For fully automated, operator assisted, and operator dialed calls billed to the	T/D
Calling Card - Option 2, the usage rate is \$0.14 per minute. The per call	T/N
charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section	Ν
4.1.2 (C) of this Tariff.	N

<sup>1</sup> This Service is no longer available to new Customers effective June 3, 2002.

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

2nd Revised Sheet 233 Replacing 1st Revised Sheet 233

## SECTION 4 - RATES AND CHARGES Missouri Public

#### 4.7 Custom Business Services (continued)

4.7.6 Business Long Distance 200<sup>1</sup>

Customer Commitment	Rate Per Minute	
Month-to-Month	\$0.100	
1 Year Term Plan	\$0.098	

For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage rate if \$0.14 per minute with a per call charge of \$0.75.



# CANEELLED



<sup>1</sup> This Service is no longer available to new Customers effective June 3, 2002.

M - Material moved to Original Sheet 233.1

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Service Commission

REC'D AUG 01 2002

Service Commission

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

1st Revised Sheet 233 Replacing Original Sheet 233

Missouri Public Service Commission

**RECTIMAY 01 2002** 

С

#### SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services (continued)
  - 4.7.6 Business Long Distance 200<sup>1</sup>

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.100
1 Year Term Plan	\$0.098

For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage rate if \$0.14 per minute with a per call charge of \$0.75.

#### 4.7.7 Long Distance for Business

The usage rate is \$0.12 per minute for outbound and TFS calls. For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage rate if \$0.17 per minute with a per call charge of \$0.75.

SEP 01 2002

<sup>1</sup> This Service is no longer available to new Customers effective June 3, 2002.

Missouri Public Service Commission

Ν

<u>FILED AUG\_0 1 2002</u>

Issued: May 1, 2002

Effective:

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588 AUG 0 1 2002

CANCELLED

## **Missouri** Public

RECD MAR Sheet 732001

Service Commission

PSC Mo. - No. 1 Southwestern Bell Communications Services, Inc. d/b/a SBC Long Distance

#### **SECTION 4 - RATES AND CHARGES**

- 4.7 Custom Business Services (continued)
  - 4.7.6 **Business Long Distance 200**

Customer Commitment	Rate Per Minute	
Month-to-Month	\$0.100	
1 Year Term Plan	\$0.098	

AUG 0 1 2002 19572à

CANCELLED

For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage rate if \$0.14 per minute with a per call charge of \$0.75.

4.7.7 Long Distance for Business

> The usage rate is \$0.12 per minute for outbound and TFS calls. For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage rate if \$0.17 per minute with a per call charge of \$0.75.

Issued: March 7, 2001

Effective April 23 209

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

DEC 072001 Missouri Public

FILED DEC 07 2001 Service Commission

SBC Long Distance, Inc.

#### 4th Revised Sheet 233.1 Replacing 3rd Revised Sheet 233.1

ł

#### **SECTION 4 - RATES AND CHARGES**

4.7 Custom Business Services (continued)

4.7.7 Long Distance for Business

The usage rate is \$0.30 per minute for outbound and TFS calls. For fullyIautomated, operator assisted, and operator dialed calls billed to the Calling Card -Option 2, the usage rate is \$0.30 per minute. The per call charge may be found inISection 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Issued: March 15, 2005

Effective: April 1, 2005

Janet Vader, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

14



4.7 Custom Business Services (continued)

4.7.7 Long Distance for Business

The usage rate is \$0.25 per minute for outbound and TFS calls. For fullyIautomated, operator assisted, and operator dialed calls billed to the Calling Card -Option 2, the usage rate is \$0.25 per minute. The per call charge may be found inISection 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Issued: August 27, 2004

Effective: September 12, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

	western Bell Communications Services, Inc. PSC Mo No. 1	2nd Revised Sheet 233.1
<u>d/b/a</u>	SBC Long Distance Repla	cing 1st Revised Sheet 233.1
	SECTION 4 - RATES AND CHARGES	Missouri Public
4.7	Custom Business Services (continued)	RECDFEB13 2004
	4.7.7 Long Distance for Business	Service Commission

The usage rate is \$0.20 per minute for outbound and TFS calls. For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.17 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

CANCELLED SEP 1 ? 2004 3-425 233 \

By State Commission Public Service Commission MISSOURI

Issued: February 13, 2004

Effective: March 16, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Public Service Commission



Southwestern Bell Communications Services, Inc. J d/b/a SBC Long Distance

PSC Mo. - No. 1 1st Revised Sheet 233.1 Replacing Original Sheet 233.1

#### SECTION 4 - RATES AND CHARGES

#### 4.7 Custom Business Services (continued)

#### 4.7.7 Long Distance for Business

The usage rate is \$0.12 per minute for outbound and TFS calls. For fullyautomated, operator assisted, and operator dialed calls billed to the Calling CardT/D- Option 2, the usage rate is \$0.17 per minute. The per call charge may beT/Nfound in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of thisNTariff.N

# CANCELLED

MAR 1 6 2004 Public ervice Commission MISSOURI

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Public Servise Commission



# RFGT JAN 07 2003

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.7 Long Distance for Business

The usage rate is \$0.12 per minute for outbound and TFS calls. For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage rate if \$0.17 per minute with a per call charge of \$0.75.

CANCELLED

M - Material moved from 1st Revised Sheet 233

Issued: August 1, 2002

Effective: September 1, 2002 Missouri Public

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

FILED SEP 01 2002

Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

# Original Sheet 233.1

FEB 07 2003 By GRS 233.1 Public Service Cogninission

Micsouri Public

RECT AUG 01 20024

Service Commission

Μ

- 4.7 Custom Business Services (continued)
  - 4.7.8 Total Solutions Plus<sup>1</sup>

С

The usage rate is \$0.10 per minute for outbound and TFS calls. For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

Issued: January 28, 2004

Effective: February 28, 2004

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 2nd Revised Sheet 234 d/b/a SBC Long Distance Replacing 1st Revised Sheet 234

## SECTION 4 - RATES AND CHARGES Misseuri Public Sorvice Commission AF67 JAN 0 7 2003 Custom Business Services (continued) 4.7.8**Total Solutions Plus**

The usage rate is \$0.10 per minute for outbound and TFS calls. For fully automated, operator assisted, and operator dialed calls billed to the Calling Card T/D - Option 2, the usage rate is \$0.15 per minute. The per call charge may be T/N found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this N Tariff. Ν

CANCELLED

## FEB 2 8 2004 ommission Public SOI

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Missouri Public Invise Commission



4.7

		Bell Communications Services, Inc. ng Distance	PSC Mo No. 1	lst Revised Sheet 234 Replacing Original Sheet 234
		SECTION 4 - RAT	ES AND CHARGES	Missouri Public
4.7	Custor	m Business Services (continued)		REC'D AUG 01 2002
	4.7.8	Total Solutions Plus		Servi <mark>ce Commissio</mark>
		The usage rate is \$0.10 per minut automated calls billed to the Prop \$0.15 per minute with a per call of	orietary Calling Card	-



M - Material moved to Original Sheet 234.1M\* - Material moved to Original Sheet 234.1.1

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Service Commission

\_\_\_\_\_

Issued

## Missouri Public

RECT MAR 0 7 2001

Original Sheet 234 Service Commission

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services (continued)
  - 4.7.8 Total Solutions Plus

SEP 0.1 2002 By STRS 2.34

CANCELLED

The usage rate is \$0.10 per minute for outbound and TFS calls. For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage rate if \$0.15 per minute with a per call charge of \$0.75.

4.7.9 Business Long Distance 50

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.100
1 Year Term Plan	\$0.099

For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage rate if \$0.15 per minute with a per call charge of \$0.75.

-4.7.10 Business Domestic Saver

The usage rate is \$0.10 per minute for outbound and TFS calls. The minimum monthly usage charge is \$5.95 per month. For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage rate is \$0.16 per minute with a per call charge of \$0.75.

Issued: March 7, 2001

Effective April 23 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588 DEC 0 7 2001 Missouri Public

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED DEC 072001 01-475 Service Commission

#### 4.7 Custom Business Services (continued)

4.7.9 Business Long Distance 50

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.100
1 Year Term Plan	\$0.099
2 Year Term Plan N	\$0.099 N

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Issued: March 23, 2004

Effective: April 25, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Issued: October 28, 2003

SECTION 4 - RATES AND CHARGES

#### 4.7 Custom Business Services (continued)

4.7.9 **Business Long Distance 50** 

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.100
1 Year Term Plan	\$0.099

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.15 per minute. The per С call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

CANCELLED

APR 2 5 2004

REC'D OCT 2 8 2003

Missouri Public

Service Commission

3rd Revised Sheet 234.1 Replacing 2nd Revised Sheet 234.1

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance



---

Effective: December 1, 2003

Missoufi Public Servico Commination

FILFO NFC 01 2003

Southwestern Bell Communications Services, Inc. d/b/a SBC Long Distance

PSC Mo. - No. 1 2nd Revised Sheet 234.1 Replacing 1st Revised Sheet 234.1

Misseurl Public Bervies Cemmination SECTION 4 - RATES AND CHARGES REGD JAN 0 7 2003 Custom Business Services (continued)

4.7

4.7.9 Business Long Distance 50

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.100
1 Year Term Plan	\$0.099

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

T/D T/N Ν N

CANCELLED

DEC 0 1 2003

Issued: January 7, 2003

Effective: February 7, 2003

Missouri Public Service Commission

FILED FEB 0 7 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc.PSC Mo. - No. 11st Revised Sheet 234.1d/b/a SBC Long DistanceReplacing Original Sheet 234.1

SECTION 4 - RATES AND CHARGES

## Missouri Public

4.7 Custom Business Services (continued)

REC'D AUG 01 2002

Service Commission

4.7.9 Business Long Distance 50

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.100
1 Year Term Plan	\$0.099

For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage rate if \$0.15 per minute with a per call charge of \$0.75.

M - Material moved to Original Sheet 234.1.2 M\* - Material moved from Original Sheet 234.

# CANCELLED



Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Miseouri Public

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Service Commission

M\*

Μ

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

#### **SECTION 4 - RATES AND CHARGES**

Missouri Public

Original Sheet 234.1

#### 4.7 Custom Business Services (continued)

Service Commission

N

REC'D DEC 07 2001

4.7.11 Business Domestic Saver 15

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.1000
1 Year Term Plan	\$0.0990

For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage rate if \$0.15 per minute with a per call charge of \$0.75.

CANCELLED

SEP 01 2002 By ISPRS 234.1 Public Scribe Comunission Missouri

Issued: December 7, 2001

Effective: January 6, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED JAN 06 2002

Service Commission

#### 4.7 Custom Business Services (continued)

4.7.10 Business Domestic Saver

The usage rate is \$0.10 per minute for outbound and TFS calls. For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.16 per minute. The per call charge may C be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Issued: October 28, 2003

Effective: December 1, 2003

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. d/b/a SBC Long Distance

PSC Mo. - No. 1 2nd Revised Sheet 234.1.1 Replacing 1st Revised Sheet 234.1.1

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.10 Business Domestic Saver

The usage rate is \$0.10 per minute for outbound and TFS calls. For fullyautomated, operator assisted, and operator dialed calls billed to the Calling CardT/D- Option 2, the usage rate is \$0.16 per minute. The per call charge may beT/Nfound in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of thisNTariff.N

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Public Service Commi**ssion** 

FLED FEB 07 2003

CANCELLED

GES Miccourt Public Sorvice Commission REGA JAN 0 7 2003

Southwestern Bell Communications Services, Inc. d/b/a SBC Long Distance

PSC Mo. - No. 1 1st Revised Sheet 234.1.1 Replacing Original Sheet 234.1.1

#### SECTION 4 - RATES AND CHARGES

# Missouri Public

4.7 Custom Business Services (continued)

4.7.10 Business Domestic Saver

#### Service Commission

М

RECT NOV 2 6 2002

The usage rate is \$0.10 per minute for outbound and TFS calls. M

For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage rate is \$0.16 per minute with a per call charge of \$0.75.

# CANCELLED

# FEB 0.7 2003 By ZndRS 234.1.1 Public Service Commission

M - Material moved to 2nd Revised Sheet 194.1

Issued: November 26, 2002

Effective: December 27, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

## Missouri Public Service Commission FILED DEC 27 2002

	western Bell Communications Services, Inc. PSC Mo No. 1 SBC Long Distance	Original Sheet 234.1.1 Miscouri Public
	SECTION 4 - RATES AND CHARGE	RECT AUG 01 2002
4.7	Custom Business Services (continued)	Service CommissMn
	4.7.10 Business Domestic Saver	
	The usage rate is \$0.10 per minute for outbound and monthly usage charge is \$5.95 per month. For fully a Proprietary Calling Card - Option 2, the usage rate is call charge of \$0.75.	automated calls billed to the

CANCELLED

DEC 27 2002 1<sup>st</sup> RS 234.1.1 Public Service Commission

M - Material moved from Original Sheet 234

Issued: August 1, 2002

CANCELLED

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED SEP 01 2002

Service Commission

#### 4.7 Custom Business Services (continued)

4.7.11 Business Domestic Saver 15

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.1000
1 Year Term Plan	\$0.0990
2 Year Term Plan N	\$0.0990 N

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Issued: March 23, 2004

Effective: April 25, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 2nd Revised Sheet 234.1.2 d/b/a SBC Long Distance Replacing 1st Revised Sheet 234.1.2

> **Missouri** Public **SECTION 4 - RATES AND CHARGES**

#### 4.7 Custom Business Services (continued)

4.7.11 Business Domestic Saver 15

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.1000
1 Year Term Plan	\$0.0990

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

С

CANCELLED APR 2 5 2004 Public WI2

Issued: October 28, 2003

Effective: December 1, 2003

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Misseuri Public Servico Commiccien

FILED DEC 01 2003

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

RECT OCT 2 8 2003

Service Commission

Southwestern Bell Communications Services, Inc. d/b/a SBC Long Distance PSC Mo. - No. 1 1st Revised Sheet 234.1.2 Replacing Original Sheet 234.1.2

Replacing Original Sheet 234.1.2 Missouri Public Service Commission

#### SECTION 4 - RATES AND CHARGES

REGA JAN 0 7 2003

4.7 Custom Business Services (continued)

4.7.11 Business Domestic Saver 15

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.1000
1 Year Term Plan	\$0.0990

For fully automated, operator assisted, and operator dialed calls billed to theT/DCalling Card - Option 2, the usage rate is \$0.15 per minute. The per callT/Ncharge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and SectionN4.1.2 (C) of this Tariff.N

CANCELLED

DEC 0 1 2003 D11011

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Public Sorvigé Commission



Southwestern Bell Communications Services, Inc.	PSC Mo No.
d/b/a SBC Long Distance	

Original Sheet 234.1.2

<del>Missouri Pub</del>lic

### SECTION 4 - RATES AND CHARGES

RECTD AUG 01 2002

Service Commission

4.7 Custom Business Services (continued)

4.7.11 Business Domestic Saver 15

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.1000
1 Year Term Plan	\$0.0990

For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage rate if \$0.15 per minute with a per call charge of \$0.75.

CANCELLED

FER 07 2003

M - Material moved from Original Sheet 234.1

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Miccouri Public

М

FILED SEP 01 2002

Service Commission

# REC'D DEC 202001

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

Ę

4

- 4.7.12 SBC Long Distance Virtual Private Network (VPN)
  - (A) Usage Rates

The per minute usage rates are as follows:

	1-Year Term Plan	2-Year Term Plan	3-Year Term Plan
Call Rate Type A			
InterLATA	\$0.1330	\$0.1300	\$0.1270
IntraLATA	\$0.1280	\$0.1250	\$0.1230
Call Rate Type B	\$0.0750	\$0.0730	<u>\$0.0710</u>
Call Rate Type C	\$0.0650	\$0.0620	\$0.0590

(B) Per Call Charges

For remote access calls, a per call charge of \$0.25 applies in addition to the usage charge shown in Section 4.7.12 of this Tariff.

N

Ν

Issued: December 19, 2001

Effective: January 23, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED JAN 23 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Service Commission

Original Sheet 234.2

#### 4.7 Custom Business Services (continued)

4.7.13 Business Long Distance 100

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute	
Month-to-Month	\$0.1000	
1-Year Term Plan	\$0.0980	
2-Year Term Plan N	\$0.0980 N	

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.14 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Issued: March 23, 2004

Effective: April 25, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 6th Revised Sheet 234.3 d/b/a SBC Long Distance Replacing 5th Revised Sheet 234.3

#### **SECTION 4 - RATES AND CHARGES**

#### 4.7 Custom Business Services (continued)

REC'D OCT 2 8 2003

Missouri Public

4.7.13 Business Long Distance 100

Service Commission

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute	
Month-to-Month	\$0.1000	
1-Year Term Plan	\$0.0980	

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.14 per minute. The per С call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.



Issued: October 28, 2003

Effective: December 1, 2003

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Servico Commicsion

FILED DEC 01 2003
Southwestern Bell Communications Services, Inc.
 PSC Mo. - No. 1
 5th Revised Sheet 234.3

 <u>d/b/a SBC Long Distance</u>
 Replacing 4th Revised Sheet 234.3

#### SECTION 4 - RATES AND CHARGES

#### 4.7 Custom Business Services (continued)

4.7.13 Business Long Distance 100

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.1000
1-Year Term Plan	\$0.0980

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.14 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

# CANCELLED



Issued: April 15, 2003

Effective: May 16, 2003

Missouri Public Service Commission

FILED MAY 16 2003

Norm Descoteaux. Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 RECD APR 15 2003

Missouri Public Service Commission

Ν

Southwestern Bell Communications Services, Inc. d/b/a SBC Long Distance

#### SECTION 4 - RATES AND CHARGES

#### 4.7 Custom Business Services (continued)

4.7.13 Business Long Distance 100

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.1000

For fully automated, operator assisted, and operator dialed calls billed to theT/DCalling Card - Option 2, the usage rate is \$0.14 per minute. The per callT/Ncharge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and SectionN4.1.2 (C) of this Tariff.N

## CANCELLED



Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

#### Missouri Public Service Commission

FILED FEB 07 2003

RFGD JAN 0 7 2003

 Southwestern Bell Communications Services, Inc.
 PSC Mo. - No. 1
 3rd Revised Sheet 234.3

 d/b/a SBC Long Distance
 Replacing 2nd Revised Sheet 234.3

#### SECTION 4 - RATES AND CHARGES

#### 4.7 Custom Business Services (continued)

4.7.13 Business Long Distance 100

Service Commission

Μ

M

**M\*** 

**M\*** 

REC'D AUG 01 2002

Missouri Public

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.1000

For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage rate if \$0.14 per minute with a per call charge of \$0.75.

CANCELLED 07 2003

M - Material moved to Original Sheet 234.3.1 M\* - Material moved to Original Sheet 234.3.2

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Micsouri Public

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Service Commission

#### SECTION 4 - RATES AND CHARGES

#### 4.7 Custom Business Services (continued)

4.7.13 Business Long Distance 100

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.1000

For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage rate if \$0.14 per minute with a per call charge of \$0.75.

4.7.14 Business Block of Time 200

The MRC is \$15.00 per BTN for a 200 minute block of time as described in Section 3.7.14 of this Tariff. For Direct-Dialed TFS and outbound calls, the rate is \$0.10 per minute for calls completed after the 200 minute block of time has been used. For fully automated calling card calls billed to the Proprietary Calling Card - Option 2, the rate is \$0.15 per minute.

4.7.15 Business Block of Time 400

The MRC is \$30.00 per BTN for a 400 minute block of time as described in Section 3.7.15 of this Tariff. For Direct-Dialed TFS and outbound calls, the rate is \$0.10 per minute for calls completed after the 400 minute block of time has been used. For fully automated calling card calls billed to the Proprietary Calling Card - Option 2, the rate is \$0.15 per minute.



Missouri Public Service Commission FILED AUG 01 2002

Issued: May 1, 2002

Effective:

Norm Descoteaux, Regulatory Manager AUG 0 1 2002 5850 W. Las Positas Blvd.; Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

#### Missourt Public Service Commission

RECTI MAY 01 2002

| N

Т

Ν

1st Revised Sheet 234.3 Replacing Original Sheet 234.3

#### SECTION 4 - RATES AND CHARGES

Missouri Public

REC'D MAY 01 2002

Service Commission

4.7 Custom Business Services (continued)

4.7.13 Reserved for future use

4.7.14 Business Block of Time 200

The MRC is \$15.00 per BTN for a 200 minute block of time as described in Section T 3.7.14 of this Tariff. For Direct-Dialed TFS and outbound calls, the rate is \$0.10 per minute for calls completed after the 200 minute block of time has been used. For fully automated calling card calls billed to the Proprietary Calling Card - Option 2, the rate is \$0.15 per minute.

4.7.15 Business Block of Time 400

The MRC is \$30.00 per BTN for a 400 minute block of time as described in Section T 3.7.15 of this Tariff. For Direct-Dialed TFS and outbound calls, the rate is \$0.10 per minute for calls completed after the 400 minute block of time has been used. For fully automated calling card calls billed to the Proprietary Calling Card - Option 2, the rate is \$0.15 per minute.



Missouri Public

FILED JUN 01 2002

Service Commission

Effective: June 1, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Issued: May 1, 2002

Original Sheet 234.3

		SECTION 4 - RATES AND CHARGES	Missouri Public
4.7	Custon	n Business Services (continued)	REC'D FEB 26 2002
	4.7.13	Reserved for future use	Service Commission
	4.7.14	Business Block of Time 200	
		The MRC is \$15.00 per line for a 200 minute block of 3.7.14 of this Tariff. For Direct-Dialed TFS and outbour minute for calls completed after the 200 minute block fully automated calling card calls billed to the Propriet the rate is \$0.15 per minute.	Ind calls, the rate is \$0.10 per
	4.7.15	Business Block of Time 400	
		The MRC is \$30.00 per line for a 400 minute block of 3.7.15 of this Tariff. For Direct-Dialed TFS and outbour minute for calls completed after the 400 minute block fully automated calling card calls billed to the Propriet	Ind calls, the rate is \$0.10 per

the rate is \$0.15 per minute.

## CANCELLED

JUN 01 2002 By SARS 234.3 Public Service Commission MISSOURI

Issued: February 26, 2002

Effective: April 1, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

FILED APR 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

## Service Commission

Ν

4.7 Custom Business Services (continued)

4.7.14 Business Block of Time 200<sup>1</sup>

С

The MRC is \$15.00 per BTN for a 200 minute block of time as described in Section 3.7.14 of this Tariff. For Direct-Dialed TFS and outbound calls, the rate is \$0.10 per minute for calls completed after the 200 minute block of time has been used. For fully automated, operator assisted, and operator dialed calling card calls billed to the Calling Card - Option 2, category 11, the rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations N effective April 1, 2004.

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

## SECTION 4 - RATES AND CHARGES Missouri Public

4.7 Custom Business Services (continued)

4.7.14 Business Block of Time 200

REC'D 007 2 8 2003

Service Commission

С

The MRC is \$15.00 per BTN for a 200 minute block of time as described in Section 3.7.14 of this Tariff. For Direct-Dialed TFS and outbound calls, the rate is \$0.10 per minute for calls completed after the 200 minute block of time has been used. For fully automated, operator assisted, and operator dialed calling card calls billed to the Calling Card - Option 2, category 11, the rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

CANCELLED



Issued: October 28, 2003

Effective: December 1, 2003

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

s

Missouri Public Servico Commission

FILED DEC 01 2003

Southwestern Bell Communications Services, Inc. d/b/a SBC Long Distance

#### SECTION 4 - RATES AND CHARGES

Misseuri Public Service Commission

REGD JAN 0.7 2003

Т

Ν

N

T/D

4.7 Custom Business Services (continued)

4.7.14 Business Block of Time 200

The MRC is \$15.00 per BTN for a 200 minute block of time as described in Section 3.7.14 of this Tariff. For Direct-Dialed TFS and outbound calls, the rate is \$0.10 per minute for calls completed after the 200 minute block of time has been used. For fully automated, operator assisted, and operator dialed calling card calls billed to the Calling Card - Option 2, the rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.



DEC 01 2003 2nd RS 23U.3.1 2nd RS 23U.3.1

Issued: January 7, 2003

Effective: February 7, 2003

1 14

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Public Service Commission

FILED FEB 072003

#### SECTION 4 - RATES AND CHARGES

#### 4.7 Custom Business Services (continued)

4.7.14 Business Block of Time 200

The MRC is \$15.00 per BTN for a 200 minute block of time as described in Section 3.7.14 of this Tariff. For Direct-Dialed TFS and outbound calls, the rate is \$0.10 per minute for calls completed after the 200 minute block of time has been used. For fully automated calling card calls billed to the Proprietary Calling Card - Option 2, the rate is \$0.15 per minute.

M - Material moved from 2nd Revised Sheet 234.3

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED SEP 01 2002

Servi**ce Commissio**n

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

#### Original Sheet 234.3.1

## REC'D AUG 01 2002

Miggouri Public

Service Commission

Μ



4.7 Custom Business Services (continued)

4.7.15 Business Block of Time  $400^1$ 

The MRC is \$30.00 per BTN for a 400 minute block of time as described in Section 3.7.15 of this Tariff. For Direct-Dialed TFS and outbound calls, the rate is \$0.10 per minute for calls completed after the 400 minute block of time has been used. For fully automated, operator assisted, and operator dialed calling card calls billed to the Calling Card - Option 2, category 11, the rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations N effective April 1, 2004.

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. d/b/a SBC Long Distance

#### SECTION 4 - RATES AND CHARGES

## Missouri Public Service Commission RECTO JAN 07 2003

T

Ν

N

T/D

4.7 Custom Business Services (continued)

4.7.15 Business Block of Time 400

The MRC is \$30.00 per BTN for a 400 minute block of time as described in Section 3.7.15 of this Tariff. For Direct-Dialed TFS and outbound calls, the rate is \$0.10 per minute for calls completed after the 400 minute block of time has been used. For fully automated, operator assisted, and operator dialed calling card calls billed to the Calling Card - Option 2, the rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

CANCELLED

FEB 2 8 2004 34.3.2 mission

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Public Sorvice Commission FILED FEB 0 7 2003

#### SECTION 4 - RATES AND CHARGES

#### 4.7 Custom Business Services (continued)

4.7.15 Business Block of Time 400

The MRC is \$30.00 per BTN for a 400 minute block of time as described in Section 3.7.15 of this Tariff. For Direct-Dialed TFS and outbound calls, the rate is \$0.10 per minute for calls completed after the 400 minute block of time has been used. For fully automated calling card calls billed to the Proprietary Calling Card - Option 2, the rate is \$0.15 per minute.

CANCELLED

M - Material moved from 2nd Revised Sheet 234.3

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Service Commission

Original Sheet 234.3.2

Missouri Public

REC'D AUG 01 2002

Service Commission

Μ

## FEB 07 2003 FEB 07 2003 FIRS 234.3.2 Public Senting Countinssion





4.7 Custom Business Services (continued)

4.7.16 Business Domestic Saver Deluxe<sup>1</sup>

The usage rate is \$0.1000 per minute for outbound and TFS calls. For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.16 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

N N

Ν

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

#### **Missouri** Public

4.7 Custom Business Services (continued)

## REC'D OCT 2 8 2003

4.7.16 Business Domestic Saver Deluxe

#### Service Commission

The usage rate is \$0.1000 per minute for outbound and TFS calls. For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.16 per minute. The per call charge may C be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

CANCELLED

FEB 2 8 2004

Issued: October 28, 2003

Effective: December 1, 2003

Amission

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Servico Commicolon

FLED DEC 01 2003

Southwestern Bell Communications Services, Inc. d/b/a SBC Long Distance PSC Mo. - No. 1 3rd Revised Sheet 234.4 Replacing 2nd Revised Sheet 234.4

SECTION 4 - RATES AND CHARGES

#### Missouri Public Servicé Commission

4.7 Custom Business Services (continued)

RECD JAN 0 7 2003

4.7.16 Business Domestic Saver Deluxe

The usage rate is \$0.1000 per minute for outbound and TFS calls. For fully automated, operator assisted, and operator dialed calls billed to the Calling Card T/D - Option 2, the usage rate is \$0.16 per minute. The per call charge may be T/N found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this N Tariff.

CANCELLED

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

#### Missouri Public Service Commi**asion**



- No. 1 2nd Revised Sheet 234.4 Replacing 1st Revised Sheet 234.4

#### SECTION 4 - RATES AND CHARGES

## Missouri Public

4.7 Custom Business Services (continued)

# REC'D NOV 2 6 2002

4.7.16 Business Domestic Saver Deluxe

## Service Commission

M M

The usage rate is \$0.1000 per minute for outbound and TFS calls.

For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage rate is \$0.16 per minute with a per call charge of \$0.75.

M - Material moved to 1st Revised Sheet 194.33

# CANCELLED

FEB 07 2003

Issued: November 26, 2002

Effective: December 27, 2002

Tracy Van Wormer, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

## FILED DEC 27 2002

lst Revised Sheet 234.4 Replacing Original Sheet 234.4

#### Miccouri Public

#### **SECTION 4 - RATES AND CHARGES**

#### REC'D AUG 01 2002

4.7 Custom Business Services (continued)

Service Commission

Μ

Μ

4.7.16 Business Domestic Saver Deluxe

The usage rate is \$0.1000 per minute for outbound and TFS calls. The minimum monthly usage charge is \$5.95 per month. For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage rate is \$0.16 per minute with a per call charge of \$0.75.

#### CANCELLED

DEC 27 2002 ANI RS 234.4 Public Service Coordination Missouri

M - Material moved to Original Sheet 234.4.1

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Miccouri Public

FILED SEP 01 2002

Service Commission

d/b/a SBC Long Distance Missouri Public **SECTION 4 - RATES AND CHARGES** REC'D JUN 2 7 2002 4.7 Custom Business Services (continued) Service Commission 4.7.16 Business Domestic Saver Deluxe The usage rate is \$0.1000 per minute for outbound and TFS calls. The minimum monthly usage charge is \$5.95 per month. For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage rate is \$0.16 per minute with a per call charge of \$0.75. 4.7.17 Business Domestic Saver 15 Deluxe The per minute usage rates for outbound and TFS calls are as follows:

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.1000
1 Year Term Plan	\$0.0990

rate is \$0.15 per minute with a per call charge of \$0.75.

Ν

Effective: August 1, 2002

Missouri Public

FILED AUG 01 2002

Service Commission

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Issued: June 27, 2002

For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage



# SEP 01 2002



Original Sheet 234.4

#### 4.7 Custom Business Services (continued)

4.7.17 Business Domestic Saver 15 Deluxe<sup>1</sup>

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.1000
1 Year Term Plan	\$0.0990

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations N effective April 1, 2004.

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 2nd Revised Sheet 234.4.1 d/b/a SBC Long Distance Replacing 1st Revised Sheet 234.4.1

#### **SECTION 4 - RATES AND CHARGES**

#### 4.7 Custom Business Services (continued)

## Missouri Public

#### 4.7.17 Business Domestic Saver 15 Deluxe

#### Service Commission

**REC'D** OCT 2 8 2003

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.1000
1 Year Term Plan	\$0.0990

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.15 per minute. The per C call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

CANCELLED

#### FEB 2 8 2004 By Service Commission Public Service Commission MISSOURI

Issued: October 28, 2003

Effective: December 1, 2003

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Servico Commission

FILED DEC 01 2003

Southwestern Bell Communications Services, Inc. d/b/a SBC Long Distance PSC Mo. - No. 1 1st Revised Sheet 234.4.1 Replacing Original Sheet 234.4.1

#### SECTION 4 - RATES AND CHARGES

#### 4.7 Custom Business Services (continued)

4.7.17 Business Domestic Saver 15 Deluxe

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute	
Month-to-Month	\$0.1000	
1 Year Term Plan	\$0.0990	

For fully automated, operator assisted, and operator dialed calls billed to theT.DCalling Card - Option 2, the usage rate is \$0.15 per minute. The per callT/Ncharge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and SectionN4.1.2 (C) of this Tariff.N

CANCELLED

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

#### Missouri Public Sorvies Commission RÉGD JAN 07 2003

Missouri Public Servise Commission FILED FEB 0 7,2003

Original Sheet 234.4.1

Missouri Public

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

Service Commission

REC'D AUG 01 200

4.7.17 Business Domestic Saver 15 Deluxe

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
Month-to-Month	\$0.1000
1 Year Term Plan	\$0.0990

For fully automated calls billed to the Proprietary Calling Card - Option 2, the usage rate is \$0.15 per minute with a per call charge of \$0.75.

M - Material moved from Original Sheet 234.4

## CANCELLED

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Micsouri Public

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Servi**ce Commissio**n

M

#### 4.7 Custom Business Services (continued)

4.7.18 Business Domestic Saver 15 Connections 3 Service<sup>1</sup>

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0950

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

N N

Issued: February 3, 2004

Effective: March 5, 2004

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.18 Business Domestic Saver 15 Connections 3 Service

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0950

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

#### CANCELLED

MISSOURI

Issued: February 21, 2003

Effective: March 23, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Publio Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

## RECT) FEB 21 2003

Ν

Missouri Public Service Gemmicalon

MAR 0 5 2004 15t RS 23 Bγ Public Service Commission

FILED MAR 23 2003

## Original Sheet 234.4.2

#### 4.7 Custom Business Services (continued)

4.7.19 Business Long Distance 50 Connections 3 Service<sup>1</sup>

С

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0950

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

#### SECTION 4 - RATES AND CHARGES

4.7.19 Business Long Distance 50 Connections 3 Service

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0950

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

CANCELLED FEB 2 8 2004 4.3 munission

Issued: February 21, 2003

Effective: March 23, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FILED MAR 23 2003

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Original Sheet 234.4.3

Missouri Public Service Germicalon

REGD FEB 21 2003

Ν

#### 4.7 Custom Business Services (continued)

4.7.20 Business Long Distance 100 Connections 3 Service<sup>1</sup>

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0940

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.14 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

N N

Issued: February 3, 2004

Effective: March 5, 2004

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.20 Business Long Distance 100 Connections 3 Service (continued)

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0940

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.14 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Ν

## CANCELLED

MAR 0 5 2004 By PRS 234.4.4 Public Service Commission MISSOURI

Issued: February 21, 2003

Effective: March 23, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FILED MAR 23 2003

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Original Sheet 234.4.4



REGID FEB 21 2003

## Distance

#### 4.7 Custom Business Services (continued)

4.7.21 Business Domestic Saver 15 Connections 2 Service

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0960

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.15 per minute. The per C call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Issued: October 28, 2003

Effective: December 1, 2003

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

#### SECTION 4 - RATES AND CHARGES

4.7.21 Business Domestic Saver 15 Connections 2 Service

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0960

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.



Issued: February 21, 2003

Effective: March 23, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Missouri Public Service Commission

FILED MAR 23 2003

Original Sheet 234.4.5

Missouri Public Service Commission

REGD FEB 21 2003

#### 4.7 Custom Business Services (continued)

4.7.22 Business Long Distance 50 Connections 2 Service

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0960

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.15 per minute. The per C call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Issued: October 28, 2003

Effective: December 1, 2003

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.22 Business Long Distance 50 Connections 2 Service

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
I Year Term Plan	\$0.0960

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Issued: February 21, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Missouri Public Service Commission

FILED MAR 23 2003

Effective: March 23, 2003



Missouri Public Sorvice Commission REGD FEB 2 1 2003

Original Sheet 234.4.6

Ν

#### 4.7 Custom Business Services (continued)

4.7.23 Business Long Distance 100 Connections 2 Service (continued)

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0950

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.14 per minute. The per C call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Issued: October 28, 2003

Effective: December 1, 2003

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.23 Business Long Distance 100 Connections 2 Service (continued)

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0950

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.14 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Issued: February 21, 2003

Effective: March 23, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

#### FILED MAR 23 2003

Missouri Public Service Commission



RECT) FEB 21 2003

Original Sheet 234.4.7



Ν



#### 4.7 Custom Business Services (continued)

4.7.24 Business Domestic Saver 15 Connections 1 Service

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0970

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.15 per minute. The per C call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Issued: October 28, 2003

Effective: December 1, 2003

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588
Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

SECTION 4 - RATES AND CHARGES

Original Sheet 234.4.8 <u>Missouri Public</u> Service Commission

RECD FEB 21 2003

### Custom Business Services (continued)

4.7

4.7.24 Business Domestic Saver 15 Connections 1 Service

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0970

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.



Issued: February 21, 2003

Effective: March 23, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

### FILED MAR 23 2003

Ν

#### SECTION 4 - RATES AND CHARGES

#### 4.7 Custom Business Services (continued)

4.7.25 Business Long Distance 50 Connections 1 Service

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0970

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.15 per minute. The per C call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Issued: October 28, 2003

Effective: December 1, 2003

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.25 Business Long Distance 50 Connections 1 Service

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0970

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.



Issued: February 21, 2003

Effective: March 23, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED MAR 23 2003



Missouri Public Service Commission

RECD FEB 2 1 2003

Southwestern Bell Communications Services, Inc.PSC Mo. - No. 11st Revised Sheet 234.4.10d/b/a SBC Long DistanceReplacing Original Sheet 234.4.10

#### SECTION 4 - RATES AND CHARGES

#### 4.7 Custom Business Services (continued)

4.7.26 Business Long Distance 100 Connections 1 Service (continued)

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0960

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.14 per minute. The per C call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Issued: October 28, 2003

Effective: December 1, 2003

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

#### SECTION 4 - RATES AND CHARGES

Original Sheet 234.4.10

## RECD FEB 21 2003

Missouri Public Service Commission

4.7 Custom Business Services (continued)

4.7.26 Business Long Distance 100 Connections 1 Service (continued)

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0960

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.14 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Issued: February 21, 2003

Effective: March 23, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED MAR 23 2003

Missouri Public Service Commission

### Ν

#### SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services (continued)
  - 4.7.27 High Volume Calling II Plus
    - (A) Outbound Calls
      - .1 High Volume Outbound Calling II Plus

f	follows:	0				
		Per Minute Rate	Per Minute Rate	N	Per Minute Rate	N

The per minute usage rates for intrastate InterLATA calls are as

	Per Minute Rate	Per Minute Rate		Per Minute Rate	N
MAC	1 Year Term Plan	2 Year Term Plan		3 Year Term Plan	
\$600	\$0.0680	\$0.0680		\$0.0680	
\$2,400	\$0.0680	\$0.0680		\$0.0680	
\$6,000	\$0.0680	\$0.0680		\$0.0680	
\$9,000	\$0.0680	\$0.0680		\$0.0680	
\$12,000	\$0.0680	\$0.0680		\$0.0680	
\$18,000	\$0.0680	\$0.0680		\$0.0680	
\$24,000	\$0.0680	\$0.0680		\$0.0680	
\$30,000	\$0.0680	\$0.0680		\$0.0680	
\$42,000	\$0.0680	\$0.0680		\$0.0680	
\$60,000	\$0.0680	\$0.0680		\$0.0680	
\$90,000	\$0.0680	\$0.0680		\$0.0680	
\$120,000	\$0.0680	\$0.0680		\$0.0680	
\$180,000	\$0.0680	\$0.0680		\$0.0680	
\$240,000	\$0.0680	\$0.0680	Ν	\$0.0680	N

Issued: March 12, 2004

Effective: April 12, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 2nd Revised Sheet 234.5 d/b/a SBC Long Distance 234.5

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

-----

4.7.27 High Volume Calling II Plus

- (A) Outbound Calls
  - .1 High Volume Outbound Calling II Plus

The per minute usage rates for intrastate InterLATA calls are as follows:

	Per Minute Rate
MAC	1 Year Term Plan
\$600	\$0.0680
\$2,400	\$0.0680
\$6,000	\$0.0680
\$9,000	\$0.0680
\$12,000	\$0.0680
\$18,000	\$0.0680
\$24,000	\$0.0680
\$30,000	\$0.0680
\$42,000	\$0.0680
\$60,000	\$0.0680
\$90,000	\$0.0680
\$120,000	\$0.0680
\$180,000	\$0.0680
\$240,000	\$0.0680

Missouri Public

**REC'D** AUG 1 5 2003

#### Service Commission

Ν

N N

Ν

Ν

CANCELLED

APR 1 2 2004 By 3<sup>rd</sup> RS 234.5 Public Service Commission MISSOURI

Issued: August 15, 2003

Effective: September 15, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missourt Public Service Commission



# Missouri Public Service Commission

# REGD JUN 05 2003

R

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

----

1st Revised Sheet 234.5 Replacing Original Sheet 234.5

#### **SECTION 4 - RATES AND CHARGES**

4.7 Custom Business Services (continued)

÷

4.7.27 High Volume Calling II Plus

- **Outbound Calls** (A)
  - High Volume Outbound Calling II Plus .1

The per minute usage rates for intrastate InterLATA calls are as follows:

	Per Minute Rate
MAC	1 Year
	Term Plan
\$600	\$0.0680
\$2,400	\$0.0680
\$6,000	\$0.0680
\$12,000	\$0.0680
\$30,000	\$0.0680
\$60,000	\$0.0680
\$120,000	\$0.0680
\$180,000	\$0.0680
\$240,000	\$0.0680

SEP 1 5 2003 Public Commission Se evice. MISSOURI

CANCELLED

Issued: June 5, 2003

Effective: June 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

-----

Missouri Public Service Commission

FILED JUN 1 8 2003

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

> **SECTION 4 - RATES AND CHARGES** Misseuri Public Service Commission

4.7 Custom Business Services (continued)

4.7.27 High Volume Calling II Plus

- **Outbound Calls** (A)
  - High Volume Outbound Calling II Plus, .1 mission

The per minute usage rates for intrastate InterLATA calls are as follows:

	Per Minute Rate
MAC	1 Year
	Term Plan
\$600	\$0.0990
\$2,400	\$0.0980
\$6,000	\$0.0970
\$12,000	\$0.0960
\$30,000	\$0.0950
\$60,000	\$0.0940
\$120,000	\$0.0930
\$180,000	\$0.0920
\$240,000	\$0.0910

Issued: March 31, 2003

Effective: April 30, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

# FILED APR 80-2003

Missouri Public Service Commission

N

JUN 1 6 2003

Original Sheet 234.5

RF(1) MAR 81 2003

CANCELLED

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

#### 4.7.27 High Volume Calling II Plus (continued)

- (A) Outbound Calls (continued)
  - .1 High Volume Outbound Calling II Plus (continued)

	Per Minute Rate	Per Minute Rate	N	Per Minute Rate	N
MAC	1 Year Term Plan	2 Year Term Plan		3 Year Term Plan	
\$600	\$0.0680	\$0.0680		\$0.0680	
\$2,400	\$0.0680	\$0.0680		\$0.0680	
\$6,000	\$0.0680	\$0.0680		\$0.0680	
\$9,000	\$0.0680	\$0.0680		\$0.0680	
\$12,000	\$0.0680	\$0.0680		\$0.0680	
\$18,000	\$0.0680	\$0.0680		\$0.0680	
\$24,000	\$0.0680	\$0.0680		\$0.0680	
\$30,000	\$0.0680	\$0.0680		\$0.0680	
\$42,000	\$0.0680	\$0.0680		\$0.0680	
\$60,000	\$0.0680	\$0.0680		\$0.0680	
\$90,000	\$0.0680	\$0.0680		\$0.0680	
\$120,000	\$0.0680	\$0.0680		\$0.0680	
\$180,000	\$0.0680	\$0.0680		\$0.0680	
\$240,000	\$0.0680	\$0.0680	Ν	\$0.0680	Ν

The per minute usage rates for intrastate IntraLATA calls are as follows:

Issued: March 12, 2004

Effective: April 12, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 2nd Revised Sheet 234.6 d/b/a SBC Long Distance \_\_\_\_\_\_\_ Replacing 1st Revised Sheet 234.6

#### SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services (continued)
  - 4.7.27 High Volume Calling II Plus (continued)
    - (A) Outbound Calls (continued)
      - .1 High Volume Outbound Calling II Plus (continued)

The per minute usage rates for intrastate IntraLATA calls are as follows:

	Per Minute Rate	
MAC	1 Year Term Plan	
\$600	\$0.0680	
\$2,400	\$0.0680	
\$6,000	\$0.0680	
\$9,000	\$0.0680	
\$12,000	\$0.0680	
\$18,000	\$0.0680	
\$24,000	\$0.0680	
\$30,000	\$0.0680	
\$42,000	\$0.0680	
\$60,000	\$0.0680	
\$90,000	\$0.0680	
\$120,000	\$0.0680	
\$180,000	\$0.0680	
\$240,000	\$0.0680	

CANCELLED

APR 1 2 2004 ice Commission MISSOURI

Issued: August 15, 2003

Effective: September 15, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FILED SEP 15 2003

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

### Missouri Public

**RECD** AUG 1 5 2003

Ν

N N

Ν

Ν



Missouri Public Service Commission

## REC'D JUN 05 2003

R

R

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u>

1st Revised Sheet 234.6 Replacing Original Sheet 234.6

- - -

#### SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services (continued)
  - 4.7.27 High Volume Calling II Plus (continued)
    - (A) Outbound Calls (continued)
      - .1 High Volume Outbound Calling II Plus (continued)

The per minute usage rates for intrastate IntraLATA calls are as follows:

	Per Minute Rate			
MAC	1 Year			
	Term Plan			
\$600	\$0.0680			
\$2,400	\$0.0680 \$0.0680 \$0.0680			
\$6,000				
\$12,000				
\$30,000	\$0.0680			
\$60,000	\$0.0680			
\$120,000	\$0.0680			
\$180,000	\$0.0680			
\$240,000	\$0.0680			

Issued: June 5, 2003

Effective: June 16, 2003

CANCELLED

SER 1 5 2003

Public Service Commission MISSOURI

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

----

Missouri Public Sorvico Commission

FILED JUN 1 8 2003

			SECTION 4 - RAT	ES AND CHARGES	
4.7	Custom Bus	iness So	Missouri Public Service Commission		
	4.7.27 High	7.27 High Volume Calling II Plus (continued)		RECT) MAR 81 2003	
	(A)	Outbo	ound Calls (continued	1)	
		.1	High Volume Outbo	und Calling II Plus (c	continued)
			follows: MAC	Per Minute Rate	
			\$600	Term Plan \$0.0990	
			\$2,400	\$0.0980	CANCELLED
			\$6,000	\$0.0970	JUN 1 6 2003
			\$12,000	\$0.0960	By Sha Commission
			\$30,000	\$0.0950	Public Service Missing
			\$60,000	\$0.0940	
			\$120,000	\$0.0930	
			\$180,000	\$0.0920	
			\$240,000	\$0.0910	

Issued: March 31, 2003

Effective: April 30, 2003

Missouri Public Service Commission

FILED APR **80** 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

. i.

Southwestern Bell Communications Services, Inc.PSC Mo. - No. 13rd Revised Sheet 234.7d/b/a SBC Long DistanceReplacing 2nd Revised Sheet 234.7

#### SECTION 4 - RATES AND CHARGES

#### 4.7 Custom Business Services (continued)

- 4.7.27 High Volume Calling II Plus (continued)
  - (A) Outbound Calls (continued)
    - .2 High Volume Dedicated Outbound Calling II Plus

	Per Minute	Per Minute	N	Per Minute	Ν
	Rate	Rate		Rate	
MAC	1 Year Term Plan	2 Year Term Plan		3 Year Term Plan	
\$600	\$0.0580	\$0.0580		\$0.0580	
\$2,400	\$0.0580	\$0.0580		\$0.0570	
\$6,000	\$0.0580	\$0.0570		\$0.0560	
\$9,000	\$0.0580	\$0.0570		\$0.0560	
\$12,000	\$0.0570	\$0.0560		\$0.0550	
\$18,000	\$0.0570	\$0.0560		\$0.0550	
\$24,000	\$0.0570	\$0.0560		\$0.0550	
\$30,000	\$0.0560	\$0.0550		\$0.0540	
\$42,000	\$0.0560	\$0.0550		\$0.0540	
\$60,000	\$0.0550	\$0.0540		\$0.0530	
\$90,000	\$0.0550	\$0.0540		\$0.0530	
\$120,000	\$0.0540	\$0.0530		\$0.0520	
\$180,000	\$0.0530	\$0.0520		\$0.0510	
\$240,000	\$0.0520	\$0.0510	Ν	\$0.0500	Ν

The per minute usage rates for InterLATA calls are as follows.

Issued: March 12, 2004

Effective: April 12, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 2nd Revised Sheet 234.7 <u>d/b/a SBC Long Distance</u> Replacing 1st Revised Sheet 234.7

#### SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services (continued)
  - 4:7.27 High Volume Calling II Plus (continued)
    - (A) Outbound Calls (continued)
      - .2 High Volume Dedicated Outbound Calling II Plus

The per minute usage rates for InterLATA calls are as follows.

	Per Minute Rate		
MAC	l Year Term Plan		
\$600	\$0.0580		
\$2,400	\$0.0580		
\$6,000	\$0.0580		
\$9,000	\$0.0580		
\$12,000	\$0.0570		
\$18,000	\$0.0570		
\$24,000	\$0.0570		
\$30,000	\$0.0560		
\$42,000	\$0.0560		
\$60,000	\$0.0550		
\$90,000	\$0.0550		
\$120,000	\$0.0540		
\$180,000	\$0.0530		
\$240,000	\$0.0520		

### CANCELLED

Ν

N N

Ν

Ν



Issued: August 15, 2003

Effective: September 15, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missourt Public Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED SEP 15 2003

#### Missouri Public

**REC'D** AUG 1 5 2003

Service Commission

## RECT) JUN 05 2003

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1
d/b/a SBC Long Distance Replace Replace

1st Revised Sheet 234.7 Replacing Original Sheet 234.7

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

\_.....

4.7.27 High Volume Calling II Plus (continued)

- (A) Outbound Calls (continued)
  - .2 High Volume Dedicated Outbound Calling II Plus

The per minute usage rates for InterLATA calls are as follows.

	Per Minute Rate		
MAC	1 Year		
	Term Plan		
\$600	\$0.0580		
\$2,400	\$0.0580		
\$6,000	\$0.0580		
\$12,000	\$0.0570		
\$30,000	\$0.0560		
\$60,000	\$0.0550		
\$120,000	\$0.0540		
\$180,000	\$0.0530		
\$240,000	\$0.0520		

CANCELLED

R

R

SEP 1 5 2003 by 2<sup>rd</sup> RS 234.7 Public Service Commission MISSOUPI

Issued: June 5, 2003

Effective: June 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Miseouri Public Servico Commission

FILED JUN 1 6 2003

			SECTION 4 -	RATES AND CHARC	BES Missourt Public Service Commission
7	Custom	Business	s Services (continu	ed)	RECD MAR <b>81</b> 2003
	4.7.27	High Vo	lume Calling II Plu	us (continued)	
		(A) Ou	tbound Calls (cont	tinued)	
		.2	High Volume I	Dedicated Outbound Ca	alling II Plus
			The per minute		ATA calls are as follows.
				Per Minute Rate	
			MAC	1 Year Term Plan	CANCELLED
		_	\$600		
			. <u></u>	Term Plan	JUN 1 6 2003
			\$600	Term Plan \$0.0720	JUN 1 6 2003 By 1 37 R 5 2 34
			\$600 \$2,400	Term Plan \$0.0720 \$0.0700	JUN 1 6 2003
			\$600 \$2,400 \$6,000	Term Plan           \$0.0720           \$0.0700           \$0.0680	JUN 1 6 2003 By 1 37 R 5 2 34
			\$600 \$2,400 \$6,000 \$12,000	Term Plan           \$0.0720           \$0.0700           \$0.0680           \$0.0660	JUN 1 6 2003 By 1 37 R 5 2 34
			\$600 \$2,400 \$6,000 \$12,000 \$30,000	Term Plan           \$0.0720           \$0.0700           \$0.0680           \$0.0660           \$0.0630	JUN 1 6 2003 By 1 37 R 5 2 34
			\$600 \$2,400 \$6,000 \$12,000 \$30,000 \$60,000	Term Plan           \$0.0720           \$0.0700           \$0.0680           \$0.0660           \$0.0630           \$0.0600	JUN 1 6 2003 By 1 37 R 5 2 34

Issued: March 31, 2003

Effective: April 30, 2003

Missouri Public Service Commission

FILED APR 80 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> te V

Southwestern Bell Communications Services, Inc.PSC Mo. - No. 13rd Revised Sheet 234.8d/b/a SBC Long DistanceReplacing 2nd Revised Sheet 234.8

#### SECTION 4 - RATES AND CHARGES

#### 4.7 Custom Business Services (continued)

- 4.7.27 High Volume Calling II Plus (continued)
  - (A) Outbound Calls (continued)
    - .2 High Volume Dedicated Outbound Calling II Plus (continued)

	Per Minute Rate	Per Minute Rate	N	Per Minute Rate	N
MAC	1 Year Term Plan	2 Year Term Plan		3 Year Term Plan	
\$600	\$0.0580	\$0.0580		\$0.0580	
\$2,400	\$0.0580	\$0.0580		\$0.0570	
\$6,000	\$0.0580	\$0.0570		\$0.0560	
\$9,000	\$0.0580	\$0.0570		\$0.0560	
\$12,000	\$0.0570	\$0.0560		\$0.0550	
\$18,000	\$0.0570	\$0.0560		\$0.0550	
\$24,000	\$0.0570	\$0.0560		\$0.0550	
\$30,000	\$0.0560	\$0.0550		\$0.0540	
\$42,000	\$0.0560	\$0.0550		\$0.0540	
\$60,000	\$0.0550	\$0.0540		\$0.0530	
\$90,000	\$0.0550	\$0.0540		\$0.0530	
\$120,000	\$0.0540	\$0.0530		\$0.0520	
\$180,000	\$0.0530	\$0.0520		\$0.0510	
\$240,000	\$0.0520	\$0.0510	Ν	\$0.0500	N

The per minute usage rates for IntraLATA calls are as follows.

Issued: March 12, 2004

Effective: April 12, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 2nd Revised Sheet 234.8 d/b/a SBC Long Distance 234.8 Replacing 1st Revised Sheet 234.8

#### SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services (continued)
  - 4.7.27 High Volume Calling II Plus (continued)
    - (A) Outbound Calls (continued)
      - .2 High Volume Dedicated Outbound Calling II Plus (continued)

The per minute usage rates for IntraLATA calls are as follows.

	Per Minute Rate		
MAC	l Year Term Plan		
\$600	\$0.0580		
\$2,400	\$0.0580		
\$6,000	\$0.0580		
\$9,000	\$0.0580		
\$12,000	\$0.0570 \$0.0570 \$0.0570		
\$18,000			
\$24,000			
\$30,000	\$0.0560		
\$42,000	\$0.0560		
\$60,000	\$0.0550		
\$90,000	\$0.0550		
\$120,000	\$0.0540		
\$180,000	\$0.0530		
\$240,000	\$0.0520		

## CANCELLED

Ν

N N

Ν

Ν

APR,1 2 2004 3<sup>rd</sup> RS 234.8 Public Service Commission MISSOURI

Issued: August 15, 2003

غم

Effective: September 15, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FLED SEP 15 2003

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

### Missouri Public

REC'D AUG 1 5 2003

Service Commission



#### Missouri Public Sorviee Commission

## RECT) JUN 05 2003

R

R

÷

 Southwestern Bell Communications Services, Inc.
 PSC Mo. - No. 1
 1st Revised Sheet 234.8

 d/b/a SBC Long Distance
 Replacing Original Sheet 234.8

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.27 High Volume Calling II Plus (continued)

- (A) Outbound Calls (continued)
  - .2 High Volume Dedicated Outbound Calling II Plus (continued)

The per minute usage rates for IntraLATA calls are as follows.

	Per Minute Rate		
MAC	1 Year		
	Term Plan		
\$600	\$0.0580		
\$2,400	\$0.0580		
\$6,000	\$0.0580		
\$12,000	\$0.0570		
\$30,000	\$0.0560		
\$60,000	\$0.0550		
\$120,000	\$0.0540		
\$180,000	\$0.0530		
\$240,000	\$0.0520		

Effective: June 16, 2003

លាស

CANCELLED

SEP, 1 5 2003

MISS

U٧.

Public Service

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Misseuri Public Service Semmission

Commission

FILED JUN 1 6 2003

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Issued: June 5, 2003



Issued: March 31, 2003

Effective: April 30, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588



FILFA APR 80 2003

Southwestern Bell Communications Services, Inc.PSC Mo. - No. 11st Revised Sheet 234.9d/b/a SBC Long DistanceReplacing Original Sheet 234.9

#### SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services (continued)
  - 4.7.27 High Volume Calling II Plus (continued)
    - (B) Inbound Toll Free Calls
      - .1 High Volume Toll Free Calling II Plus Usage Rates
        - .a Without CMR

The per minute usage rates are the same as Section 4.7.27 (A).1 of this Tariff.

.b With CMR

The rate is \$.016 per minute which applies in addition to the R per minute usage rates in Section 4.7.27 (A).1 of this Tariff.

Issued: March 10, 2004

Effective: April 2, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

	western Bell Commu SBC Long Distance	inications	s Services, Inc. PSC Mo No. 1	Original Sheet 234.9	
<u></u>		SEC	TION 4 - RATES AND CHARGES	Missourt Public Sorvice Commission	9 <b>n</b>
4.7	Custom Business	REED MAR <b>31</b> 201	03		
	4.7.27 High Vol	ume Cal	ling II Plus (continued)		
	(B) Int	ound To	oll Free Calls		
	.1	High	Volume Toll Free Calling II Plus - Usa	age Rates	
		.a	Without CMR		
			The per minute usage rates are the sate (A).1 of this Tariff.	me as Section 4.7.27	   
		.b	With CMR		 
			The rate is \$.03 per minute which ap minute usage rates in Section 4.7.27		   N



APR 0 2 2004 By 1<sup>S+</sup> RS 234.9 Public Service Commission MISSOURI

Issued: March 31, 2003

Effective: April 30, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missourt Public Service Commission

FILED APR 80 2003

÷

Southwestern Bell Communications Services, Inc.PSC Mo. - No. 11st Revised Sheet 234.10d/b/a SBC Long DistanceReplacing Original Sheet 234.10

#### SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services (continued)
  - 4.7.27 High Volume Calling II Plus (continued)
    - (B) Inbound Toll Free Calls (continued)
      - .2 High Volume Dedicated Toll Free Calling II Plus Usage Rates
        - .a Without CMR

The per minute usage rates are the same as Section 4.7.27 (A).2 of this Tariff.

.b With CMR

The rate is \$.016 per minute which applies in addition to the R per minute usage rates in Section 4.7.27 (A).2 of this Tariff.

.3 Optional Feature Charges

The description and rates for available optional features may be found in the Company's interstate Voice Product Reference and Pricing Guide which may be found at www.sbc.com.

Issued: March 10, 2004

Effective: April 2, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

	western E <u>SBC Lon</u>			ications	Services, Inc.	. PSC Mo No. 1	Original Sheet 234.10	
				SEC	TION 4 - RA	TES AND CHARGES	Missourt Public Sorvico Commission	
4.7								
	4.7.27	High	Volu	me Cal	ling II Plus (c	continued)	RECT) MAR 31 2003	
		(B)	Inbo	ound To	ll Free Calls	(continued)		
			.2	High	Volume Ded	icated Toll Free Calling	II Plus - Usage Rates	
				.a	Without Cl	MR		
					The per mi (A).2 of thi	nute usage rates are the : is Tariff.	same as Section 4.7.27	
				.b	With CMR		1	
						\$.03 per minute which a ge rates in Section 4.7.2	applies in addition to the per   7 (A).2 of this Tariff.	
			.3	Optio	onal Feature (	Charges		
				in the	e Company's	nd rates for available opt interstate Voice Product be found at www.sbc.co	Ç I	

\_

## CANCELLED

APR 0 2 2004 St RS 234.10 Public Service Commission MISSOURI

Issued: March 31, 2003

. . . . . .

Effective: April 30, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

## FILED APR 30 2003

 Southwestern Bell Communications Services, Inc.
 PSC Mo. - No. 1

 <u>d/b/a SBC Long Distance</u>
 R

1st Revised Sheet 234.11 Replaces Original Page No. 234.11

D

#### SECTION 4 - RATES AND CHARGES

Issued: June 10, 2003

Effective: July 10, 2003

D

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

• ` • Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.27 High Volume Calling II Plus (continued)

(C) Sign On Bonus

Customers that subscribe to High Volume Outbound Calling II Plus and/or High Volume Dedicated Outbound Calling II Plus under a term plan agreement for the provision of their IntraLATA or their IntraLATA and InterLATA long distance intrastate calling and sign up for or already subscribe to CompleteLink service from an Affiliated LEC or Affiliated CLEC will receive a one-time bonus to be fulfilled to the Customer in the form of a check within ninety (90) days of the provisioning of their High Volume Calling II Plus. The amount of the bonus is shown in the following chart:

MAC	Term	Bonus Amo	ount
\$600	1 Year	\$6.00	
\$2,400	l Year	\$24.00	CANCELLED
\$6,000	1 Year	\$60.00	JUL 1 0 2003
\$12,000	1 Year	\$120.00	Public Service Commission MISSOURI
\$30,000 - \$240,000	1 Year	\$300.00	 N

Issued: March 31, 2003

Effective: April 30, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

### Missouri Public Service Commission

FILED APR 30 2003

Original Sheet 234.11



Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 Original Sheet 234.12 d/b/a SBC Long Distance

#### **SECTION 4 - RATES AND CHARGES**

4.7 Custom Business Services (continued)

÷

---- --

4.7.28 Reserved for future use

Issued: April 15, 2003

Effective: May 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Ν Ν Southwestern Bell Communications Services, Inc.PSC Mo. - No. 13rd Revised Sheet 234.13d/b/a SBC Long DistanceReplacing 2nd Revised Sheet 234.13

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.29 Business Domestic Saver 15 Plus 1 Year<sup>1</sup>

The usage rate for outbound calls and TFS calls is \$0.07 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

Issued: March 1, 2004

Effective: April 1, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, In <u>d/b/a SBC Long Distance</u>			PSC Mo No. 1	2nd Revised Sheet 234.13 Replacing 1st Revised Sheet 234.13
		SECTION 4 - R.	ATES AND CHAR	IGES Missouri Public
4.7	Custon	n Business Services (continued)		RECTO JAN 2 8 2004
	4.7.29	Business Domestic Saver 15 Plus 1 Year <sup>1</sup>		C Service Commission
		The usage rate for outbound calls and TFS calls is \$0.07 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.15 per minute. The per call charge may be found in Section		

4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

## CANCELLED

APR 0 1 2004 By 3' RS 334.13 Public Service Commission MISSOURI

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

N N

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missoufi Public Service Commission

FILED FEB 28 2004

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a\_SBC Long Distance

1st Revised Sheet 234.13 Replacing Original Sheet 234.13

Missouri Public Service Commission

RECT MAY 16 2003

#### **SECTION 4 - RATES AND CHARGES**

4.7 Custom Business Services (continued)

4.7.29 Business Domestic Saver 15 Plus 1 Year

The usage rate for outbound calls and TFS calls is \$0.07 per minute. The usage rate N for fully automated, operator assisted, and operator dialed calls billed to the Calling ľ Card - Option 2 is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff. Ν

# CANCELLED

amission

Issued: May 16, 2003

Effective: June 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Missouri Public Servico Commission

FILED JUN 1 6 2003





Т

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u>

SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.29 Reserved for future use

CANCELLED

Issued: April 15, 2003

Effective: May 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

#### Original Sheet 234.13

Missouri Public Service Commission RECD APR 15 2003





#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.30 Business Long Distance 50 Plus 1 Year<sup>1</sup>

The usage rate for outbound calls and TFS calls is \$0.07 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.15 per minute. The per call charge may C/R be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

Issued: March 1, 2004

Effective: April 1, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u> SECTION 4 - RATES AND CHARGES Missouri Public

4.7 Custom Business Services (continued)

REC'D JAN 282004

4.7.30 Business Long Distance 50 Plus 1 Year<sup>1</sup> Service Commission<sup>C</sup>

The usage rate for outbound calls and TFS calls is \$0.07 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

## CANCELLED

APR 0 1 2004 By 3<sup>nd</sup> RS 234.14 Public Service Commission MISSOURI

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

N N

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FILED FEB 28 2004

	SBC Long	SECTION 4 - RATES AND CHARGES	Replacing Original Sheet 234.14
		SECTION 4 - KATES AND CHARGES	Missouri Public Service Commissien
4.7	Custom	Business Services (continued)	REG'D MAY 16 2003
	4.7.30	Business Long Distance 50 Plus 1 Year	T
		The usage rate for outbound calls and TFS calls is \$0.0	7 per minute. The usage rate N
		for fully automated, operator assisted, and operator diale	ed calls billed to the Calling
		Card - Option 2 is \$0.15 per minute. The per call charge	e may be found in Section
		4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) o	f this Tariff. N

# CANCELLED

FEB 2 8 2004 By 2nd RS 234.14 Public Service Commission MISSOURI

Issued: May 16, 2003

. . .

:

Effective: June 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

17

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Public Service Commission

FILED JUN 1 6 2003

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

Original Sheet 234.14

#### **SECTION 4 - RATES AND CHARGES**

- 4.7 Custom Business Services (continued)
  - 4.7.30 Reserved for future use

JUN 1 8 2003 pub

Issued: April 15, 2003

Effective: May 16, 2003

Tawnya Rechtin. Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Missourt Public Service Commission



RFM APR 152003 Ν

Missouri Public Service Commission




#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.31 Business Long Distance 100 Plus 1 Year<sup>1</sup>

The usage rate for outbound calls and TFS calls is \$0.07 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.14 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

C/R

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

Issued: March 1, 2004

Effective: April 1, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

	western Bel SBC Long I		PSC Mo No. 1	2nd Revised Sheet 234.15 Replacing 1st Revised Sheet 234.15
		SECTION 4 - RA	ATES AND CHAP	IGES Missouri Public
4.7	Custom	Business Services (continued)		RECD JAN 2 8 2004
	4.7.31	Business Long Distance 100 Pl	us 1 Year <sup>1</sup>	Service Commission
		•		\$0.07 per minute. The usage rate or dialed calls billed to the Calling

4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Card - Option 2 is \$0.14 per minute. The per call charge may be found in Section

CANCELLED

APR 0 1 2004 By 3<sup>rd</sup> RS 234.15 Public Service Commission MISSOURI

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations	Ν
effective April 1, 2004.	Ν

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Serviso Cemmission

FILED FEB 2 8 2004

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

#### SECTION 4 - RATES AND CHARGES

Replacing Original Sheet 234.15 Missouri Public Servico Commination

1st Revised Sheet 234.15

4.7 Custom Business Services (continued)

# REED MAY 16 2003

4.7.31 Business Long Distance 100 Plus I Year

The usage rate for outbound calls and TFS calls is \$0.07 per minute. The usage rate R for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.14 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

# CANCELLED

FEB 2 8 2004 201 RS 23415 Public

Issued: May 16, 2003

Effective: June 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FILED JUN 1 6 2003



Original Sheet 234.15

#### SECTION 4 - RATES AND CHARGES

## Missouri Public Service Commission

4.7 Custom Business Services (continued)

RFCD APR 15 2003

N

4.7.31 Business Long Distance 100 Plus I Year

The usage rate for outbound calls and TFS calls is \$0.098 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.14 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.



Issued: April 15, 2003

Effective: May 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Public Service Commission

FILED MAY 16 2003

Southwestern Bell Communications Services, Inc.	PSC Mo No. 1
d/b/a SBC Long Distance	

i

#### SECTION 4 - RATES AND CHARGES

4.7	Custom Business Services (continued)	N
		1
	4.7.32 Reserved For Future Use	N

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

. .....

•

.

#### Original Sheet 234.17

#### SECTION 4 - RATES AND CHARGES

4.7	Custom Business Services (continued)	Ν
	4.7.33 Reserved For Future Use	N

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

•

.

#### Original Sheet 234.18

i

#### SECTION 4 - RATES AND CHARGES

4.7	Custom Business Services (continued)	Ν
		l
	4.7.34 Reserved For Future Use	Ν

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.35 Business Domestic Saver 15 Connections 1 Plus Service 1 Year<sup>1</sup>

The usage rate for outbound calls and TFS calls is \$0.0690 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

Issued: March 1, 2004

Effective: April 1, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 2nd Revised Sheet 234.19 <u>d/b/a SBC Long Distance</u> <u>SECTION 4 - RATES AND CHARGES</u> <u>SECTION 5 - RATES AND CHARGES</u>

Custom Business Services (continued)

4.7

**RECD JAN 2 8 2004** 

4.7.35 Business Domestic Saver 15 Connections 1 Plus Service 1 Service Commission

The usage rate for outbound calls and TFS calls is \$0.0690 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

## CANCELLED

APR 0 1 2004 By 3rd RS 234.19 Public Service Commission MISSOURI

<sup>1</sup> This service is no longer available to new Customers or existing Customers	at new locations N
effective April 1, 2004.	Ν

Issued: January 28, 2004

Effective: February 28, 2004

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Servico Commicción

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED FEB 28 2004

Missouri Public Service Commission

# REG'D JUN 05 2003

R

1st Revised Sheet 234.19

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance Replacing Original Sheet 234.19

#### **SECTION 4 - RATES AND CHARGES**

4.7 Custom Business Services (continued)

4.7.35 Business Domestic Saver 15 Connections 1 Plus Service 1 Year

The usage rate for outbound calls and TFS calls is \$0.0690 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

---

## CANCELLED

FEB 2 8 2004 1R3234.19 Public Service Commission MISSOURI

Issued: June 5, 2003

Effective: June 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

FILED JUN 1 6 2003

Original Sheet 234.19

	SECTION 4 - RATES AND CHARGES	Missouri Public Service Commission	
Custon	Business Services (continued)	IN IN	1
4.7.35	Business Domestic Saver 15 Connections 1 Plus Service	REGD APR 18 2003	
	The usage rate for outbound calls and TFS calls is \$0.09 rate for fully automated, operator assisted, and operator of Calling Card - Option 2 is \$0.15 per minute. The per cal Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2	70 per minute. The usage   lialed calls billed to the   l charge may be found in	_

CANCELLED JUN 1 6 2003 PUD

Issued: April 18, 2003

4.7

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Public Sorvieo Commiasion ł

FILED MAY 19 2003

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.36 Business Long Distance 50 Connections 1 Plus Service 1 Year<sup>1</sup>

The usage rate for outbound calls and TFS calls is \$0.0690 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

Issued: March 1, 2004

Effective: April 1, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

	SECTION 4 - RATES AND CHARGES Missouri Public
4.7	Custom Business Services (continued) RECD JAN 282004
	4.7.36 Business Long Distance 50 Connections 1 Plus Service 1 Year <sup>1</sup> C Service Commission
	The usage rate for outbound calls and TFS calls is \$0.0690 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.15 per minute. The per call charge may be found in
	Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Southwestern Bell Communications Services, Inc. PSC Mo. - No. I

d/b/a SBC Long Distance

# CANCELLED

2nd Revised Sheet 234.20

Replacing 1st Revised Sheet 234.20

APR 0 1 2004 By 3rd RS 234.20 Public Service Commission MISSOURI

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations	N
effective April 1, 2004.	Ν

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED FEB 28 2004

# Misseufi Public Service Commission

# RECT JUN 05 2003

R

1st Revised Sheet 234.20

Replacing Original Sheet 234.20

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.36 Business Long Distance 50 Connections 1 Plus Service 1 Year

The usage rate for outbound calls and TFS calls is \$0.0690 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

CANCELLED

FEB 2 8 2004 Amission

Issued: June 5, 2003

Effective: June 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Missouri Publie Service Commission

FILED JUN 1 6 2003

Original Sheet 234.20

#### SECTION 4 - RATES AND CHARGES

Missouri Public Service Commission RECD APR 18 2003N

Ν

4.7 Custom Business Services (continued)

4.7.36 Business Long Distance 50 Connections 1 Plus Service 1 Year

The usage rate for outbound calls and TFS calls is \$0.0970 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.



Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Public Servico Commission

FILED MAY 19 2003

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.37 Business Long Distance 100 Connections 1 Plus Service 1 Year<sup>1</sup>

The usage rate for outbound calls and TFS calls is \$0.0690 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.14 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

C/R

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

Issued: March 1, 2004

Effective: April 1, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

 Southwestern Bell Communications Services, Inc.
 PSC Mo. - No. 1
 2nd Revised Sheet 234.21

 <u>d/b/a SBC Long Distance</u>
 Replacing 1st Revised Sheet 234.21

#### SECTION 4 - RATES AND CHARGES

Missouri Public

4.7 Custom Business Services (continued)

4.7.37 Business Long Distance 100 Connections 1 Plus Service 1 Year D JAN 2 8 2004

Service Commission The usage rate for outbound calls and TFS calls is \$0.0690 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.14 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

## CANCELLED

APR, 0 1 2004 RS 234.21 Public Service Commission MISSOURI

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations	Ν
effective April 1, 2004.	N

Issued: January 28, 2004

Effective: February 28, 2004

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

## FILED FEB 28 2004

### Missouri Public Service Commission RFCD JUN 05 2003

1st Revised Sheet 234.21 Replacing Original Sheet 234.21

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.37 Business Long Distance 100 Connections 1 Plus Service 1 Year

The usage rate for outbound calls and TFS calls is \$0.0690 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.14 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

R

CANCELLED FEB 2 8 2004 By 2 RS 234 21 By 2 RS 234 21 By 2 RS 234 21 RS 2004 MISSOURI

Issued: June 5, 2003

Effective: June 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Missouri Publie Service Commission

FILED JUN 1 6 2003

Original Sheet 234.21

SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

Missouri Public Servise Commission REGD APR 18 2003

Ν

4.7.37 Business Long Distance 100 Connections 1 Plus Service 1 Year

The usage rate for outbound calls and TFS calls is \$0.0960 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.14 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

# CANCELLED

JUN 1 6 2003 By (SHRS 234.21 Public Service Commission MISSING

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Public Service Commicalon

FILED MAY 19 2003

#### Original Sheet 234.22

#### **SECTION 4 - RATES AND CHARGES**

4.7	Custom Business Services (continued)	Ν
	4.7.38 Reserved for future use	N

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

#### **SECTION 4 - RATES AND CHARGES**

4.7	Custom Business Services (continued)	N
	4.7.39 Reserved for future use	 N

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

- ----

#### SECTION 4 - RATES AND CHARGES

4.7	Custom Business Services (continued)	N
	4.7.40 Reserved for future use	 N

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

----

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.41 Business Domestic Saver 15 Connections 2 Plus Service 1 Year<sup>1</sup>

The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

C/R

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

Issued: March 1, 2004

Effective: April 1, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

		ell Communications Services, Inc. PSC Mo No. 1 Distance Replacing 1st Revised Sheet 234.25	
		SECTION 4 - RATES AND CHARGES Missouri Public	
4.7	Custon	n Business Services (continued)	
	4.7.41	RECD JAN 2 8 2004 Business Domestic Saver 15 Connections 2 Plus Service 1 Year C	
		Service Commission	
		The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage	
		rate for fully automated, operator assisted, and operator dialed calls billed to the	
		Calling Card - Option 2 is \$0.15 per minute. The per call charge may be found in	

Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

# CANCELLED

APR 0 1 2004 By 3<sup>rd</sup> RS 234.25 Public Service Commission MISSOURI

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations	Ν
effective April 1, 2004.	Ν

Issued: January 28, 2004

Effective: February 28, 2004

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missoufi Public Servico Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED FEB 28 2004

## Missouri Public Service Commission RECO JUN 05 2003

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

#### o. 1 Ist Revised Sheet 234.25 Replacing Original Sheet 234.25

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.41 Business Domestic Saver 15 Connections 2 Plus Service 1 Year

The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate R for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

CANCELLED

282004 2 minission

Issued: June 5, 2003

Effective: June 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Missouri Publie Servise Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED JUN 1 6 2003

Original Sheet 234.25

REGIDAPR 18 2003  $_{
m N}$ 

Ν

# SECTION 4 - RATES AND CHARGES Service Commission

4.7 Custom Business Services (continued)

4.7.41 Business Domestic Saver 15 Connections 2 Plus Service 1 Year

The usage rate for outbound calls and TFS calls is \$0.0960 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

# CANCELLED

JUN 1 6 2003 6ap<sub>1</sub>

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Public Service Commission



#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year<sup>1</sup>

The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

Issued: March 1, 2004

Effective: April 1, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 2nd Revised Sheet 234.26 <u>d/b/a SBC Long Distance</u> \_\_\_\_\_\_ Replacing 1st Revised Sheet 234.26

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

Missouri Public

4.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year RECD JAN 2 8 2004

Service Commission The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

## CANCELLED

APR 0 1 2004 By 3<sup>rd</sup> RS 234.26 Public Bervice Commission MISSOURI

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations N effective April 1, 2004.

Issued: January 28, 2004

Effective: February 28, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Public Service Commission



# REC'D JUN 05 2083

1st Revised Sheet 234.26

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

#### Replacing Original Sheet 234.26 SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year

The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate R for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

# CANCELLED

FEB 2 8 2004 By 200 RS 234.24 Public Service Commission MISSOURI

Issued: June 5, 2003

Effective: June 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Missouri Public Service Commission

FILED JUN 1 6 2003

Original Sheet 234.26

Sorviee Commission

RECT APR 18 2003

N

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year

The usage rate for outbound calls and TFS calls is \$0.0960 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

# CANCELLED

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Public Service Commission



JUN 16 20034.24 Service Constriction



#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year<sup>1</sup>

The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.14 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

C/R

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

Issued: March 1, 2004

Effective: April 1, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc.	PSC Mo No. 1	2nd Revised Sheet 234,27
d/b/a SBC Long Distance	······································	Replacing 1st Revised Sheet 234,27
SECTION 4 - RA	ATES AND CHAR	GES Missouri Public

4.7 Custom Business Services (continued)

REC'D JAN 2 8 2004

4.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year<sup>1</sup> C Service Commission

The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.14 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

# CANCELLED

APR 0 1 2004 31 3rd RS 234.27 -unlic burvice Commission MISSOURI

<sup>1</sup> This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

N N

Issued: January 28, 2004

Effective: February 28, 2004

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED FEB 28 2004

Missouri Public Sorvico Commission

# **RECT JUN 05 2003**

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u> 1st Revised Sheet 234.27 Replacing Original Sheet 234.27

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year

The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate R for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.14 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

# CANCELLED

FEB 2 8 2004

Issued: June 5, 2003

Effective: June 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Missouri Publie Service Commission

FILED JUN 1 6 2003

Original Sheet 234.27

Missourl Public Service Commication

**RECD** APR 18 2003 $_{\rm N}$ 

Ν

#### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year

The usage rate for outbound calls and TFS calls is \$0.0950 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.14 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

# CANCELLED

JUN 1 6 2003

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Public Service Commission

FILED MAY 19 2003

#### **SECTION 4 - RATES AND CHARGES**

4.7	Custom Business Services (continued)	Ν
	4.7.44 Reserved for future use	 N

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

#### Original Sheet 234.29

#### **SECTION 4 - RATES AND CHARGES**

4.7	Custom Business Services (continued)	
	4.7.45 Reserved for future use	 N

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588
Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u>

#### Original Sheet 234.30

## SECTION 4 - RATES AND CHARGES

4.7	Custom Business Services (continued)	Ν
	4.7.46 Reserved for future use	 N

. \_

Issued: April 18, 2003

Effective: May 19, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

4.7 Custom Business Services (continued)

## 4.7.47 Value Plans

(A) Business Domestic Value Saver 15

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0590
2 Year Term Plan N	\$0.0580 N

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Issued: April 12, 2004

Effective: May 13, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 1st Revised Sheet 234.31 d/b/a SBC Long Distance Replacing Original Sheet 234.31

**SECTION 4 - RATES AND CHARGES** 

4.7 Custom Business Services (continued)

4.7.47 Value Plans

**Business Domestic Value Saver 15** (A)

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0590

For fully automated, operator assisted, and operator dialed calls billed to Т the Calling Card - Option 2, category 11 the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

## CANCELLED

MISSOURI

Issued: March 1, 2004

Effective: April 1, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Missouri Public Service Commission

FILED APR 01 2004



## Miégouri Public

REC'D MAR 01 2004

Service Commission

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

SECTION 4 - RATES AND CHARGES

Original Sheet 234.31

4.7 Custom Business Services (continued)

REC'D MAY 1 6 2003

Missouri Public Servico Commission

- 4.7.47 Value Plans
  - (A) Business Domestic Value Saver 15

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0590

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

## CANCELLED

APR 0 1 2004 B, 1<sup>3f</sup> RS 234.31 Public Service Commission MISSOURI

Issued: May 16, 2003

Effective: June 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public Servico Commission

N

FILED JUN 1 6 2003

- 4.7 Custom Business Services (continued)
  - 4.7.47 Value Plans (continued)
    - (B) Business Long Distance Value 50

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment		Rate Per Minute	
1 Year Term Plan		\$0.0580	
2 Year Term Plan	Ν	\$0.0570	Ν

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Issued: April 2, 2004

Effective: May 3, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc.	PSC Mo No. 1	1st Revised Sheet 234.32
<u>d/b/a SBC Long Distance</u>		Replacing Original Sheet 234.32

4.7 Custom Business Services (continued)

4.7.47 Value Plans (continued)

**(B) Business Long Distance Value 50** 

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0580

For fully automated, operator assisted, and operator dialed calls billed to Τ the Calling Card - Option 2, category 11 the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

## CANCELLED

Issued: March 1, 2004

Effective: April 1, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Missouri Public Service Commission

FILED APR 01 2004

Missouri Public

REC'D MAR 01 2004

Service Commission

Southwestern Bell Communications Services. Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services (continued)
  - 4.7.47 Value Plans (continued)
    - (B) Business Long Distance Value 50

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Rate Per Minute
1 Year Term Plan	\$0.0580

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.15 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

## CANCELLED

## APR 0 1 2004 *J<sup>SF</sup> RS 234.32* Public Survice Commission MISSOURI

Issued: May 16, 2003

Effective: June 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missourl Public Service Commission

FILED JUN 1 6 2003

Missouri Public Service Commission

**REC'D MAY 16 200B** 

Ν

i

Original Sheet 234.32

ng Distance

### 4.7 Custom Business Services (continued)

## 4.7.47 Value Plans

(C) Business Long Distance Value 100

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment		Rate Per Minute	
1 Year Term Plan		\$0.0570	
2 Year Term Plan	Ν	\$0.0560	Ν

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 the usage rate is \$0.14 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Issued: April 2, 2004

Effective: May 3, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

	western Bell Comm SBC Long Distance	unications Services, Inc.	PSC Mo No. 1	Original	Sheet 234.33
		SECTION 4 - RAT	ES AND CHARGES	Miècouri	Public
4.7	Custom Business	s Services (continued)		REC'D MAR	<b>01</b> 2004
	4.7.47 Value Pl	ans	s	ervice Cor	
	(C) <b>B</b>	isiness Long Distance V			
	Tł	ne per minute usage rate	s for outbound and TF	S calls are as for	llows:
		Customer Commitme	nt Rate Per Mir	nute	
		l Year Term Plan	\$0.0570		

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 the usage rate is \$0.14 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

CANCELLED MAY 0 3 2004 Public S ominission Souri

Issued: March 1, 2004

Effective: April 1, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

Ń

I

FILED APR 01 2004

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance 1st Revised Sheet 234.34 Replacing Original Sheet 234.34

### SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.48 Business Unlimited Long Distance Plans

**Business Unlimited Long Distance Plans** 

The per minute usage rate for switched TFS is as follows:

	Rate Per Minute
Switched TFS	\$0.0700

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.1400 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

The MRC for unlimited interstate and intrastate 1+ outbound calling is as follows:

Number of Access Line Subscribed to Business Unlimited Long Distance Plans	MRC
1	\$20
2	\$40
3	\$60
4	\$80
5	\$100
6	\$120
7	\$140
8	\$160
9	\$180
10	\$200

Issued: September 12, 2003

Effective: October 13, 2003

N N N N N

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

## SECTION 4 - RATES AND CHARGES

- 4.7 Custom Business Services (continued)
  - 4.7.48 Business Unlimited Long Distance Plans

Business Unlimited Long Distance Plans

The per minute usage rate for switched TFS is as follows:

	Rate Per Minute
Switched TFS	\$0.0700

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.1400 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

The MRC for unlimited interstate and intrastate 1+ outbound calling is as follows:

Number of Access Line Subscribed to Business Unlimited Long Distance Plans	MRC
1	\$20
2	\$40
3	\$60
4	\$80
5	<b>\$</b> 100

CANCELLED OCT 1 3 2003 UN 15t RS 234.34 Public Service Commission Missouri

Issued: June 12, 2003

Effective: July 17, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

## FILED JUL 17 2003



RECT JUN 12 2003

Original Sheet 234.34

- 4.7 Custom Business Services (continued)
  - 4.7.49 Business Long Distance Solutions
    - (A) Business Long Distance Solutions 15

The per minute usage rates for outbound and switched TFS calls and for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 category 11 are listed in the table below. The per call charges may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Call Type	1-Year Term	2-Year Term	Ν
Outbound and Switched TFS	\$0.0700	\$0.0700	
Calling Card - Option 11	\$0.1500	\$0.1500	Ν

Issued: April 2, 2004

Effective: May 3, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

d/b/a	SBC Long Dista	nce		······································
		SECTION 4 - RATES A	AND CHARGES	Missouri Public
4.7	Custom Busi	ness Services (continued)	E	ECD FEB 23 2004
	4.7.49 Busir	tess Long Distance Solutions	۴	
		-	Se	rvice Commissior
	(A)	Business Long Distance Solut	ions 15	
		The per minute usage rates for fully automated, operator assi Calling Card - Option 2 categ call charges may be found in Section 4.1.2 (C) of this Tarif	sted, and operator ory 11 are listed ir Section 4.1.1 (B).2 f.	dialed calls billed to the the table below. The per
		Call Type	1-Year Term	
		Outbound and Switched TFS	\$0.0700	
	l l	Calling Card - Option 11	\$0.1500	

1

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1

# CANCELLED

---

Original Sheet 234.35

MAY 0 3 2004 by 1 Stars 234.35 Public Service Commission MISSOURI

Issued: February 23, 2004

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 ° Missouri Public Servise Commission

N

FILED APR 01 2004

- 4.7 Custom Business Services (continued)
  - 4.7.49 Business Long Distance Solutions (continued)
    - (B) Business Long Distance Solutions 50

The per minute usage rates for outbound and switched TFS calls and for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 category 11 are listed in the table below. The per call charges may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Call Type	1-Year Term	2-Year Term	Ν
Outbound and Switched TFS	\$0.0700	\$0.0700	
Calling Card - Option 11	\$0.1500	\$0.1500	Ν

Issued: April 2, 2004

Effective: May 3, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 Original Sheet 234.36 <u>d/b/a SBC Long Distance</u> SECTION 4 - RATES AND CHARGES **Missouri Public** 4.7 Custom Business Services (continued) 4.7.49 Business Long Distance Solutions (continued) (B) Business Long Distance Solutions 50 The per minute usage rates for outbound and switched TFS calls and for fully automated, operator assisted, and operator dialed calls billed to the

Calling Card - Option 2 category 11 are listed in the table below. The per call charges may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Call Type	1-Year Term
Outbound and Switched TFS	\$0.0700
Calling Card - Option 11	\$0.1500

# CANCELLED



Issued: February 23, 2004

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILÉD APR 01 2004

Missouri Public Service Commission

Ν

- 4.7 Custom Business Services (continued)
  - 4.7.49 Business Long Distance Solutions (continued)
    - (C) Business Long Distance Solutions 100

The per minute usage rates for outbound and switched TFS calls and for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 category 11 are listed in the table below. The per call charges may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Call Type	1-Year Term	2-Year Term	Ν
Outbound and Switched TFS	\$0.0700	\$0.0700	
Calling Card - Option 11	\$0.1400	\$0.1400	Ν

Issued: April 2, 2004

Effective: May 3, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

	western Bell Con SBC Long Dista	mmunications Services, Inc. PSC Mo No. 1 nce	Original Sheet 234.37
		SECTION 4 - RATES AND CHARC	JES
4.7	Custom Busi	ness Services (continued)	Missouri Public
	4.7.49 Busin	ness Long Distance Solutions (continued)	REC'D FEB 23 2004
	(C)	Business Long Distance Solutions 100	Service Commission
		The per minute usage rates for outbound ar fully automated, operator assisted, and ope Calling Card - Option 2 category 11 are lis call charges may be found in Section 4.1.1 Section 4.1.2 (C) of this Tariff.	rator dialed calls billed to the steel in the table below. The per

Call Type	1-Year Term
Outbound and Switched TFS	\$0.0700
Calling Card - Option 11	\$0.1400

# CANCELLED



Issued: February 23, 2004

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588



CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

# FILED APR 01 2004

4.7	Custon	n Business Services (continued)	Ν
	4.7.50	Business Block of Time 5000	
		The MRC is \$750.00 per BTN for a 5000 minute block of time as described in	
		Section 3.7.50 of this Tariff. For Direct-Dialed outbound one-plus (1+), Toll Free	
		Service, and Calling Card – Option 2, Category 11, the rate is \$0.15 per minute	
		for calls completed after the 5000 minute block of time has been used.	
		The per call charge for fully automated, operator assisted, and operator dialed	
		calling cards billed to the Calling Card – Option 2, category 11 may be found in	
		Section 4.1.1 (B) .2 .a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.	
			Ν

Issued: December 10, 2004

Effective: January 10, 2005

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

SBC Long Distance, Inc.

PSC Mo. - No. 1

#### 4th Revised Sheet 235 Replacing 3rd Revised Sheet 235

### **SECTION 4 - RATES AND CHARGES**

### 4.8 Custom Consumer Services

4.8.1 Block of Time: 300 Minutes

The monthly recurring charge is \$18.00 per BTN for a 300 minute block of time for (1+) Direct-Dialed intrastate and interstate calling. The rate is \$0.09 per I minute for all (1+) Direct-Dialed outbound intrastate calls completed after the 300 minute block of time has been used. For fully automated, operator assisted, and operator dialed calling card calls billed to the Calling Card - Option 1, the rate is \$0.09 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, I Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

- 4.8.2 Reserved for future use
- 4.8.3 Reserved for future use
- 4.8.4 Reserved for future use

Issued: January 26, 2005

Effective: February 12, 2005

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

.



### 4.8 Custom Consumer Services

a

4.8.1 Block of Time: 300 Minutes

The monthly recurring charge is \$18.00 per BTN for a 300 minute block of time for (1+) Direct-Dialed intrastate and interstate calling. The rate is \$0.07 per minute for all I (1+) Direct-Dialed outbound intrastate calls completed after the 300 minute block of time has been used. For fully automated, operator assisted, and operator dialed calling card calls billed to the Calling Card - Option 1, the rate is \$0.07 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 I (C) of this Tariff.

- 4.8.2 Reserved for future use
- 4.8.3 Reserved for future use
- 4.8.4 Reserved for future use

Issued: June 18, 2003

Effective: July 3, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. d/b/a SBC Long Distance PSC Mo. - No. 1 2nd Revised Sheet 235 Replacing 1st Revised Sheet 235

SECTION 4 - RATES AND CHARGES

#### 4.8 Custom Consumer Services

## REGE JAN 07 2003

Т

Ν

N

T/D

Missouri Public Bervice Cemmission

4.8.1 Block of Time: 300 Minutes

The monthly recurring charge is \$18.00 per BTN for a 300 minute block of time for (1+) Direct-Dialed intrastate and interstate calling. The rate is \$0.06 per minute for all (1+) Direct-Dialed outbound intrastate calls completed after the 300 minute block of time has been used. For fully automated, operator assisted, and operator dialed calling card calls billed to the Calling Card -Option 1, the rate is \$0.06 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

CANCELLED

- 4.8.2 Reserved for future use
- 4.8.3 Reserved for future use
- 4.8.4 Reserved for future use

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

## Missouri Public Service Commission

FILED FEB 072003

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

1st Revised Sheet 235 **Replacing Original Sheet 235** 

## Miasouri Public

## **SECTION 4 - RATES AND CHARGES**

**Custom Consumer Services** 4.8

RECD DEC 07 2001 Service Commission

4.8.1 Block of Time: 300 Minutes

> The monthly recurring charge is \$18.00 per BTN for a 300 minute block of time for (1+) Direct-Dialed intrastate and interstate calling. The rate is \$0.06 per minute for С all (1+) Direct-Dialed outbound intrastate calls completed after the 300 minute block С of time has been used. For calling card calls billed to the Proprietary Calling Card -Option 1, the rate is \$0.06 per minute. С

4.8.2 Reserved for future use

Reserved for future use 4.8.3

Reserved for future use 4.8.4

CANCELLED

**MaSIOA** 

Issued: December 7, 2001

Effective: January 6, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED JAN 06 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Service Commission

2

## Missouri Public

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance RECD MAR 0 7 2001 Original Sheet 235 Service Commission

## SECTION 4 - RATES AND CHARGES

4.8 Custom Consumer Services

4.8.1 Block of Time: 300 Minutes

The monthly recurring charge is \$18.00 per BTN for a 300 minute block of time for intrastate and interstate calling. The rate is \$0.06 per minute for all outbound intrastate calls completed after the 300 minute block of time has been used. For fully automated calling card calls billed to the Proprietary Calling Card - Option 1 after the 300 minute block of time has been used the rate is \$0.06 per minute.

4.8.2 Reserved for future use

- 4.8.3 Reserved for future use
- 4.8.4 Reserved for future use



JAN 06 2002 incion

Issued: March 7, 2001

Effective April 23 v2091

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588 DEC 0 7 2001 Missouri Public

FILED DEC 072001 01 = 475 Service Commission

## RECT MAR 0 7 2001

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance Original Sheet 236 Service Commission

## **SECTION 4 - RATES AND CHARGES**

Reserved for future use

Issued: March 7, 2001

Effective April 23, 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588 DEC 0 7 2001 Missouri Public

FILED DEC 0 7 2001 **D** 1 - 4 7 5 Service Commission

#### SECTION 5 - MISCELLANEOUS CHARGES

## REC'D AUG 01 2002

#### 5.1 Return Check Charge

## Service Commission

When another telecommunications carrier provides the billing function on behalf of the Company, the other carrier's bad check charge applies. Otherwise, the Company will assess the Customer a return check charge of \$25.00 for any check that is returned for any reason by the financial institution on which it is drawn.

Μ

M - Material moved to Original Sheet 237.1

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Regulatory Director 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

FILED SEP 01 2002

Service Commission

## **SECTION 5 - MISCELLANEOUS CHARGES**

RECTD DEC 07 2001

#### 5.1 **Return Check Charge**

## Service Commission

ommission

When another telecommunications carrier provides the billing function on behalf of the Company, the other carrier's bad check charge applies. Otherwise, the Company will assess the Customer a return check charge of \$25.00 for any check that is returned the bar of by the financial institution on which it is drawn.

#### 5.2 Additional Labor Charges

dditional Labor Charges	SEP El 2nd 4	012002 523 m Come
	Rate Per Fifteen Minutes	GOURI
8:00 am to but not including 5:00 pm Monday through Friday excluding holidays	\$25.00	
Holidays (New Years Day, Federally Observed Memorial Day, Independence Day, Labor Day, Thanksgiving, and Christmas)	\$31.25	
All Other Times	\$31.25	

The Additional Labor Charges shown above apply for all Services which are provided by the Company as stand alone intrastate Services when the Customer subscribes to one of the Company's outbound Service offerings for intrastate IntraLATA calling and selects another company for the provision of the Customer's intrastate InterLATA calling. When intrastate Service is offered by the Company as an add-on to one of the Company's interstate service offerings (i.e. Switched Services), the Additional Labor Charges apply pursuant to Company's Voice Product Reference and Pricing Guide which may be found at www.sbc.com/PublicAffairs/LDByState/.

.

Т Т

Issued: December 7, 2001

Effective: January 6, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILFD .IAN 06 2002

Service Commission

## Missouri Public

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

Original Sheet 237 Service Commission

**RECD MAR 07 2001** 

## SECTION 5 - MISCELLANEOUS CHARGES

### 5.1 Return Check Charge

When another telecommunications carrier provides the billing function on behalf of the Company, the other carrier's bad check charge applies. Otherwise, the Company will assess the Customer a return check charge of \$25.00 for any check that is returned for any reason by the financial institution on which it is drawn.

### 5.2 Additional Labor Charges

	Rate Per Fifteen Minutes
8:00 am to but not including 5:00 pm Monday through Friday excluding holidays	\$25.00
Holidays (New Years Day, Federally Observed Memorial Day, Independence Day, Labor Day, Thanksgiving, and Christmas)	\$31.25
All Other Times	\$31.25

The Additional Labor Charges shown above apply for all Services which are provided by the Company as stand alone intrastate Services when the Customer subscribes to one of the Company's outbound Service offerings for intrastate IntraLATA calling and selects another company for the provision of the Customer's intrastate InterLATA calling. When intrastate Service is offered by the Company as an add-on to one of the Company's interstate service offerings (i.e. Switched Services), the Additional Labor Charges apply pursuant to Company's interstate tariff.



Issued: March 7, 2001

Effective: April 23, 2001

DEC 07 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED DEC 072001 01-475 Service Commission

## SECTION 5 - MISCELLANEOUS CHARGES

### 5.2 Additional Labor Charges

	Rate Per Fifteen Minutes
8:00 am to but not including 5:00 pm Monday through Friday excluding holidays	\$25.00
Holidays (New Years Day, Federally Observed Memorial Day, Independence Day, Labor Day, Thanksgiving, and Christmas)	\$31.25
All Other Times	\$31.25

The Additional Labor Charges shown above apply for all Services which are provided by the Company as stand alone intrastate Services when the Customer subscribes to one of the Company's outbound Service offerings for intrastate IntraLATA calling and selects another company for the provision of the Customer's intrastate InterLATA calling. When intrastate Service is offered by the Company as an add-on to one of the Company's interstate service offerings (i.e. Switched Services), the Additional Labor Charges apply pursuant to Company's Voice Product Reference and Pricing Guide which may be found at www.sbc.com.

D

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Regulatory Director 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance Ist Revised Sheet 237.1 Replacing Alegence Sheet 237.1

## SECTION 5 - MISCELLANEOUS CHARGES

### 5.2 Additional Labor Charges

Service Commission

REC'D SEP 2 7 2002

	Rate Per Fifteen Minutes
8:00 am to but not including 5:00 pm Monday through Friday excluding holidays	\$25.00
Holidays (New Years Day, Federally Observed Memorial Day, Independence Day, Labor Day, Thanksgiving, and Christmas)	\$31.25
All Other Times	\$31.25

The Additional Labor Charges shown above apply for all Services which are provided by the Company as stand alone intrastate Services when the Customer subscribes to one of the Company's outbound Service offerings for intrastate IntraLATA calling and selects another company for the provision of the Customer's intrastate InterLATA calling. When intrastate Service is offered by the Company as an add-on to one of the Company's interstate service offerings (i.e. Switched Services), the Additional Labor Charges apply pursuant to Company's Voice Product Reference and Pricing Guide which may be found at www.sbc.com/public\_affairs/.

Т

CANCELLED

Issued: September 27, 2002

Effective: October 30, 2002

Norm Descoteaux, Associate Regulatory Director 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED OCT 3 0 2002

Service Commission

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance Original Sheet 237.1 ~ Miczouri Publio

## SECTION 5 - MISCELLANEOUS CHARGES

#### 5.2 Additional Labor Charges

Service Commission

REC'D AUG 01 2002

	Rate Per Fifteen Minutes
8:00 am to but not including 5:00 pm Monday through Friday excluding holidays	\$25.00
Holidays (New Years Day, Federally Observed Memorial Day, Independence Day, Labor Day, Thanksgiving, and Christmas)	\$31.25
All Other Times	\$31.25

The Additional Labor Charges shown above apply for all Services which are provided by the Company as stand alone intrastate Services when the Customer subscribes to one of the Company's outbound Service offerings for intrastate IntraLATA calling and selects another company for the provision of the Customer's intrastate InterLATA calling. When intrastate Service is offered by the Company as an add-on to one of the Company's interstate service offerings (i.e. Switched Services), the Additional Labor Charges apply pursuant to Company's Voice Product Reference and Pricing Guide which may be found at www.sbc.com/PublicAffairs/LDByState/.

Μ

Μ

## M - Material moved from 1st Revised Sheet 237

# CANCELLED



Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Regulatory Director 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Service Commission

## SECTION 5 - MISCELLANEOUS CHARGES

## 5.3 Order Expedite Charge

Customers may request a change in the requested Service due date for pending Service Orders. When the Company accepts a request to expedite an order, the Company does not promise to deliver on the desired due date in advance of the normal service order interval. The Company will use its best effort to meet the desired due date. A onetime charge applies when the Customer requests a Service due date sooner than the standard interval due date, and Service is provided sooner than the standard interval due date. An Order Expedite Charge applies when a change of requested Service due date is the only Customer requested change to the original or supplemental Service Order. Any expedite charges incurred for the provisioning of local access are not included in this Order Expedite Charge and will be passed through to the Customer. The Order Expedite Charge is as follows:

	Non-Recurring
Outbound Service Provided Exclusively for	\$300
IntraLATA Calling Per Order	

The Order Expedite Charge shown above applies for all Services which are provided by the Company as stand alone intrastate Services or when the Customer subscribes to one of the Company's outbound Service offerings for intrastate IntraLATA calling and selects another company for the provision of the Customer's intrastate InterLATA calling. When intrastate Service is offered by the Company as an add-on to one of the Company's interstate service offerings (i.e. Switched Services), the Order Expedite Charge applies pursuant to Company's Voice Product Reference and Pricing Guide which may be found at www.sbc.com.

 $\mathbf{D}$ 

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance 2nd Revised Sheet 238 Replacing 1st Revised Sheet 238

# SECTION 5 - MISCELLANEOUS CHARGES Missouri Public

### 5.3 Order Expedite Charge

REC'D SEP 272002

Customers may request a change in the requested Service due date for pending Service Orders. When the Company accepts a request to expedite an order, the Company does not promise to deliver on the desired due date in advance of the normal service order interval. The Company will use its best effort to meet the desired due date. A one-time charge applies when the Customer requests a Service due date sooner than the standard interval due date, and Service is provided sooner than the standard interval due date. An Order Expedite Charge applies when a change of requested Service due date is the only Customer requested change to the original or supplemental Service Order. Any expedite charges incurred for the provisioning of local access are not included in this Order Expedite Charge and will be passed through to the Customer. The Order Expedite Charge is as follows:

	Non-Recurring
Outbound Service Provided Exclusively for	\$300
IntraLATA Calling Per Order	

The Order Expedite Charge shown above applies for all Services which are provided by the Company as stand alone intrastate Services or when the Customer subscribes to one of the Company's outbound Service offerings for intrastate IntraLATA calling and selects another company for the provision of the Customer's intrastate InterLATA calling. When intrastate Service is offered by the Company as an add-on to one of the Company's interstate service offerings (i.e. Switched Services), the Order Expedite Charge applies pursuant to Company's Voice Product Reference and Pricing Guide which may be found at www.sbc.com/public\_affairs/.



Issued: September 27, 2002

Effective: October 30, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

Т

FILED OCT 3 0 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

## Service Commission

1st Revised Sheet 238 Replacing Original Sheet 238

## Missourl Public

#### SECTION 5 - MISCELLANEOUS CHARGES

REC'D DEC 07 2001

### 5.3 Order Expedite Charge

## Service Commission

Customers may request a change in the requested Service due date for pending Service Orders. When the Company accepts a request to expedite an order, the Company does not promise to deliver on the desired due date in advance of the normal service order interval. The Company will use its best effort to meet the desired due date. A one-time charge applies when the Customer requests a Service due date sooner than the standard interval due date, and Service is provided sooner than the standard interval due date. An Order Expedite Charge applies when a change of requested Service due date is the only Customer requested change to the original or supplemental Service Order. Any expedite charges incurred for the provisioning of local access are not included in this Order Expedite Charge and will be passed through to the Customer. The Order Expedite Charge is as follows:

	Non-Recurring
Outbound Service Provided Exclusively for	\$300
IntraLATA Calling Per Order	

The Order Expedite Charge shown above applies for all Services which are provided by the Company as stand alone intrastate Services or when the Customer subscribes to one of the Company's outbound Service offerings for intrastate IntraLATA calling and selects another company for the provision of the Customer's intrastate InterLATA calling. When intrastate Service is offered by the Company as an add-on to one of the Company's interstate service offerings (i.e. Switched Services), the Order Expedite Charge applies pursuant to Company's Voice Product Reference and Pricing Guide which may be found at www.sbc.com/PublicAffairs/LDByState/.

0 2002 nssion

Issued: December 7, 2001

Effective: January 6, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

Т

Т

FILED JAN 06 2002

Service Commission

## Missouri Public

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance RECD MAR 0 7 2001 Original Sheet 238 Service Commission

#### SECTION 5 - MISCELLANEOUS CHARGES

## 5.3 Order Expedite Charge

Customers may request a change in the requested Service due date for pending Service Orders. When the Company accepts a request to expedite an order, the Company does not promise to deliver on the desired due date in advance of the normal service order interval. The Company will use its best effort to meet the desired due date. A one-time charge applies when the Customer requests a Service due date sooner than the standard interval due date, and Service is provided sooner than the standard interval due date. An Order Expedite Charge applies when a change of requested Service due date is the only Customer requested change to the original or supplemental Service Order. Any expedite charges incurred for the provisioning of local access are not included in this Order Expedite Charge and will be passed through to the Customer. The Order Expedite Charge is as follows:

	Non-Recurring
Outbound Service Provided Exclusively for	\$300
IntraLATA Calling Per Order	

The Order Expedite Charge shown above applies for all Services which are provided by the Company as stand alone intrastate Services or when the Customer subscribes to one of the Company's outbound Service offerings for intrastate IntraLATA calling and selects another company for the provision of the Customer's intrastate InterLATA calling. When intrastate Service is offered by the Company as an add-on to one of the Company's interstate service offerings (i.e. Switched Services), the Order Expedite Charge applies pursuant to Company's interstate Tariff.



Issued: March 7, 2001

## Effective April 23, 2004

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

. . . . . .

DEC 072001

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

## Missouri Public

FILED DEC 072001 01 - 475 Service Commission SBC Long Distance, Inc.

PSC Mo. - No. 1

5th Revised Sheet 239 Replacing 4th Revised Sheet 239

## SECTION 5 - MISCELLANEOUS CHARGES

## 5.4 Payphone Origination Charge

Pursuant to the FCC's Order in CC Docket 96-128, this charge applies only to dial-around calls, i.e., calls originating using a carrier's access code, a Customer's 800/877/888 (and other area code assignments as appropriate) number and other toll free numbers and debit card calls, from payphone instruments. The Customer shall pay the Company a per call charge of \$0.60 per call for all such traffic.

Ι

Issued: February 25, 2005

Effective: March 12, 2005

Janet Vader, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588



## SECTION 5 - MISCELLANEOUS CHARGES

## 5.4 Payphone Origination Charge

Pursuant to the FCC's Order in CC Docket 96-128, this charge applies only to dial-around calls, i.e., calls originating using a carrier's access code, a Customer's 800/877/888 (and other area code assignments as appropriate) number and other toll free numbers and debit card calls, from payphone instruments. The Customer shall pay the Company a per call charge of \$0.47 per call for all such traffic.

Ι

Issued: November 30, 2004

Effective: December 12, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588
Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

3rd Revised Sheet 239 Replacing 2nd Revised Sheet 239 Public

### **SECTION 5 - MISCELLANEOUS CHARGES**

5.4 Payphone Origination Charge

RECT MAR 1 0 2004

Pursuant to the FCC's Order in CC Docket 96-128, this charge applies only to dial-around calls, i.e., calls originating using a carrier's access code, a Customer's 800/877/888 (and other area code assignments as appropriate) number and other toll free numbers and debit card calls, from payphone instruments. The Customer shall pay the Company a per call charge of \$0.43 per call for all such traffic. I

## CANCELLED

DEC 1 0 2004 13239 Public Service Commission MISSOURI

Issued: March 10, 2004

Effective: April 8, 2004

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Missouri Public Service Commission

FILED APR 08 2004

Southwestern Bell Communications Services, Inc. d/b/a SBC Long Distance PSC Mo. - No. 1 2nd Revised Sheet 239 Replacing 1st Revised Sheet 239

## Missouri Public SECTION 5 - MISCELLANEOUS CHARGES Sorvice Commission

5.4 Payphone Origination Charge

## RECT JAN 07 2003

D D

Pursuant to the FCC's Order in CC Docket 96-128, this charge applies only to dialaround calls, i.e., calls originating using a carrier's access code, a Customer's 800/877/888 (and other area code assignments as appropriate) number and other toll free numbers and debit card calls, from payphone instruments.

The Customer shall pay the Company a per call charge of \$0.25 per call for all such traffic.

## CANCELLED

APR. 0'8 2004 By 3rd RS 339 Public Service Commission MISSOURI

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Publie Service Commission

FLED FEB 0 7 2003



1st Revised Sheet 239 Replacing Original Sheet 239

### Micsouri Public

### SECTION 5 - MISCELLANEOUS CHARGES

### REC'D AUG 01 2002

5.4 Payphone Origination Charge

### Service Commission

Pursuant to the FCC's Order in CC Docket 96-128, this charge applies only to dial-around calls, i.e., calls originating using a carrier's access code, a Customer's 800/877/888 (and other area code assignments as appropriate) number and other toll free numbers and debit card calls, from payphone instruments. This charge does not apply for 0+ call for which the payphone provider would otherwise receive compensation. The Customer shall pay the Company a per call charge of \$0.25 per call for all such traffic.



M - Material moved to Original Page 239.1

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

## Missouri Public

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance RECD MAR 07 2001 Service Commission

#### SECTION 5 - MISCELLANEOUS CHARGES

### 5.4 Payphone Origination Charge

Pursuant to the FCC's Order in CC Docket 96-128, this charge applies only to dial-around calls, i.e., calls originating using a carrier's access code, a Customer's 800/877/888 (and other area code assignments as appropriate) number and other toll free numbers and debit card calls, from payphone instruments. This charge does not apply for 0+ call for which the payphone provider would otherwise receive compensation. The Customer shall pay the Company a per call charge of \$0.25 per call for all such traffic.

### 5.5 PIC Change Rebate

If local telephone company is requested to change the subscriber's PIC from one long distance service provider to another long distance service provider, the local telephone company may charge the Customer for the PIC change. If a Customer incurs such a charge from its local telephone company for changing the PIC to the Company, the Company will rebate that charge to the Customer. The rebate will be in the form of a credit on the Customer's bill. The credit will appear within two (2) billing cycles after the Customer provides the Company proof that the local telephone company billed the Customer for the PIC change.

CANCELLED

SEP 01 2002 By 154RS 239 Public Service Scient

Issued: March 7, 2001

Effective: April 23, 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588 DEC 0 7 2001

Missouri Public

FILED DEC\_077201

Service Commission

M - Material moved to Original Sheet 239.1.2 M\* - Material moved from Original Sheet 239

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Miscouri Public

M\*

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

### SECTION 5 - MISCELLANEOUS CHARGES

## RECTD AUG 01 2002

#### 5.6 Multiple Bill Copies

## Service Commission

5.6.1 General

Customers that are direct-billed by the Company or an authorized billing agent may receive additional paper bill copies at the charges specified in Section 5.6.2 or Section 5.6.3 of this Tariff. Customers must receive a fully-itemized monthly billing statement in order to subscribe to Multiple Bill Copies.

5.6.2 Customer Commits to MAC

The charge per additional paper bill copy varies based on the Customer's MAC and whether the request is at the Child BAN or Invoice Point BAN and are as follows:

	MAC \$30,000 or below	MAC greater than \$30,000
Child BAN	\$20	ICB
Invoice Point BAN	\$40	ICB

#### 5.6.3 All Other Direct-Billed Customers

The charge is \$40 per copy for each additional bill copy.

M - Material moved from 1st Revised Sheet 239.1

Issued: August 1, 2002

Effective: September 1, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Miceouri Public

М

FILED SEP 01 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

### 6.1 General

From time-to-time, the Company may offer special promotions to its Customers waiving certain charges, offering Service(s) at special rates, and/or offering promotional discounts. Promotional discounts include but are not limited to reduced monthly rates or charges for an existing Service, incentive subscription bonuses, free Service periods, full or partial waivers of installation charges or optional feature charges or any combination thereof. Terms and conditions of promotions may be limited to certain dates, times, market segments, and/or locations. The Company may engage in national and/or intrastate special promotional offerings or trial Service offerings designed to attract new Customers, retain existing Customers, win back former Customers, or stimulate Customer usage. The terms of national promotional offerings are set forth in the applicable interstate Voice Product Reference and Pricing Guide which may be found at www.sbc.com governing such programs. To the extent these programs may extend to intrastate Services, the terms of these national offerings are incorporated by reference. Promotional offerings are subject to prior approval of the Commission. The Company will provide written notice to the Commission no less than seven (7) days prior to the beginning of each promotion period. The Company will offer all promotions in a nondiscriminatory manner.

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

### REC'D SEP 2 7 2002

### 6.1 General

### Service Commission

From time-to-time, the Company may offer special promotions to its Customers waiving certain charges, offering Service(s) at special rates, and/or offering promotional discounts. Promotional discounts include but are not limited to reduced monthly rates or charges for an existing Service, incentive subscription bonuses, free Service periods, full or partial waivers of installation charges or optional feature charges or any combination thereof. Terms and conditions of promotions may be limited to certain dates, times, market segments, and/or locations. The Company may engage in national and/or intrastate special promotional offerings or trial Service offerings designed to attract new Customers, retain existing Customers, win back former Customers, or stimulate Customer usage. The terms of national promotional offerings are set forth in the applicable interstate Voice Product Reference and Pricing Guide which may be found at www.sbc.com/public affairs/ governing such programs. To the extent these programs may extend to intrastate Services, the terms of these national offerings are incorporated by reference. Promotional offerings are subject to prior approval of the Commission. The Company will provide written notice to the Commission no less than seven (7) days prior to the beginning of each promotion period. The Company will offer all promotions in a nondiscriminatory manner.

## CANCELLED

FEB 07 2003 Broches 240 Pablic Service Costumetion

Issued: September 27, 2002

Effective: October 30, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

;

### Missouri Public

Т

FILED OCT 3 0 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

1st Revised Sheet 240 Replacing Original Sheet 240 Missouri Public

### **SECTION 6 - PROMOTIONS**

### RECD DEC 07 2001

#### 6.1 General

### Service Commission

T T

From time-to-time, the Company may offer special promotions to its Customers waiving certain charges, offering Service(s) at special rates, and/or offering promotional discounts. Promotional discounts include but are not limited to reduced monthly rates or charges for an existing Service, incentive subscription bonuses, free Service periods, full or partial waivers of installation charges or optional feature charges or any combination thereof. Terms and conditions of promotions may be limited to certain dates, times, market segments, and/or locations. The Company may engage in national and/or intrastate special promotional offerings or trial Service offerings designed to attract new Customers, retain existing Customers, win back former Customers, or stimulate Customer usage. The terms of national promotional offerings are set forth in the applicable interstate Voice Product Reference and Pricing Guide which may be found at www.sbc.com/PublicAffairs/LDByState/ governing such programs. To the extent these programs may extend to intrastate Services, the terms of these national offerings are incorporated by reference. Promotional offerings are subject to prior approval of the Commission. The Company will provide written notice to the Commission no less than seven (7) days prior to the beginning of each promotion period. The Company will offer all promotions in a nondiscriminatory manner.

## CANCELLED

OCT 3 0 2002 ្រប់ប្រើរំដែ ម

Issued: December 7, 2001

Effective: January 6, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

### Missouri Public

FILED JAN 06 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

## Missouri Public

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance RECD MAR 0 7 2001 Original Sheet 240 Service Commission

### **SECTION 6 - PROMOTIONS**

#### 6.1 General

From time-to-time, the Company may offer special promotions to its Customers waiving certain charges, offering Service(s) at special rates, and/or offering promotional discounts. Promotional discounts include but are not limited to reduced monthly rates or charges for an existing Service, incentive subscription bonuses, free Service periods, full or partial waivers of installation charges or optional feature charges or any combination thereof. Terms and conditions of promotions may be limited to certain dates, times, market segments, and/or locations. The Company may engage in national and/or intrastate special promotional offerings or trial Service offerings designed to attract new Customers, retain existing Customers, win back former Customers, or stimulate Customer usage. The terms of national promotional offerings are set forth in the applicable interstate tariffs governing such programs. To the extent these programs may extend to intrastate Services, the terms of these national offerings are incorporated by reference. Promotional offerings are subject to prior approval of the Commission. The Company will provide written notice to the Commission no less than seven (7) days prior to the beginning of each promotion period. The Company will offer all promotions in a nondiscriminatory manner.

## CANCELLED

JAN 06 2002 By KHRS Q40 Public Service Commission MISSOURI

Issued: March 7, 2001

Effective: April 23, 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588 DEC 0 7 2001 Missouri Public

FILED DEC 072001 01-475 Service Commission

### Missouri Public Original Sheet 241 RECD DEC-07 2001

### **SECTION 6 - PROMOTIONS**

### Service Commission

Ν

N

6.2 Business Total Solutions Plus 200 Promotion #49

The sign-up period for Business Total Solutions Plus 200 Promotion #49 is December 14, 2001 through February 22, 2002. Orders for new Service must be activated by March 25, 2002. The promotion is available to Business Customers that (1) subscribe to local/vertical package from Southwestern Bell Telephone Company during the sign-up period; (2) subscribe to one of the Company's optional calling plans for the provision of outbound service; and (3) request to participate in this promotional offering. Business Customers participating in this promotion will be given up to 100 free minutes of use per month for two (2) consecutive months. Qualified minutes of use include interstate and intrastate outbound (1+) Direct-Dialed minutes of use and Toll Free Calling. All working telephone numbers under the billed telephone number will accumulate towards the free minutes. If the Customer fails to use the 100 free minutes in a given monthly billing period, no credit is carried forward to the next monthly billing period. This promotion cannot be combined with any other domestic promotional offering. If the Customer switches or cancels its optional calling plan before the end of the first full bill cycle, the Customer will only receive the portion of the free minutes that have been used up to the date the optional calling plan is cancelled or changed. If the Customer cancels Service before the first full bill cycle starts, no free minutes will be given to the Customer on the Customer's final invoice.

Issued: December 7, 2001

Effective: December 14, 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public

FILED DEC 1 4 2001

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

### **Missouri Public**

Original Sheet 242

RECTD DEC 07 2001

### **SECTION 6 - PROMOTIONS**

### 6.3 Reserved for future use

ï

Service Commission

Ν

Issued: December 7, 2001

Effective: December 14, 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

4

Missouri Public

FILED DEC 1 4 2001

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

# RECD DEC 072001

### **SECTION 6 - PROMOTIONS**

### Service Commission

N

- 6.4 500 BOT/Online Billing 30 Min. Free Promotion #62
  - 6.4.1 The sign-up period for 500 BOT/Online Billing 30 Min. Free Promotion #62 is December 14, 2001 through January 29, 2002. Orders for new Service must be activated by March 1, 2002. Residential Customers or Applicants in the State who subscribe to Long Distance Block of Time 500 Minutes and sign up for online billing during the sign-up period will be automatically enrolled in 500 BOT/Online Billing 30 Min. Free Promotion #62.
  - 6.4.2 Residential Customers or Applicants enrolled in this promotion will be given thirty (30) free minutes of use as a sign-up bonus. For each BAN, new Applicants will not be billed for the first thirty (30) qualified interstate and intrastate MOU in their first full bill cycle after subscribing to Long Distance Block of Time 500 Minutes. Qualified MOU include outbound (1+) Direct-Dialed MOU from presubscribed lines. For existing Customers subscribing to Long Distance Block of Time 500 Minutes in the middle of a billing cycle, the promotion begins on the day the order is processed. The Customer may receive up to thirty (30) free minutes for the partial billing month. Up to thirty (30) free minutes are applied to the next full billing cycle. All WTNs under the BAN will accumulate towards the free minutes. If the Customer fails to use the thirty (30) free minutes in a given monthly billing period, no credit is carried forward to the next monthly billing period. This promotion cannot be combined with any other domestic promotional offering.
  - 6.4.3 If the Customer switches or cancels its optional calling plan before the end of the first full bill cycle, the Customer will only receive the portion of the free minutes that have been used up to the date the optional calling plan is cancelled or changed.
  - 6.4.4 If the Customer cancels Service before the first full bill cycle starts, no free minutes will be given to the Customer on the Customer's final invoice.

Ν

Issued: December 7, 2001

Effective: December 14, 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

FILED DEC 1 4 2001

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Southwestern Bell Communications Services, Inc.	PSC Mo No. 1	Original Sheet 244
d/b/a SBC Long Distance		RECD DEC 07 2001

### Service Commission

Missouri Public

Ν

- 6.5 Domestic Saver/Online Billing 30 Min. Free Promotion #63
  - 6.5.1 The sign-up period for Domestic Saver/Online Billing 30 Min. Free Promotion #63 is December 14, 2001 through January 29, 2002. Orders for new Service must be activated by March 1, 2002. Residential Customers or Applicants in the State who subscribe to Domestic Saver and sign up for online billing during the sign-up period will be automatically enrolled in Domestic Saver/Online Billing 30 Min. Free Promotion #63.
  - 6.5.2 Residential Customers or Applicants enrolled in this promotion will be given thirty (30) free minutes of use as a sign-up bonus. For each BAN, new Applicants will not be billed for the first thirty (30) qualified interstate and intrastate MOU in their first full bill cycle after subscribing to Long Distance Block of Time 500 Minutes. Qualified MOU include outbound (1+) Direct-Dialed MOU from presubscribed lines. For existing Customers subscribing to Long Distance Block of Time 500 Minutes in the middle of a billing cycle, the promotion begins on the day the order is processed. The Customer may receive up to thirty (30) free minutes for the partial billing month. Up to thirty (30) free minutes are applied to the next full billing cycle. All WTNs under the BAN will accumulate towards the free minutes. If the Customer fails to use the thirty (30) free minutes in a given monthly billing period, no credit is carried forward to the next monthly billing period. This promotion cannot be combined with any other domestic promotional offering.
  - 6.5.3 If the Customer switches or cancels its optional calling plan before the end of the first full bill cycle, the Customer will only receive the portion of the free minutes that have been used up to the date the optional calling plan is cancelled or changed.

Issued: December 7, 2001

Effective: December 14, 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public

Ν

FILED DEC 1 4 2001

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance Original Sheet 245 RECD DEC 07 2001

### SECTION 6 - PROMOTIONS

### Service Commission

6.6 Winback Domestic Saver MRC Waiver Promotion #67

ï

- 6.6.1 The sign up period for Winback Domestic Saver MRC Waiver Promotion #67 is December 14, 2001 through December 29, 2001. Orders for new Service must be activated by February 1, 2002. This promotion is available to Residential Customers in the State that (1) are currently subscribing to Domestic Saver and advise the Company they wish to cancel Service or (2) previously subscribed to Domestic Saver and cancelled Service. This promotion cannot be combined with any other domestic promotional offering.
- 6.6.2 For the first three full billing cycles following the Customer participating in Promotion #67, Customers will receive a credit equal to the monthly recurring charge shown in Section 4.4.3 (G) of this Tariff.

N

N

1

Ì.

}

Issued: December 7, 2001

Effective: December 14, 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

44

Missouri Public

FILED DEC 1 4 2001

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

### Missouri Public

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

RECD DEC 0722001

Service Commission

### SECTION 6 - PROMOTIONS

### 6.7 30 Free Domestic Saver Promotion #71

- 6.7.1 The sign-up period for 30 Free Domestic Saver Promotion #71 is December 14, 2001 through January 29, 2002. Orders for new Service must be activated by March 1, 2002. Residential Customers or Applicants in the State who subscribe to Domestic Saver will be automatically enrolled in 30 Free Domestic Saver Promotion #71.
- 6.7.2 Residential Customers or Applicants enrolled in this promotion will be given thirty (30) free minutes of use as a sign-up bonus. For each BAN, new Applicants will not be billed for the first thirty (30) qualified interstate and intrastate MOU in their first full bill cycle after subscribing to Domestic Saver. Qualified MOU include outbound (1+) Direct-Dialed MOU from presubscribed lines. For Customers subscribing to Domestic Saver in the middle of a billing cycle, the promotion begins on the day the order is processed. The Customer may receive up to thirty (30) free minutes for the partial billing month. Up to thirty (30) free minutes are applied to the next full billing cycle. All WTNs under the BAN will accumulate towards the free minutes. If the Customer fails to use the thirty (30) free minutes in a given monthly billing period, no credit is carried forward to the next monthly billing period. This promotion cannot be combined with any other domestic promotional offering.
- 6.7.3 If the Customer switches or cancels its optional calling plan before the end of the first full bill cycle, the Customer will only receive the portion of the free minutes that have been used up to the date the optional calling plan is cancelled or changed.

Issued: December 7, 2001

Effective: December 14, 2001

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

.

### Missouri Public

FILED DEC 1 4 2001

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Service Commission

Ν

Ν

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance 1st Revised Sheet 247 Replacing Original Sheet 247 Wissouri Public

### SECTION 6 - PROMOTIONS

### REC'D APR 2 9 2002

Service Commission

6.8 Acquisition Coupon Domestic Promotion #84

The sign up period for Promotion #84 is May 6, 2002 through August 4, 2002. New N Applicants in the State that subscribe to Block of Time: 300 Minutes, Long Distance Block | of Time 500 Minutes or 500 Block of Time Gold during the sign up period will be mailed a | coupon that may be redeemed for a check equal to one month's MRC of the optional calling | plan selected. The Customer must return the coupon to the Company by October 4, 2002 to | redeem the check. The check will be mailed to the Customer within eight (8) weeks of the | Company's receipt of the coupon. The Customer has until December 4, 2002 to cash the | check. Checks cashed after December 4, 2002 will be returned by the bank. This promotion | cannot be combined with any other domestic promotional offering.

Ν

### Missouri Public

## FILED MAY 06 2002

## Service Commission

Issued: April 29, 2002

Effective: May 6, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

\_\_\_\_\_

Original Sheet 247

### SECTION 6 - PROMOTIONS

RECD APR 1 6 2002

Service Commission

**Missouri** Public

### 6.8 RESERVED FOR FUTURE USE

CANCELLED

MAY 0 6 2002 JSt RS 247 Public Service Commission MISSOURI

Missouri Public

FILED APR 23 2002

Cervice Commission

Issued: April 16, 2002

Effective: April 23, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149



Ν

Original Sheet 248

Missouri Public

### SECTION 6 - PROMOTIONS

## REC'D APR 1 6 2002

6.9 Acquisition 30 Minutes Domestic Saver Promotion #88

Ν

- 6.9.1 The sign up period for Promotion #88 is April 23, 2002 through July 20, 2002. Orders for new Service must be activated by August 20, 2002. This promotion is available to new or existing Residential Customers in the State that subscribe to Domestic Saver during the sign up period. This promotion cannot be combined with any other domestic promotional offering.
- 6.9.2 Customers or Applicants participating in this promotion will be given thirty (30) qualified minutes of use per month for three (3) consecutive months at no additional charge beyond the MRC (no per minute charge) as a sign-up bonus. Customers will receive a credit for the first thirty (30) qualified interstate and intrastate minutes of use in their first three (3) full bill cycles after subscribing to Domestic Saver 30 Minutes Promotion #36. Qualified minutes of use include interstate and intrastate outbound (1+) Direct-Dialed minutes of use. Calling card and operator assisted calls are not included as qualified minutes. All working telephone numbers under the billed telephone number will accumulate towards the qualified minutes. If the Customer fails to use the thirty (30) qualified minutes in a given monthly billing period, no credit is carried forward to the next monthly billing period.
- 6.9.3 If the Customer switches or cancels its optional calling plan before the end of the first full bill cycle, the Customer will only receive the portion of the qualified minutes that have been used up to the date the optional calling plan is cancelled or changed.
- 6.9.4 If the Customer cancels Service before the first full bill cycle starts, no qualified minutes will be credited to the Customer on the Customer's final invoice.

. N

Issued: April 16, 2002

Effective: April 23, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missouri Public FILED APR 2 3 2002

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance Original Sheet 248.1 Missouri Public

### SECTION 6 - PROMOTIONS REC'D MAY 2 4 2002

### 6.10 Acquisition Coupon Domestic Promotion #85

21

### Service Commission

The sign up period for Promotion #85 is June 1, 2002 through August 4, 2002. Orders for new Service must be activated by September 4, 2002. New Applicants in the State that subscribe to 200 Block of Time or 200 Block of Time Gold during the sign up period will be mailed a coupon that may be redeemed for a check equal to one month's MRC of the optional calling plan selected. The Customer must return the coupon to the Company by October 4, 2002 to redeem the check. The check will be mailed to the Customer within eight (8) weeks of the Company's receipt of the coupon. The Customer has until December 4, 2002 to cash the check. Checks cashed after December 4, 2002 will be returned by the bank. This promotion cannot be combined with any other domestic promotional offering.

Ν

ł

I

Ν

### Missouri Public

FILED JUN 01 2002

Service Commission

Issued: May 24, 2002

Effective: June 1, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

1st Revised Sheet 249 Replacing Original Sheet 249

### **SECTION 6 - PROMOTIONS**

## REC'D APR 2 3 2002

### 6.11 Yellow Page Coupon Promotion #89

Service Commission

The sign up period for Promotion #89 is May 1, 2002 through July 29, 2002. Orders for new N Service must be activated by August 12, 2002. This promotion is available to existing N Customers who subscribe to High Volume Toll Free Calling during the sign-up period. This promotion is also available to Applicants that subscribe to one of the following service options during the sign-up period for the provision of outbound and/or toll free service: Business Long Distance 200, Business Long Distance 50, Business Domestic Saver 15, Business Domestic Saver or High Volume Toll Free Calling. To qualify for this promotion, Applicants or Customers must order online and must complete a marketing survey. Customers participating in this promotion will be mailed a SBC Southwestern Bell Yellow Pages certificate that may be redeemed for yellow pages advertising with SBC Southwestern Bell.

The amount of the certificate is determined by the optional calling plan selected by the Applicant or Customer during the sign-up period:

Optional Calling Plan	Certificate Amount
Business Long Distance 200	\$250
Business Long Distance 50	\$250
Business Domestic Saver 15	\$100
Business Domestic Saver	\$100
High Volume Toll Free Calling	\$100

The SBC Southwestern Bell Yellow Page certificate will be mailed to the Customer within ten (10) business days of online completion of the marketing survey. The Customer has one (1) year from the date on the certificate to redeem the certificate.

Issued: April 23, 2002

FILED MAY 01 2002

Service Commission

Effective: May 1, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

Misserier Commission

#### **SECTION 6 - PROMOTIONS**

## **RECD** APR 1 9 2002

### 6.11 Yellow Page Coupon Promotion #89

The sign up period for Promotion #89 is May 1, 2002 through July 29, 2002. This promotion is available to existing Customers who subscribe to High Volume Toll Free Calling during the sign-up period. This promotion is also available to Applicants that subscribe to one of the following service options during the sign-up period for the provision of outbound and/or toll free service: Business Long Distance 200, Business Long Distance 50, Business Domestic Saver 15, Business Domestic Saver or High Volume Toll Free Calling. To qualify for this promotion, Applicants or Customers must order online and must complete a marketing survey. Customers participating in this promotion will be mailed a SBC Southwestern Bell Yellow Pages certificate that may be redeemed for yellow pages advertising with SBC Southwestern Bell.

The amount of the certificate is determined by the optional calling plan selected by the Applicant or Customer during the sign-up period:

Optional Calling Plan	Certificate Amount
Business Long Distance 200	\$250
Business Long Distance 50	\$250
Business Domestic Saver 15	\$100
Business Domestic Saver	\$100
High Volume Toll Free Calling	\$100

The SBC Southwestern Bell Yellow Page certificate will be mailed to the Customer within ten (10) business days of online completion of the marketing survey. The Customer has one (1) year from the date on the certificate to redeem the certificate.

Missouri Public CANCELLED FILED MAY 01 2002 MAY 01 2002 Service Commission гŝł MISSOURI Effective: May 1, 2002 Issued: April 19, 2002

Norm Descoteaux, Regulatory Manager 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Ν

Ν

6.12 Simple Solutions® Block of Time 100 Promotion #106

The sign up period for Promotion #106 is October 25, 2002 through October 29, 2002. Orders for new Service must be activated by November 4, 2002. The availability requirements for Simple Solutions Block of Time 100, as specified in Section 3.4.3 (I).1 of the Tariff, will be waived for Customers participating in this promotional offering. Simple Solutions Block of Time 100 Promotion #105 is available to Residential Customers that (1) use Switched Access to reach the long distance network and (2) meet one of the requirements specified below:

- .a new Residential Customers must subscribe to an affiliated LEC's or affiliated CLEC's CallerID Name and Number; or
- .b new Residential Customers must subscribe to cellular service as a new subscriber of cingular WIRELESS at the same time the Customer places an order to subscribe to the Company's optional calling plan, Simple Solutions Block of Time 100; or
- .c existing Residential Customers that currently subscribe to MTS, Long Distance, Long Distance II, Simple Solutions or Simple Solutions II must subscribe to cellular service as a new subscriber of cingular WIRELESS at the same time the Customer places an order to move its long distance Service from the Customer's existing optional calling plan referenced above to Simple Solutions Block of Time 100; or
- .d existing Residential Customers that advise they wish to cancel any of the Company's outbound long distance Service(s) described in Section 3.4.3 and/or Section 3.8 of this Tariff and/or MTS.

Issued: January 7, 2003

Effective: February 7, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

RECT OCT 1 8 2002

С

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 2nd Revised Sheen 250 ission <u>d/b/a SBC Long Distance</u>
Replacing 1st Revised Sheet 250

### **SECTION 6 - PROMOTIONS**

6.12 Simple Solutions Block of Time 100 Promotion #106

The sign up period for Promotion #106 is October 25, 2002 through October 29, 2002. Orders for new Service must be activated by November 4, 2002. The availability requirements for Simple Solutions Block of Time 100, as specified in Section 3.4.3 (I).1 of the Tariff, will be waived for Customers participating in this promotional offering. Simple Solutions Block of Time 100 Promotion #105 is available to Residential Customers that (1) use Switched Access to reach the long distance network and (2) meet one of the requirements specified below:

- .a new Residential Customers must subscribe to an affiliated LEC's or affiliated CLEC's CallerID Name and Number; or
- .b new Residential Customers must subscribe to cellular service as a new subscriber of Cingular WIRELESS at the same time the Customer places an order to subscribe to the Company's optional calling plan, Simple Solutions Block of Time 100; or
- .c existing Residential Customers that currently subscribe to MTS, Long Distance, Long Distance II, Simple Solutions or Simple Solutions II must subscribe to cellular service as a new subscriber of cingular WIRELESS at the same time the Customer places an order to move its long distance Service from the Customer's existing optional calling plan referenced above to Simple Solutions Block of Time 100; or
- .d existing Residential Customers that advise they wish to cancel any of the Company's outbound long distance Service(s) described in Section 3.4.3 and/or Section 3.8 of this Tariff and/or MTS.

CANCELLED ISSION

Issued: October 18, 2002

Effective: October 25, 2002

Lisa Andrejko, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED OCT 2 5 2002

Missouri Publia

Service Commission

С

Ţ

С

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u> Ist Revised Sheet 250 Replacing Original Sheet 250

### SECTION 6 - PROMOTIONS

6.12 Simple Solutions Block of Time 100 Promotion #106

RECD SEP 2 7 2002 Service Commission

The sign up period for Promotion #106 is August 2, 2002 through October 29, 2002. Orders for new Service must be activated by November 4, 2002. The availability requirements for Simple Solutions Block of Time 100, as specified in Section 3.4.3 (I).1 of the Tariff, will be waived for Customers participating in this promotional offering. Simple Solutions Block of Time 100 Promotion #105 is available to Residential Customers that (1) use Switched Access to reach the long distance network and (2) meet one of the requirements specified below:

.a new Residential Customers must subscribe to an affiliated LEC's or affiliated CLEC's CallerID Name and Number; or

.b new Residential Customers must subscribe to cellular service as a new subscriber of cingular WIRELESS at the same time the Customer places an order to subscribe to the Company's optional calling plan, Simple Solutions Block of Time 100; or

.c existing Residential Customers that currently subscribe to MTS, Long Distance, Long Distance II, Simple Solutions or Simple Solutions II must subscribe to cellular service as a new subscriber of cingular WIRELESS at the same time the Customer places an order to move its long distance Service from the Customer's existing optional calling plan referenced above to Simple Solutions Block of Time 100; or

.d existing Residential Customers that advise they wish to cancel (1) an affiliated LEC's T or affiliated CLEC's access line service or (2) any of the Company's outbound long distance Service(s) described in Section 3.4.3 and/or Section 3.8 of this Tariff and/or N MTS.



Issued: September 27, 2002

-ublic Service Commission MISSOURI E

Effective: October 4, 2002

Lisa Andrejko, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

### Missouri Publia Service Commission

Ν



		Bell Communications Services, Inc. PSC Mo No. 1	Niginal Sheet 250
		<b>SECTION 6 - PROMOTIONS</b>	REC'D JUL 26 2002
6.12	Simp	ble Solutions Block of Time 100 Promotion #106	Service Commission
	Orde requi of the Simp Cust	sign up period for Promotion #106 is August 2, 2002 throu ers for new Service must be activated by November 4, 2002 irements for Simple Solutions Block of Time 100, as speci e Tariff, will be waived for Customers participating in this ole Solutions Block of Time 100 Promotion #105 is available omers that (1) use Switched Access to reach the long distant of the requirements specified below:	2. The availability   fied in Section 3.4.3 (I).1   promotional offering.   ble to Residential
	.a	new Residential Customers must subscribe to an affilia CLEC's CallerID Name and Number; or	ited LEC's or affiliated
	.b	new Residential Customers must subscribe to cellular s of cingular WIRELESS at the same time the Customer pl to the Company's optional calling plan, Simple Solutio	aces an order to subscribe
	.c	existing Residential Customers that currently subscribe Long Distance II, Simple Solutions or Simple Solution cellular service as a new subscriber of cingular WIRELE Customer places an order to move its long distance Ser existing optional calling plan referenced above to Simp 100.	s II must subscribe to ss at the same time the rvice from the Customer's ble Solutions Block of Time
		CANCELLED	N Missouri Public
		OCT 2 5 2002	FILED AUG 02 2002
		ublic Service Commission	Service Commissi

-

Service Commission

Effective: August 2, 2002

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Issued: July 26, 2002

.

Ν

Ν

### **SECTION 6 - PROMOTIONS**

- 6.13 Domestic Saver/Domestic Saver Gold Promotion #141
  - 6.13.1 Except for subscriptions through the Internet, the sign up period for Promotion #141 is February 1, 2003 through May 1, 2003. For subscriptions through the Internet, the sign up period for Promotion #141 is February 14, 2003 through May 1, 2003. Orders for new Service must be activated by May 6, 2003. This promotion is available to new residential Applicants in the State that (1) select the Company as the Applicant's choice of long distance carrier for the provision of outbound Service and (2) newly subscribe to Domestic Saver or Domestic Saver Gold during the sign-up period. This promotion cannot be combined with any other domestic promotional offering.
  - 6.13.2 For the first full billing cycle following the Customer participating in Promotion #141, Customers subscribing to Domestic Saver will receive a one-time credit equal to the monthly recurring charge shown in Section 4.4.3 (G) of this Tariff. For the first full billing cycle following the Customer participating in Promotion #141, Customers subscribing to Domestic Saver Gold will receive a one-time credit equal to the monthly recurring charge shown in Section 4.4.3 (P) of this Tariff.

Issued: January 23, 2003

Effective: February 1, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

#### 6.14 Value Plus Flat Rate Promotion #156

The sign-up period for the Value Plus Flat Rate Promotion #156 is April 12, 2003 through June 30, 2003. Service must be activated by July 5, 2003. This promotion is available to new and existing Residential Customers in Missouri that (1) subscribe to the Value Plus Flat Rate optional calling plan during the sign-up period; (2) subscribe to the Company's interstate Value Plus Flat Rate Promotion #156; (3) maintain the requirements specified in Section 3.4.3 (Y).2.a, Section 3.4.3 (Y).2.b, and Section 3.4.3 (Y).3 of this Tariff; and (4) continue to subscribe to the Value Plus Flat Rate optional calling plan for the provision of interstate and intrastate InterLATA calling during the benefit period of this promotional offering.

The Company will waive the requirements shown in Section 3.4.3 (Y).4.a of this Tariff for Applicants and Customers participating in this promotion. For Customers participating in this promotion, the following per minute usage rates apply for the first twelve (12) months from activation date of Service.

-	interstate usage	\$0.05
-	intrastate usage - peak rate period	\$0.27
-	intrastate usage - off peak rate period	\$0.17

The peak rate period is 8:00 a.m. to but not including 5:00 p.m., Monday through Friday. The off-peak rate period is all other times. The off-peak rates apply on the following holidays: New Year's Day, Independence Day, Labor Day, Thanksgiving Day, and Christmas Day.

If the Customer fails to maintain the requirements specified in Section 3.4.3 (Y).2.a, Section 3.4.3 (Y).2.b, and Section 3.4.3 (Y).3 of this Tariff and/or fails to continue to subscribe to the Value Plus Flat Rate optional calling plan for the provision of interstate and intrastate InterLATA calling, the Customer will no longer qualify for the promotional reduced rates listed above.

The usage rate that applies after the expiration of the promotional rate may be found in Section 4.4.3 (Y).1.a of this Tariff.

Issued: March 27, 2003

Effective: April 12, 2003

Ν

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

6.15 Value Plus 60 Promotion #157

The sign-up period for the Value Plus 60 Promotion #157 is April 12, 2003 through June 30, 2003. Service must be activated by July 5, 2003. This promotion is available to new and existing Residential Customers in Missouri that (1) subscribe to the Value Plus 60 optional calling plan during the sign-up period; (2) subscribe to the Company's interstate Value Plus 60 Promotion #157; (3) maintain the requirements specified in Section 3.4.3 (Y).2.a, Section 3.4.3 (Y).2.b, and Section 3.4.3 (Y).3 of this Tariff; and (4) continue to subscribe to the Value Plus 60 optional calling plan for the provision of interstate and intrastate InterLATA calling during the benefit period of this promotional offering.

The Company will waive the requirements shown in Section 3.4.3 (Y).5.a.i of this Tariff for Applicants and Customers participating in this promotion. For Customers participating in this promotion, the following monthly recurring charges and per minute usage rates apply for the first twelve (12) months from activation date of Service. The monthly recurring charge is \$2.00. The interstate and intrastate rate is \$0.05 per minute after the 60 minute block of time has been exhausted.

If the Customer fails to maintain the requirements specified in Section 3.4.3 (Y).2.a, Section 3.4.3 (Y).2.b, and Section 3.4.3 (Y).3 of this Tariff and/or fails to continue to subscribe to the Value Plus 60 optional calling plan for the provision of interstate and intrastate InterLATA calling, the Customer will no longer qualify for the promotional reduced rates listed above.

The usage rate that applies after the expiration of the promotional rate may be found in Section 4.4.3 (Y).2.a of this Tariff.

Issued: March 27, 2003

Effective: April 12, 2003

Ν

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Original Sheet 253

#### 6.16 Value Plus 200 Promotion #158

The sign-up period for the Value Plus 200 Promotion #158 is April 12, 2003 through June 30, 2003. Service must be activated by July 5, 2003. This promotion is available to new and existing Residential Customers in Missouri that (1) subscribe to the Value Plus 200 optional calling plan during the sign-up period; (2) subscribe to the Company's interstate Value Plus 200 Promotion #158; (3) maintain the requirements specified in Section 3.4.3 (Y).2.a, Section 3.4.3 (Y).2.b, and Section 3.4.3 (Y).3 of this Tariff; and (4) continue to subscribe to the Value Plus 200 optional calling plan for the provision of interstate and intrastate InterLATA calling during the benefit period of this promotional offering.

The Company will waive the requirements shown in Section 3.4.3 (Y).5.b.i of this Tariff for Applicants and Customers participating in this promotion. For Customers participating in this promotion, the following per minute usage rates apply for the first twelve (12) months from activation date of Service. The interstate and intrastate rate is \$0.05 per minute after the 200 minute block of time has been exhausted.

If the Customer fails to maintain the requirements specified in Section 3.4.3 (Y).2.a, Section 3.4.3 (Y).2.b, and Section 3.4.3 (Y).3 of this Tariff and/or fails to continue to subscribe to the Value Plus 200 optional calling plan for the provision of interstate and intrastate InterLATA calling, the Customer will no longer qualify for the promotional reduced rates listed above.

The usage rate that applies after the expiration of the promotional rate may be found in Section 4.4.3 (Y).2.b of this Tariff.

Ν

Issued: March 27, 2003

Effective: April 12, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Ν

Ν

#### **SECTION 6 - PROMOTIONS**

6.17 Value Plus 500 Promotion #159

The sign-up period for the Value Plus 500 Promotion #159 is April 12, 2003 through June 30, 2003. Service must be activated by July 5, 2003. This promotion is available to new and existing Residential Customers in Missouri that (1) subscribe to the Value Plus 500 optional calling plan during the sign-up period; (2) subscribe to the Company's interstate Value Plus 500 Promotion #159; (3) maintain the requirements specified in Section 3.4.3 (Y).2.a, Section 3.4.3 (Y).2.b, and Section 3.4.3 (Y).3 of this Tariff; and (4) continue to subscribe to the Value Plus 500 optional calling plan for the provision of interstate and intrastate InterLATA calling during the benefit period of this promotional offering.

The Company will waive the requirements shown in Section 3.4.3 (Y).5.c.i of this Tariff for Applicants and Customers participating in this promotion. For Customers participating in this promotion, the following per minute usage rates apply for the first twelve (12) months from activation date of Service. The interstate and intrastate rate is \$0.05 per minute after the 500 minute block of time has been exhausted.

If the Customer fails to maintain the requirements specified in Section 3.4.3 (Y).2.a, Section 3.4.3 (Y).2.b, and Section 3.4.3 (Y).3 of this Tariff and/or fails to continue to subscribe to the Value Plus 500 optional calling plan for the provision of interstate and intrastate InterLATA calling, the Customer will no longer qualify for the promotional reduced rates listed above.

The usage rate that applies after the expiration of the promotional rate may be found in Section 4.4.3 (Y).2.c of this Tariff.

Ν

Issued: March 27, 2003

Effective: April 12, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

### 6.18 National Connections Promotion #193

The sign-up period for the National Connections Promotion #193 is April 17, 2003 through May 17, 2003. Service must be activated by May 22, 2003. This promotion is available to new and existing Residential Customers in Missouri that subscribe to the National Connections optional calling plan during the sign-up period. For Customers participating in this promotion, the Company will waive the requirements shown in Section 3.4.3 (X).4.a.i of this Tariff if the Customer subscribes to and maintains an access line service, Caller ID, and a minimum of any three custom calling service features from Group C Large Package from an SBC Affiliated.

If the Customer fails to maintain the requirements specified above, the Customer will no longer qualify for National Connections and will be moved to Long Distance II unless the Customer selects another optional calling plan.

Ν

N

Issued: April 10, 2003

Effective: April 17, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

#### 6.19 ValueSaver Promotion #195

÷,

The sign-up period for the ValueSaver Promotion #195 is April 18, 2003 through October 8, C 2003. Service must be activated by October 13, 2004. This promotion is available to new and C existing Residential Customers in Missouri that:

- (1) use Switched Access to reach the long distance network;
- (2) subscribe to the Company for the provision of interstate and intrastate InterLATA Service or subscribe to the Company for the provision of interstate, intrastate InterLATA, and intrastate IntraLATA Service;
- (3) subscribe to ValueSaver during the sign-up period;
- (4) demonstrate to the satisfaction of the Company at the time of subscribing to the plan that the Residential Customer subscribes to an access line service of a SBC Affiliate;
- (5) request to participate in this promotional offering;
- (6) provide the Company the same billing name and address for the SBC affiliate access line service and the Company's Service;
- (7) limit the use of Service to that which is of a standard, domestic, residential nature; and
- (8) bill the access line service of a SBC Affiliate to the same BTN as the Customer's long distance Service.

All other requirements specified in Section 3.4.3 (V).1 of this Tariff will be waived for Customers participating in this promotion. Customers participating in this promotion will be billed the per minute rates described in Section 4.4.3 (V) of this Tariff which are \$0.07 per minute as of the effective date of the promotional offering. The per minute rates may change pursuant to appropriate notice to the Customer and the Commission.

Issued: September 8, 2003

Effective: October 8, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

# Southwestern Bell Communications Services, Inc.PSC Mo. - No. 13rd Revised Sheet 257d/b/a SBC Long DistanceReplacing 2nd Revised Sheet 257Replacing 2nd Revised Sheet 257VIISSOURI Public

### **SECTION 6 - PROMOTIONS**

### REC'D AUG 2 8 2003

### 6.19 ValueSaver Promotion #195

Service Commission

The sign-up period for the ValueSaver Promotion #195 is April 18, 2003 through December 31, 2003. Service must be activated by January 6, 2004. This promotion is available to new C and existing Residential Customers in Missouri that:

- (1) use Switched Access to reach the long distance network;
- (2) subscribe to the Company for the provision of interstate and intrastate InterLATA Service or subscribe to the Company for the provision of interstate, intrastate InterLATA, and intrastate IntraLATA Service;
- (3) subscribe to ValueSaver during the sign-up period;
- (4) demonstrate to the satisfaction of the Company at the time of subscribing to the plan that the Residential Customer subscribes to an access line service of a SBC Affiliate;
- (5) request to participate in this promotional offering;
- (6) provide the Company the same billing name and address for the SBC affiliate access line service and the Company's Service;
- (7) limit the use of Service to that which is of a standard, domestic, residential nature; and
- (8) bill the access line service of a SBC Affiliate to the same BTN as the Customer's long distance Service.

All other requirements specified in Section 3.4.3 (V).1 of this Tariff will be waived for Customers participating in this promotion. Customers participating in this promotion will be billed the per minute rates described in Section 4.4.3 (V) of this Tariff which are \$0.07 per minute as of the effective date of the promotional offering. The per minute rates may change pursuant to appropriate notice to the Customer and the Commission.

## CANCELLED

 OCT 0.8 2003

 Jubility Statute

 Issued: August 28, 2003

 Jubility Commission

 Tawnya Rechtin, Associate Director Regulatory

 5850 W. Las Positas Blvd., Pleasanton, California 94588

 CANCELLED

 December 24, 2004

 Missouri Public

 Service Commission

 TN-2005-0149

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

2nd Revised Sheet 257 Replacing 1st Revised Sheet 257 Missouri Public

### **SECTION 6 - PROMOTIONS**

6.19 ValueSaver Promotion #195

> The sign-up period for the ValueSaver Promotion #195 is April 18, 2003 through September С 30, 2003. Service must be activated by October 5, 2003. This promotion is available to new and existing Residential Customers in Missouri that:

- (I)use Switched Access to reach the long distance network;
- (2) subscribe to the Company for the provision of interstate and intrastate InterLATA Service or subscribe to the Company for the provision of interstate, intrastate InterLATA, and intrastate IntraLATA Service;
- subscribe to ValueSaver during the sign-up period; (3)
- demonstrate to the satisfaction of the Company at the time of subscribing to the plan (4) that the Residential Customer subscribes to an access line service of a SBC Affiliate;
- request to participate in this promotional offering; (5)
- provide the Company the same billing name and address for the SBC affiliate access (6) line service and the Company's Service;
- (7)limit the use of Service to that which is of a standard, domestic, residential nature; and
- (8) bill the access line service of a SBC Affiliate to the same BTN as the Customer's long distance Service.

All other requirements specified in Section 3.4.3 (V).1 of this Tariff will be waived for Customers participating in this promotion. Customers participating in this promotion will be billed the per minute rates described in Section 4.4.3 (V) of this Tariff which are \$0.07 per minute as of the effective date of the promotional offering. The per minute rates may change pursuant to appropriate notice to the Customer and the Commission.

> SEP 0.6 2003 Effective: July 14, 2003

CANCELLED

Issued: July 7, 2003

Tawnya Rechtin, Associate Encoder 20 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

## RECT JUL 07 2003

Service Commission

С

FILED JUL 14 2003

### Missourl Public Service Commission

6.19 ValueSaver Promotion #195

RFC'D MAY 23 2003

The sign-up period for the ValueSaver Promotion #195 is April 18, 2003 through July 31, C 2003. Service must be activated by August 5, 2003. This promotion is available to new and C existing Residential Customers in Missouri that:

- (1) use Switched Access to reach the long distance network;
- (2) subscribe to the Company for the provision of interstate and intrastate InterLATA Service or subscribe to the Company for the provision of interstate, intrastate InterLATA, and intrastate IntraLATA Service;
- (3) subscribe to ValueSaver during the sign-up period;
- (4) demonstrate to the satisfaction of the Company at the time of subscribing to the plan that the Residential Customer subscribes to an access line service of a SBC Affiliate;
- (5) request to participate in this promotional offering;
- (6) provide the Company the same billing name and address for the SBC affiliate access line service and the Company's Service;
- (7) limit the use of Service to that which is of a standard, domestic, residential nature; and
- (8) bill the access line service of a SBC Affiliate to the same BTN as the Customer's long distance Service.

All other requirements specified in Section 3.4.3 (V).1 of this Tariff will be waived for Customers participating in this promotion. Customers participating in this promotion will be billed the per minute rates described in Section 4.4.3 (V) of this Tariff which are \$0.07 per minute as of the effective date of the promotional offering. The per minute rates may change pursuant to appropriate notice to the Customer and the Commission.

## CANCELLED

JUL 1 4 2003 MISSION Jildu

Effective: May 31, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public Service Commission

FILED MAY 31 2003

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Issued: May 23, 2003
Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

Original Sheet 257

Missouri Public Service Commission

RF(T) APR 11 2003

Ν

### SECTION 6 - PROMOTIONS

#### 6.19 ValueSaver Promotion #195

TN-2005-0149

The sign-up period for the ValueSaver Promotion #195 is April 18, 2003 through May 31, 2003. Service must be activated by June 5, 2003. This promotion is available to new and existing Residential Customers in Missouri that:

- (1) use Switched Access to reach the long distance network;
- (2) subscribe to the Company for the provision of interstate and intrastate InterLATA Service or subscribe to the Company for the provision of interstate, intrastate InterLATA, and intrastate IntraLATA Service;
- (3) subscribe to ValueSaver during the sign-up period;
- (4) demonstrate to the satisfaction of the Company at the time of subscribing to the plan that the Residential Customer subscribes to an access line service of a SBC Affiliate;
- (5) request to participate in this promotional offering;
- (6) provide the Company the same billing name and address for the SBC affiliate access line service and the Company's Service;
- (7) limit the use of Service to that which is of a standard, domestic, residential nature; and
- (8) bill the access line service of a SBC Affiliate to the same BTN as the Customer's long distance Service.

All other requirements specified in Section 3.4.3 (V).1 of this Tariff will be waived for Customers participating in this promotion. Customers participating in this promotion will be billed the per minute rates described in Section 4.4.3 (V) of this Tariff which are \$0.07 per minute as of the effective date of the promotional offering. The per minute rates may change pursuant to appropriate notice to the Customer and the Commission.

CANCELLED MAY 3 1 2003 Issued: April 11, 2003 BMISSION Effective: April 18, 2003 Public Service Co Tawnya Rechtin, Associate Director Regulatory Tawnya Rechtin, Associate Director Regeneration 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missourt Public Service Commission CANCELLED FILED APR 18 2003 December 24, 2004 Missouri Public Service Commission

### 6.20 Promotion #189 Business Domestic Saver Plus

¢

The sign-up period for Promotion #189, Business Domestic Saver Plus, is June 16, 2003 through March 31, 2004. Service must be activated by April 30, 2004. To participate in this C promotion, the Customer must subscribe to Business Domestic Saver, as referenced in Section 3.7.10 of this Tariff.

For Customers participating in this promotion, the following rates apply in lieu of the rate shown in Section 4.7.10 of this Tariff. The usage rate for intrastate 1+ Direct-Dialed outbound calls and switched Toll Free Service is \$0.07 per minute for the first 364 days of subscribing to Business Domestic Saver.

Issued: December 1, 2003

Effective: December 31, 2003

Norm Descoteaux, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

1st Revised Sheet 258 Replacing Original Sheet 258

### **SECTION 6 - PROMOTIONS**

Missouri Public REC'D SEP 2 4 2003

6.20 Promotion #189 Business Domestic Saver Plus

Service Commission

The sign-up period for Promotion #189, Business Domestic Saver Plus, is June 16, 2003 through December 31, 2003. Service must be activated by January 31, 2004. To participate if this promotion, the Customer must subscribe to Business Domestic Saver, as referenced in Section 3.7.10 of this Tariff.

For Customers participating in this promotion, the following rates apply in lieu of the rate shown in Section 4.7.10 of this Tariff. The usage rate for intrastate 1+ Direct-Dialed outbound calls and switched Toll Free Service is \$0.07 per minute for the first 364 days of subscribing to Business Domestic Saver.



DEC 3 1 2003 by 2<sup>nd</sup> RS 258 Public Bervice Commission Missouri

Issued: September 24, 2003

Effective: October 1, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Missouri Public Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

# FLED OCT 01 2003

### Missouri Public Service Commissión

# REC'D JUN 0 5 2003

	western Bell Communications Services, Inc.	PSC Mo. ~ No. 1	Original Sheet 258			
	SECTION 6 - PROMOTIONS					
6.20	Promotion #189 Business Domestic Save	er Plus				
	The sign-up period for Promotion #189,	Business Domestic Saver	Plus, is June 16, 2003			
	through September 30, 2003. Service m	ust be activated by Octobe	π 30, 2003. To participate			
	in this promotion, the Customer must su	bscribe to Business Dome.	stic Saver, as referenced in			
	Section 3.7.10 of this Tariff.					
	For Customers participating in this prom	otion, the following rates	apply in lieu of the rate			
	shown in Section 4.7.10 of this Tariff. T	The usage rate for intrastate	e 1+ Direct-Dialed			
	outbound calls and switched Toll Free S	ervice is \$0.07 per minute	for the first 364 days of			
	subscribing to Business Domestic Saver.					

# CANCELLED

OCT 0 1 2003 STRS 258 Public Commission Miceory

Issued: June 5, 2003

Effective: June 16, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588



FILED JUN 1 6 2003

### Original Sheet 259

### **SECTION 6 - PROMOTIONS**

### 6.21 60 Block of Time II and 200 Block of Time II Promotion #214

The sign up period for Promotion #214 is September 15, 2003 through November 13, 2003. Orders for new Service must be activated by November 18, 2003. This promotion is only available to Residential Customers that subscribe to one of the Block of Time II plans, described in Section 3.4.3 (Z) of this Tariff, during the sign-up period. For the initial two full billing cycles following the Customer participating in this promotion, the Customer will not be billed the MRC described in Section 4.4.3 (Z) of this Tariff. For existing Customers who sign up for this promotion in the middle of a billing cycle, the Customer will not be billed the MRC described in Section 4.4.3 (Z) of this Tariff for the month of the sign-up and for the two full billing cycles following the Customer participating in this promotion.

Ν

Ν

Issued: September 4, 2003

Effective: September 15, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

7

### 6.22 Value Plus Flat Rate Promotion #156

Э

The sign-up period for the Value Plus Flat Rate Promotion #156 is October 8, 2003 through December 31, 2003. Service must be activated by January 6, 2004. This promotion is available to new and existing Residential Customers in Missouri that (1) subscribe to the Value Plus Flat Rate optional calling plan during the sign-up period; (2) subscribe to the Company's interstate Value Plus Flat Rate Promotion #156; (3) maintain the requirements specified in Section 3.4.3 (Y).2.a, Section 3.4.3 (Y).2.b, and Section 3.4.3 (Y).3 of this Tariff; and (4) continue to subscribe to the Value Plus Flat Rate optional calling plan for the provision of interstate and intrastate InterLATA calling during the benefit period of this promotional offering.

The Company will waive the requirements shown in Section 3.4.3 (Y).4.a of this Tariff for Applicants and Customers participating in this promotion. For Customers participating in this promotion, the following per minute usage rates apply for the first twelve (12) months from activation date of Service.

-	interstate usage	\$0.05
-	intrastate usage - peak rate period	\$0.27
-	intrastate usage - off peak rate period	\$0.17

The peak rate period is 8:00 a.m. to but not including 5:00 p.m., Monday through Friday. The off-peak rate period is all other times. The off-peak rates apply on the following holidays: New Year's Day, Independence Day, Labor Day, Thanksgiving Day, and Christmas Day.

If the Customer fails to maintain the requirements specified in Section 3.4.3 (Y).2.a, Section 3.4.3 (Y).2.b, and Section 3.4.3 (Y).3 of this Tariff and/or fails to continue to subscribe to the Value Plus Flat Rate optional calling plan for the provision of interstate and intrastate InterLATA calling, the Customer will no longer qualify for the promotional reduced rates listed above.

The usage rate that applies after the expiration of the promotional rate may be found in Section 4.4.3 (Y).1.a of this Tariff.

Issued: September 8, 2003

Effective: October 8, 2003

N

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

6.23 Value Plus 60 Promotion #157

The sign-up period for the Value Plus 60 Promotion #157 is October 8, 2003 through March 31, 2004. Service must be activated by April 15, 2004. This promotion is available to new and existing Residential Customers in Missouri that (1) subscribe to the Value Plus 60 optional calling plan during the sign-up period; (2) subscribe to the Company's interstate Value Plus 60 Promotion #157; (3) maintain the requirements specified in Section 3.4.3 (Y).2.a, Section 3.4.3 (Y).2.b, and Section 3.4.3 (Y).3 of this Tariff; and (4) continue to subscribe to the Value Plus 60 optional calling plan for the provision of interstate and intrastate InterLATA calling during the benefit period of this promotional offering.

The Company will waive the requirements shown in Section 3.4.3 (Y).4.b of this Tariff for Applicants and Customers participating in this promotion.

Issued: December 17, 2003

Effective: December 31, 2003

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Original Sheet 261

N

N

### **SECTION 6 - PROMOTIONS**

# REC'D SEP 08 2003

6.23 Value Plus 60 Promotion #157

Service Commission

The sign-up period for the Value Plus 60 Promotion #157 is October 8, 2003 through December 31, 2003. Service must be activated by January 6, 2004. This promotion is available to new and existing Residential Customers in Missouri that (1) subscribe to the Value Plus 60 optional calling plan during the sign-up period; (2) subscribe to the Company's interstate Value Plus 60 Promotion #157; (3) maintain the requirements specified in Section 3.4.3 (Y).2.a, Section 3.4.3 (Y).2.b, and Section 3.4.3 (Y).3 of this Tariff; and (4) continue to subscribe to the Value Plus 60 optional calling plan for the provision of interstate and intrastate InterLATA calling during the benefit period of this promotional offering.

The Company will waive the requirements shown in Section 3.4.3 (Y).4.b of this Tariff for Applicants and Customers participating in this promotion.



DEC 3 1 2003 St RS 261 Public Service Commission MISSOURI

Issued: September 8, 2003

Effective: October 8, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

ł

Missouri Public Service Commission

FLED OCT 08 2003

С

### **SECTION 6 - PROMOTIONS**

6.24 Value Plus 200 Promotion #158

The sign-up period for the Value Plus 200 Promotion #158 is October 8, 2003 through March 31, 2004. Service must be activated by April 15, 2004. This promotion is available to new and existing Residential Customers in Missouri that (1) subscribe to the Value Plus 200 optional calling plan during the sign-up period; (2) subscribe to the Company's interstate Value Plus 200 Promotion #158; (3) maintain the requirements specified in Section 3.4.3 (Y).2.a, Section 3.4.3 (Y).2.b, and Section 3.4.3 (Y).3 of this Tariff; and (4) continue to subscribe to the Value Plus 200 optional calling plan for the provision of interstate and intrastate InterLATA calling during the benefit period of this promotional offering.

The Company will waive the requirements shown in Section 3.4.3 (Y).4.c of this Tariff for Applicants and Customers participating in this promotion.

Issued: December 17, 2003

Effective: December 31, 2003

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance Original Sheet 262

N

Ν

Missouri Public

### SECTION 6 - PROMOTIONS

## REC'D SEP 08 2003

6.24 Value Plus 200 Promotion #158

Service Commission The sign-up period for the Value Plus 200 Promotion #158 is October 8, 2003 through December 31, 2003. Service must be activated by January 6, 2004. This promotion is available to new and existing Residential Customers in Missouri that (1) subscribe to the Value Plus 200 optional calling plan during the sign-up period; (2) subscribe to the Company's interstate Value Plus 200 Promotion #158; (3) maintain the requirements specified in Section 3.4.3 (Y).2.a, Section 3.4.3 (Y).2.b, and Section 3.4.3 (Y).3 of this Tariff; and (4) continue to subscribe to the Value Plus 200 optional calling plan for the provision of interstate and intrastate InterLATA calling during the benefit period of this promotional offering.

The Company will waive the requirements shown in Section 3.4.3 (Y).4.c of this Tariff for Applicants and Customers participating in this promotion.

CANCELLED

DEC 3 1 2003 St RS 262 Public Service Commission MISSOURI

Issued: September 8, 2003

Effective: October 8, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Missouri Public Service Commission

FILÉD OCT 08 2003

С

### **SECTION 6 - PROMOTIONS**

6.25 Value Plus 500 Promotion #159

The sign-up period for the Value Plus 500 Promotion #159 is October 8, 2003 through March 31, 2004. Service must be activated by April 15, 2004. This promotion is available to new and existing Residential Customers in Missouri that (1) subscribe to the Value Plus 500 optional calling plan during the sign-up period; (2) subscribe to the Company's interstate Value Plus 500 Promotion #159; (3) maintain the requirements specified in Section 3.4.3 (Y).2.a, Section 3.4.3 (Y).2.b, and Section 3.4.3 (Y).3 of this Tariff; and (4) continue to subscribe to the Value Plus 500 optional calling plan for the provision of interstate and intrastate InterLATA calling during the benefit period of this promotional offering.

The Company will waive the requirements shown in Section 3.4.3 (Y).4.d of this Tariff for Applicants and Customers participating in this promotion.

Issued: December 17, 2003

Effective: December 31, 2003

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 Micoouri Publicinal Sheet 263 d/b/a SBC Long Distance

## SECTION 6 - PROMOTIONS () SEP 08 2003

6.25 Value Plus 500 Promotion #159

### Service Commission

Ν

N

The sign-up period for the Value Plus 500 Promotion #159 is October 8, 2003 through December 31, 2003. Service must be activated by January 6, 2004. This promotion is available to new and existing Residential Customers in Missouri that (1) subscribe to the Value Plus 500 optional calling plan during the sign-up period; (2) subscribe to the Company's interstate Value Plus 500 Promotion #159; (3) maintain the requirements specified in Section 3.4.3 (Y).2.a, Section 3.4.3 (Y).2.b, and Section 3.4.3 (Y).3 of this Tariff; and (4) continue to subscribe to the Value Plus 500 optional calling plan for the provision of interstate and intrastate InterLATA calling during the benefit period of this promotional offering.

The Company will waive the requirements shown in Section 3.4.3 (Y).4.d of this Tariff for Applicants and Customers participating in this promotion.

# CANCELLED

DEC 3 1 2003 IS RS 243 -ubijo Service Commission MISSOURI

Issued: September 8, 2003

Effective: October 8, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Missourl Public Service Commission

FLED OCT 08 2003

### 6.26 500 Block of Time II Promotion #223

The sign up period for Promotion #223 is November 1, 2003 through January 6, 2004. Orders for new Service must be activated by January 11, 2004. This promotion is only available to Residential Customers that subscribe to 500 Block of Time II described in Section 3.4.3 (Z) of this Tariff during the sign-up period. For the initial two full billing cycles following the Customer participating in this promotion, the Customer will be billed an MRC of \$11.47 in lieu of the MRC described in Section 4.4.3 (Z).3 of this Tariff. For existing Customers who sign up for this promotion in the middle of a billing cycle, the Customer will be billed an MRC of \$11.47 for the month of the sign-up and for the two full billing cycles following the Customer participating in this promotion.

Issued: October 20, 2003

Effective: November 1, 2003

Tawnya Rechtin, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588 Original Sheet 264

N

### 6.27 JustCall<sup>SM</sup> Unlimited Weekends Promotion #228

The sign-up period for Promotion #228, JustCall<sup>SM</sup> Unlimited Weekends, is January 12, 2004 through March 31, 2004. Service must be activated by April 15, 2004. To participate in this C promotion, the Customer must subscribe to JustCall<sup>SM</sup> Unlimited Weekends, as referenced in Section 3.4.3 (AD) of this Tariff, during the sign-up period and request to participate in this promotional offering.

For Customers participating in this promotion, the usage rate for peak rate period MOU is \$0.05 per minute for the first twelve (12) months of subscribing to JustCall<sup>SM</sup> Unlimited Weekends in lieu of the per minute usage rate shown in Section 4.4.3 (AD) of this Tariff.

Issued: January 12, 2004

Effective: January 19, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 d/b/a SBC Long Distance

### SECTION 6 - PROMOTIONS

# PROMOTIONS REC'D DEC 11 2003

Original Sheet 265

N

Ν

Missouri Public

Service Commission

### 6.27 JustCall<sup>SM</sup> Unlimited Weekends Promotion #228

The sign-up period for Promotion #228, JustCall<sup>SM</sup> Unlimited Weekends, is January 12, 2004 through April 12, 2004. Service must be activated by April 27, 2004. To participate in this promotion, the Customer must subscribe to JustCall<sup>SM</sup> Unlimited Weekends, as referenced in Section 3.4.3 (AD) of this Tariff, during the sign-up period and request to participate in this promotional offering.

For Customers participating in this promotion, the usage rate for peak rate period MOU is \$0.05 per minute for the first twelve (12) months of subscribing to JustCall<sup>SM</sup> Unlimited Weekends in lieu of the per minute usage rate shown in Section 4.4.3 (AD) of this Tariff.



Issued: December 11, 2003

Effective: January 12, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

> Missouri Public Service Commission

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

FILED JAN 12 2004

### 6.28 Business Domestic Saver Solutions Promotion #230

The sign up period for Promotion #230 is April 1, 2004 through March 31, 2005. Service C must be activated by May 31, 2005. To participate in this promotion, the Customer must C subscribe to Business Domestic Saver as described in Section 3.7.10 of this Tariff for the provision of interstate Service. For Customers participating in this promotion, the following rate applies in lieu of the rate shown in Section 4.7.10 of this Tariff. The usage rate for intrastate 1+ Direct-Dialed outbound and switched Toll Free Service calls is \$0.0700 per MOU for the first twelve (12) months of subscribing to Business Domestic Saver under this promotional offering.

Issued: September 20, 2004

Effective: September 30, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

## Missouri Public

REC'D MAR 0 8 2004

Service Commission Original Sheet 266

Ν

Ν

Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u>

### **SECTION 6 - PROMOTIONS**

6.28 Business Domestic Saver Solutions Promotion #230

The sign up period for Promotion #230 is April 1, 2004 through September 30, 2004. Service must be activated by October 30, 2004. To participate in this promotion, the Customer must subscribe to Business Domestic Saver as described in Section 3.7.10 of this Tariff for the provision of interstate Service. For Customers participating in this promotion, the following rate applies in lieu of the rate shown in Section 4.7.10 of this Tariff. The usage rate for intrastate 1+ Direct-Dialed outbound and switched Toll Free Service calls is \$0.0700 per MOU for the first twelve (12) months of subscribing to Business Domestic Saver under this promotional offering.

CANCELLED

SEP 3.0 2004 SEP 3.0 2004 SEP 3.0 2004 Service Commission

Issued: March 8, 2004

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Missouri Public Sorvice Commission

FLED APR 01 2004

### 6.29 JustCallK 60 Preferred Promotion #252

The sign-up period for the JustCallK 60 Preferred Promotion #252 is April 1, 2004 through March 31, 2005. Service must be activated by April 15, 2005. This promotion is available to Residential Customers in the State that have (a) previously subscribed to local dial tone service from an SBC Affiliate and have cancelled that service or (b) previously subscribed to long distance Service from the Company and have cancelled that Service. To participate in the promotion, Residential Customers must (1) subscribe to the JustCallK 60 Preferred optional calling plan during the sign-up period; (2) subscribe to the Company's interstate JustCallK 60 Preferred Promotion #252; (3) maintain the requirements specified in Section 3.4.3 (AE).2, Section 3.4.3 (AE).3, and Section 3.4.3 (AE).4 of this Tariff; and (4) continue to subscribe to the JustCallK 60 Preferred optional calling plan for the provision of interstate, intrastate InterLATA, and intrastate IntraLATA calling during the benefit period of this promotional offering.

For Customers participating in this promotion, the following monthly recurring charge applies for the first twelve (12) months from activation date of Service. The monthly recurring charge is \$2.00. The intrastate rate is \$0.07 per minute after the 60 minute block of time has been exhausted.

If the Customer fails to maintain the requirements specified in Section 3.4.3 (AE).2, Section 3.4.3 (AE).3, and Section 3.4.3 (AE).4 of this Tariff and/or fails to continue to subscribe to the JustCallK 60 Preferred optional calling plan for the provision of interstate and intrastate InterLATA and intrastate IntraLATA calling, the Customer will no longer qualify for the promotional reduced MRC rate listed above. The MRC rate that applies after the expiration of the promotional rate may be found in Section 4.4.3 (AE).10.a of this Tariff.

Ν

Issued: March 8, 2004

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

### 6.30 JustCallK 200 Preferred Promotion #253

The sign-up period for the JustCallK 200 Preferred Promotion #253 is April 1, 2004 through March 31, 2005. Service must be activated by April 15, 2005. This promotion is available to Residential Customers in the State that have (a) previously subscribed to local dial tone service from an SBC Affiliate and have cancelled that service or (b) previously subscribed to long distance Service from the Company and have cancelled that Service. To participate in the promotion, Residential Customers must (1) subscribe to the JustCallK 200 Preferred optional calling plan during the sign-up period; (2) subscribe to theCompany's interstate JustCallK 200 Preferred Promotion #253; (3) maintain the requirements specified in Section 3.4.3 (AE).2, Section 3.4.3 (AE).3, and Section 3.4.3 (AE).4 of this Tariff; and (4) continue to subscribe to the JustCallK 200 Preferred optional calling plan for the provision of interstate, intrastate InterLATA, and intrastate IntraLATA calling during the benefit period of this promotional offering.

For Customers participating in this promotion, the following monthly recurring charge applies for the first twelve (12) months from activation date of Service. The monthly recurring charge is \$6.00. The intrastate rate is \$0.07 per minute after the 200 minute block of time has been exhausted.

If the Customer fails to maintain the requirements specified in Section 3.4.3 (AE).2, Section 3.4.3 (AE).3, and Section 3.4.3 (AE).4 of this Tariff and/or fails to continue to subscribe to the JustCallK 200 Preferred optional calling plan for the provision of interstate and intrastate InterLATA and intrastate IntraLATA calling, the Customer will no longer qualify for the promotional reduced MRC rate listed above. The MRC rate that applies after the expiration of the promotional rate may be found in Section 4.4.3 (AE).10.b of this Tariff.

Ν

Issued: March 8, 2004

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

Ν

### **SECTION 6 - PROMOTIONS**

### 6.31 JustCallK 400 Preferred Promotion #254

The sign-up period for the JustCallK 400 Preferred Promotion #254 is April 1, 2004 through March 31, 2005. Service must be activated by April 15, 2005. This promotion is available to Residential Customers in the State that have (a) previously subscribed to local dial tone service from an SBC Affiliate and have cancelled that service or (b) previously subscribed to long distance Service from the Company and have cancelled that Service. To participate in the promotion, Residential Customers must (1) subscribe to the JustCallK 400 Preferred optional calling plan during the sign-up period; (2) subscribe to the Company's interstate JustCallK 400 Preferred Promotion #254; (3) maintain the requirements specified in Section 3.4.3 (AE).2, Section 3.4.3 (AE).3, and Section 3.4.3 (AE).4 of this Tariff; and (4) continue to subscribe to the JustCallK 400 Preferred optional calling plan for the provision of interstate, intrastate InterLATA, and intrastate IntraLATA calling during the benefit period of this promotional offering.

For Customers participating in this promotion, the following monthly recurring charge applies for the first twelve (12) months from activation date of Service. The monthly recurring charge is \$10.00. The intrastate rate is \$0.07 per minute after the 400 minute block of time has been exhausted.

If the Customer fails to maintain the requirements specified in Section 3.4.3 (AE).2, Section 3.4.3 (AE).3, and Section 3.4.3 (AE).4 of this Tariff and/or fails to continue to subscribe to the JustCallK 400 Preferred optional calling plan for the provision of interstate and intrastate InterLATA and intrastate IntraLATA calling, the Customer will no longer qualify for the promotional reduced MRC rate listed above. The MRC rate that applies after the expiration of the promotional rate may be found in Section 4.4.3 (AE).10.c of this Tariff.

Ν

Issued: March 8, 2004

Effective: April 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

- 6.32 JustCall<sup>SM</sup> 7 Cents Preferred Promotion #276
  - (A) The sign-up period for this promotion is June 15, 2004 through March 31, 2005. Orders for new Service must be activated by April 15, 2005. This promotion is available to Residential Customers that previously subscribed to local dial tone service from an SBC Affiliate and/or subscribed to a long distance Service of the Company and have cancelled that service. To receive the promotion, a Customer must now subscribe to local dial tone service from an SBC Affiliate and the Company's interstate interexchange JustCall<sup>SM</sup> 7 Cent optional calling plan through a Company-designated outbound sales contact or by calling a toll-free number provided through Company-designated Teleservices sales channels that are specific to this promotion.
  - (B) To participate in the promotion, Residential Customers must (1) subscribe to the JustCall<sup>SM</sup> 7 Cents Preferred optional calling plan, as referenced in Section 3.4.3 (AE).9.b of this Tariff during the sign-up period; (2) maintain the requirements specified in Section 3.4.3 (AE).1, 3.4.3 (AE).3, and Section 3.4.3 (AE).4, of this Tariff; and (3) continue to subscribe to the JustCall<sup>SM</sup> 7 Cents Preferred optional calling plan for the provision of intrastate calling during the benefit period of this promotional offering.
  - (C) Customers participating in this promotion will receive up to the first 30 minutes of interstate/intrastate usage free-of-charge per month and a reduction of the \$0.07 intrastate per minute rate after the first 30 minutes to \$0.05 per minute for the first six (6) months from activation of Service.
  - (D) If the Customer fails to maintain the requirements specified in Section 3.4.3 (AE).1, Section 3.4.3 (AE).3, and Section 3.4.3 (AE).4 of this Tariff and/or fails to continue to subscribe to the JustCall<sup>SM</sup> 7 Cents Preferred optional calling plan for the provision of intrastate calling, the Customer will no longer qualify for the promotion. The per minute usage rate that applies after the expiration of the promotion may be found in Section 4.4.3 (AE).9.b of this Tariff.

Issued: December 22, 2004

Effective: December 31, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

С

C

### **SECTION 6 - PROMOTIONS**

### 6.32 JustCall<sup>SM</sup> 7 Cents Preferred Promotion #276

- (A) The sign-up period for this promotion is June 15, 2004 through December 31, 2004. Orders for new Service must be activated by January 15, 2005. This promotion is available to Residential Customers that previously subscribed to local dial tone service from an SBC Affiliate and/or subscribed to a long distance Service of the Company and have cancelled that service. To receive the promotion, a Customer must now subscribe to local dial tone service from an SBC Affiliate and the Company's interstate interexchange JustCall<sup>SM</sup> 7 Cent optional calling plan through a Company-designated outbound sales contact or by calling a toll-free number provided through Company-designated Teleservices sales channels that are specific to this promotion.
- (B) To participate in the promotion, Residential Customers must (1) subscribe to the JustCall<sup>SM</sup> 7 Cents Preferred optional calling plan, as referenced in Section 3.4.3 (AE).9.b of this Tariff during the sign-up period; (2) maintain the requirements specified in Section 3.4.3 (AE).1, 3.4.3 (AE).3, and Section 3.4.3 (AE).4, of this Tariff; and (3) continue to subscribe to the JustCall<sup>SM</sup> 7 Cents Preferred optional calling plan for the provision of intrastate calling during the benefit period of this promotional offering.
- (C) Customers participating in this promotion will receive up to the first 30 minutes of interstate/intrastate usage free-of-charge per month and a reduction of the \$0.07 intrastate per minute rate after the first 30 minutes to \$0.05 per minute for the first six (6) months from activation of Service.
- (D) If the Customer fails to maintain the requirements specified in Section 3.4.3 (AE).1, Section 3.4.3 (AE).3, and Section 3.4.3 (AE).4 of this Tariff and/or fails to continue to subscribe to the JustCall<sup>SM</sup> 7 Cents Preferred optional calling plan for the provision of intrastate calling, the Customer will no longer qualify for the promotion. The per minute usage rate that applies after the expiration of the promotion may be found in Section 4.4.3 (AE).9.b of this Tariff.

## CANCELLED

DEC 3 1 2004

Issued: August 20, 2004

Public Service Commission

Effective: August 31, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588



Southwestern Bell Communications Services, Inc. PSC Mo. - No. 1 <u>d/b/a SBC Long Distance</u>

Original Sheet 270

### SECTION 6 - PROMOTIONS

### Missouri Public Service Commission

6.32 JustCall<sup>SM</sup> 7 Cents Preferred Promotion #276

REC'D JUN 08 2004

- (A) The sign-up period for this promotion is June 15, 2004 through August 31, 2004. Orders for new Service must be activated by September 15, 2004. This promotion is available to Residential Customers that previously subscribed to local dial tone service from an SBC Affiliate and/or subscribed to a long distance Service of the Company and have cancelled that service. To receive the promotion, a Customer must now subscribe to local dial tone service from an SBC Affiliate and the Company's interstate interexchange JustCall<sup>SM</sup> 7 Cent optional calling plan through a Company-designated outbound sales contact or by calling a toll-free number provided through Company-designated Teleservices sales channels that are specific to this promotion.
- (B) To participate in the promotion, Residential Customers must (1) subscribe to the JustCall<sup>SM</sup> 7 Cents Preferred optional calling plan, as referenced in Section 3.4.3 (AE).9.b of this Tariff during the sign-up period; (2) maintain the requirements specified in Section 3.4.3 (AE).1, 3.4.3 (AE).3, and Section 3.4.3 (AE).4, of this Tariff; and (3) continue to subscribe to the JustCall<sup>SM</sup> 7 Cents Preferred optional calling plan for the provision of intrastate calling during the benefit period of this promotional offering.
- (C) Customers participating in this promotion will receive up to the first 30 minutes of interstate/intrastate usage free-of-charge per month and a reduction of the \$0.07 intrastate per minute rate after the first 30 minutes to \$0.05 per minute for the first six (6) months from activation of Service.
- (D) If the Customer fails to maintain the requirements specified in Section 3.4.3 (AE).1, Section 3.4.3 (AE).3, and Section 3.4.3 (AE).4 of this Tariff and/or fails to continue to subscribe to the JustCall<sup>5M</sup> 7 Cents Preferred optional calling plan for the provision of intrastate calling, the Customer will no longer qualify for the promotion. The per minute usage rate that applies after the expiration of the promotion may be found in Section 4.4.3 (AE).9.b of this Tariff.

Missouri Public Service Commission

FILED JUN 15 2004

Issued: June 8, 2004

Effective: June 15, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588



Ν

AUG 3 1 2004 Joy RS 270 Public Service Commission MISSOURI

- 6.33 Business Unlimited Long Distance Summer Promotion #277
  - 6.33.1 The sign-up period for Business Unlimited Long Distance Plans Summer Promotion #277 is July 1, 2004 through March 31, 2005. Service must be activated by May 31, 2005.
  - 6.33.2 This promotion is available to Business Customers that during the sign-up period:
    - (A) subscribe to Business Unlimited Long Distance Plans or
    - (B) commit to a new 1-year term agreement for the Customer's existing Business Unlimited Long Distance Plans optional calling plan. (If the Customer commits to a 1-year term plan, the Customer may upgrade or downgrade the Business Unlimited Long Distance Plans if the Customer adds or removes a business access line, the Company will waive the early termination fee associated with the change in the plan.)
  - 6.33.3 New customers must subscribe to and maintain or existing Customers currently subscribe to and maintain a business access line of a SBC Affiliate and any service or product of a SBC Affiliate listed below:
    - (A) SimpleLink<sup>SM</sup> or
    - (B) Business Solutions<sup>™</sup> or
    - (C) Centrex Service (1 to 10 stations lines only) or
    - (D) Custom  $BizSaver^{SM}$ .
  - 6.33.4 If the Customer fails to maintain the requirements specified in Section 3.7.48 (B).3, Section 3.7.48 (B).6 and Section 6.33.3 of the Tariff and/or fails to continue to subscribe to the Business Unlimited Long Distance Plans optional calling plan for provisions of interstate and intrastate InterLATA, or interstate, intrastate InterLATA, and intrastate IntraLATA calling, the Customer will no longer qualify for the promotional benefits.
  - 6.33.5 For new Customers subscribing to Business Unlimited Long Distance Plans during the sign-up period, the Customer will receive a \$10.04 per month credit off the MRC for unlimited interstate and intrastate 1+ outbound calling for the first access line for twelve (12) months from the activation of Service.
  - 6.33.6 For existing Customers extending their term plan agreement, the Customer will receive a \$10.04 per month credit off the MRC for unlimited interstate and intrastate 1+ outbound calling for the first access line for twelve (12) months beginning the month in which the order is processed.

Issued: December 10, 2004

Effective: December 20, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

 $C \\ C$ 

#### **SECTION 6 - PROMOTIONS**

- 6.33 Business Unlimited Long Distance Summer Promotion #277
  - 6.33.1 The sign-up period for Business Unlimited Long Distance Plans Summer Promotion #277 is July 1, 2004 through December 31, 2004. Service must be activated by March 1, 2005.
  - 6.33.2 This promotion is available to Business Customers that during the sign-up period:
    - (A) subscribe to Business Unlimited Long Distance Plans or
    - (B) commit to a new 1-year term agreement for the Customer's existing Business Unlimited Long Distance Plans optional calling plan. (If the Customer commits to a 1-year term plan, the Customer may upgrade or downgrade the Business Unlimited Long Distance Plans if the Customer adds or removes a business access line, the Company will waive the early termination fee associated with the change in the plan.)
  - 6.33.3 New customers must subscribe to and maintain or existing Customers currently subscribe to and maintain a business access line of a SBC Affiliate and any service or product of a SBC Affiliate listed below:
    - (A) SimpleLink<sup>SM</sup> or
    - (B) Business Solutions<sup>™</sup> or
    - (C) Centrex Service (1 to 10 stations lines only) or
    - (D) Custom BizSaver<sup>SM</sup>.
  - 6.33.4 If the Customer fails to maintain the requirements specified in Section 3.7.48 (B).3, Section 3.7.48 (B).6 and Section 6.33.3 of the Tariff and/or fails to continue to subscribe to the Business Unlimited Long Distance Plans optional calling plan for provisions of interstate and intrastate InterLATA, or interstate, intrastate InterLATA, and intrastate IntraLATA calling, the Customer will no longer qualify for the promotional benefits.
  - 6.33.5 For new Customers subscribing to Business Unlimited Long Distance Plans during the sign-up period, the Customer will receive a \$10.04 per month credit off the MRC for unlimited interstate and intrastate 1+ outbound calling for the first access line for twelve (12) months from the activation of Service.
  - 6.33.6 For existing Customers extending their term plan agreement, the Customer will receive a \$10.04 per month credit off the MRC for unlimited interstate and intrastate 1+ outbound calling for the first access line for twelve (12) months beginning the month in which the order is processed.



DEC 2/0 2004

SOURI

Issued: September 20, 2004

Public Service Commission Effective: September 30, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588



Southwestern Bell Communications Services, Inc.	PSC Mo No.
d/b/a SBC Long Distance	

Original Sheet 271

REC'D JUN 16 2004

Missouri Public Service Commission

- 6.33 Business Unlimited Long Distance Summer Promotion #277
  - 6.33.1 The sign-up period for Business Unlimited Long Distance Plans Summer Promotion #277 is July 1, 2004 through September 29, 2004. Service must be activated by November 30, 2004.
  - 6.33.2 This promotion is available to Business Customers that during the sign-up period:

**SECTION 6 - PROMOTIONS** 

- (A) subscribe to Business Unlimited Long Distance Plans or
- (B) commit to a new 1-year term agreement for the Customer's existing Business Unlimited Long Distance Plans optional calling plan. (If the Customer commits to a 1-year term plan, the Customer may upgrade or downgrade the Business Unlimited Long Distance Plans if the Customer adds or removes a business access line, the Company will waive the early termination fee associated with the change in the plan.)
- 6.33.3 New customers must subscribe to and maintain or existing Customers currently subscribe to and maintain a business access line of a SBC Affiliate and any service or product of a SBC Affiliate listed below:
  - (A) SimpleLinkK or
  - (B) Business Solutions K or
  - (C) Centrex Service (1 to 10 stations lines only) or
  - (D) Custom BizSaver K.
- 6.33.4 If the Customer fails to maintain the requirements specified in Section 3.7.48 (B).3, Section 3.7.48 (B).6 and Section 6.33.3 of the Tariff and/or fails to continue to subscribe to the Business Unlimited Long Distance Plans optional calling plan for provisions of interstate and intrastate InterLATA, or interstate, intrastate InterLATA, and intrastate IntraLATA calling, the Customer will no longer qualify for the promotional benefits.
- 6.33.5 For new Customers subscribing to Business Unlimited Long Distance Plans during the sign-up period, the Customer will receive a \$10.04 per month credit off the MRC for unlimited interstate and intrastate 1+ outbound calling for the first access line for twelve (12) months from the activation of Service.
- 6.33.6 For existing Customers extending their term plan agreement, the Customer will receive a \$10.04 per month credit off the MRC for unlimited interstate and intrastate 1+ outbound calling for the first access line for twelve (12) months beginning the month in which the order is processed. Missouri Public

Service Commission

FILED JUL 01 2004

Ν

ommission

Issued: June 16, 2004

Effective: July 1, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588



MISSOURI

- 6.34 JustCall<sup>SM\*</sup> Three/30 Promotion #282 (\* JustCall<sup>SM</sup> 7 Cents and JustCall<sup>SM</sup> 9 Cents Standard)
  - (A) The sign-up period for this promotion is September 16, 2004 through December 13, 2004. Orders for new Service must be activated by December 28, 2004. This promotion is available to new and existing Residential Customers that subscribe to local dial tone service from an SBC Affiliate and one of the following Company's interstate/intrastate interexchange JustCall<sup>SM</sup> 7 Cents, or JustCall<sup>SM</sup> 9 Cents Standard optional calling plans through a Company-designated outbound sales contact or by calling a toll-free number provided through Company-designated Teleservices sales channels and specific to this promotion.
  - (B) To participate in this promotion, Residential Customers must (1) subscribe to either the JustCall<sup>SM</sup> 7 Cents, or JustCall<sup>SM</sup> 9 Cents Standard, optional calling plan, as referenced in Section 3.4.3 (AE).11.b and 3.4.3 (AE).5.b, respectively, of this Tariff, during the sign-up period; (2) maintain the requirements specified in Section 3.4.3 (AE).1, 3.4.3 (AE).3, and Section 3.4.3 (AE).4, of this Tariff; and (3) continue to subscribe to the JustCall<sup>SM</sup> 7 Cents, or JustCall<sup>SM</sup> 9 Cents Standard optional calling plans for the provision of interstate/intrastate calling during the benefit period of this promotional offering.
  - (C) Customers participating in this promotion will receive up to the first 30 minutes of interstate/intrastate per-minute usage free-of-charge per month for the first three (3) months from activation date for all WTNs under the BTN in which the promotion is applied. In the event a Customer subscribes to this promotion in the middle of a billing cycle, the promotion will be pro-rated for that month. If the Customer fails to use the first 30 minutes of interstate/intrastate per minute usage within a billing cycle, the minutes will not be carried over into the following monthly billing cycle.
  - (D) If the Customer fails to maintain the requirements specified in Section 3.4.3 (AE).1, Section 3.4.3 (AE).3, and Section 3.4.3 (AE).4 of this Tariff and/or fails to continue to subscribe to either the JustCall<sup>SM</sup> 7 Cents, or JustCall<sup>SM</sup> 9 Cents Standard, optional calling plan for the provision of interstate calling, the Customer will no longer qualify for the promotion. The per minute usage rate that applies after the expiration of the promotion may be found in Sections 4.4.3 (AE).11.b and 4.4.3 (AE).5.b, respectively of this Tariff.

Issued: September 8, 2004

Effective: September 16, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149

- 6.35 JustCall<sup>SM\*</sup> Three/30 Promotion #282A (\*JustCall<sup>SM</sup> 7 Cents and JustCall<sup>SM</sup> 9 Cents Standard)
  - (A) The sign-up period for this promotion is December 14, 2004 through March 31, 2005. Orders for new Service must be activated by April 15, 2005. This promotion is available to new and existing Residential Customers that subscribe to local dial tone service from an SBC Affiliate and one of the following Company's interstate/intrastate interexchange JustCall<sup>SM</sup> 7 Cents, or JustCall<sup>SM</sup> 9 Cents Standard optional calling plans through a Company-designated outbound sales contact or by calling a toll-free number provided through Company-designated Teleservices sales channels and specific to this promotion.
  - (B) To participate in this promotion, Residential Customers must (1) subscribe to either the JustCall<sup>SM</sup> 7 Cents, or JustCall<sup>SM</sup> 9 Cents Standard, optional calling plan, as referenced in Section 3.4.3 (AE).11.b and 3.4.3 (AE).5.b, respectively, of this Tariff, during the sign-up period; (2) maintain the requirements specified in Section 3.4.3 (AE).1, 3.4.3 (AE).3, and Section 3.4.3 (AE).4, of this Tariff; and (3) continue to subscribe to the JustCall<sup>SM</sup> 7 Cents, or JustCall<sup>SM</sup> 9 Cents Standard optional calling plans for the provision of interstate/intrastate calling during the benefit period of this promotional offering.
  - (C) Customers participating in this promotion will receive up to the first 30 minutes of interstate/intrastate per-minute usage free-of-charge per month for the first three (3) months from activation date for all WTNs under the BTN in which the promotion is applied. In the event a Customer subscribes to this promotion in the middle of a billing cycle, the promotion will be pro-rated for that month. If the Customer fails to use the first 30 minutes of interstate/intrastate per minute usage within a billing cycle, the minutes will not be carried over into the following monthly billing cycle.
  - (D) If the Customer fails to maintain the requirements specified in Section 3.4.3 (AE).1, Section 3.4.3 (AE).3, and Section 3.4.3 (AE).4 of this Tariff and/or fails to continue to subscribe to either the JustCall<sup>SM</sup> 7 Cents, or JustCall<sup>SM</sup> 9 Cents Standard, optional calling plan for the provision of interstate calling, the Customer will no longer qualify for the promotion. The per minute usage rate that applies after the expiration of the promotion may be found in Sections 4.4.3 (AE).11.b and 4.4.3 (AE).5.b, respectively of this Tariff. This promotion cannot be combined with any other promotional offer.

Issued: December 7, 2004

Effective: December 14, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149 Ν

N

6.36 JustCall<sup>SM</sup> Standard II Three/30 Promotion #284

- (A) The sign-up period for this promotion is December 14, 2004 through March 31, 2005. Orders for new Service must be activated by April 15, 2005. This promotion is available to new and existing Residential Customers that subscribe to local dial tone service from an SBC Affiliate and the following Company's interstate/intrastate interexchange JustCall<sup>SM</sup> Standard II optional calling plan through a Company-designated outbound sales contact, by calling a toll-free number provided through Company-designated Teleservices sales channels, or Customers that call in to a Company-designated sales channel to discuss changes to their Value Plus Flat Rate optional calling plan (as referenced in Section(s) 3.4.3 (Y) and 4.4.3 (Y) of this Tariff), and as determined by the Company.
- (B) To participate in the promotion, Residential Customers must (1) subscribe to the JustCall<sup>SM</sup> Standard II optional calling plan, as referenced in Section 3.4.3 (AE).7.a of this Tariff, during the sign-up period; (2) maintain the requirements specified in Section 3.4.3 (AE).1, 3.4.3 (AE).3, and Section 3.4.3 (AE).4, of this Tariff; and (3) continue to subscribe to the JustCall<sup>SM</sup> Standard II optional calling plan for the provision of interstate/intrastate calling during the benefit period of this promotional offering.

Ν

Issued: December 7, 2004

Effective: December 14, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

### 6.36 JustCall<sup>SM</sup> Standard II Three/30 Promotion #284 (continued)

- (C) Customers participating in this promotion will receive up to the first 30 minutes of interstate/intrastate per minute usage free-of-charge per month for the first three (3) months from activation date of Service for all WTNs under the BTN in which the promotion is applied. In the event a Customer subscribes to this promotion in the middle of a billing cycle, the promotion will be pro-rated for that month. If the Customer fails to use the first 30 minutes of interstate/intrastate per minute usage within a billing cycle, the minutes will not be carried over into the following monthly billing cycle.
- (D) If the Customer fails to maintain the requirements specified in Section 3.4.3 (AE).1, Section 3.4.3 (AE).3, and Section 3.4.3 (AE).4 of this Tariff and/or fails to continue to subscribe to the JustCall<sup>SM</sup> Standard II optional calling plan for the provision of interstate calling, the Customer will no longer qualify for the promotion. The per minute usage rate that applies after the expiration of the promotion may be found in Section 4.4.3 (AE).7.a of this Tariff. This promotion cannot be combined with any other promotional offer.

Issued: December 7, 2004

Effective: December 14, 2004

Lisa Porterfield, Associate Director Regulatory 5850 W. Las Positas Blvd., Pleasanton, California 94588

CANCELLED December 24, 2004 Missouri Public Service Commission TN-2005-0149